UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): March 18, 2019

CONCRETE PUMPING HOLDINGS, INC.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation) **001-38166** (Commission File Number) 83-1779605 (IRS Employer Identification No.)

6461 Downing Street Denver, Colorado 80229

(Address of principal executive offices, including zip code)

Registrant's telephone number, including area code: (303) 289-7497

N/A

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company \square

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

1

Item 1.01 Entry into a Material Definitive Agreement.

Interest Purchase Agreement

On March 18, 2019, Concrete Pumping Holdings, Inc., a Delaware corporation (the "Company"), Brundage-Bone Concrete Pumping, Inc., a Colorado corporation and wholly-owned indirect subsidiary of the Company ("BBCP") and CPH Acquisition, LLC, a Delaware limited liability company and wholly owned subsidiary of BBCP ("CPHA LLC" and, together with BBCP, the "Buyers"), entered into an Interest Purchase Agreement (the "Interest Purchase Agreement") with ASC Equipment, LP, a Texas limited partnership ("ASC"), Capital Pumping, LP, a Texas limited partnership ("CP"), MC Services, LLC, a Texas limited liability company ("MCS" and, together with ASC and CP, the "Acquired Companies"), Capital Rentals, LLC, a Texas limited liability company ("CTCS"), A. Keith Crawford ("A. Crawford") and Melinda Crawford ("M. Crawford" and, together with the Acquired Companies, CR LLC, CTCS and A. Crawford, the "Sellers"), pursuant to which, subject to the satisfaction or waiver of certain conditions set forth therein, (i) BBCP will purchase all of the outstanding (x) limited partnership interests in CP and ASC from A. Crawford and (y) limited liability company interests in MCS from M. Crawford, and (ii) CPHA LLC will purchase all of the general partnership interests in CP and ASC from CR LLC and CTCS, respectively (the transactions contemplated by the Interest Purchase Agreement, the "Transactions"). As a result of the Transactions, each of the Acquired Companies of the Company.

Transaction Consideration

The consideration for the Transactions will consist of an aggregate of \$129.2 million payable in cash, subject to adjustments as described in the Interest Purchase Agreement.

Representations, Warranties and Covenants; Indemnification

The parties to the Interest Purchase Agreement have made customary representations, warranties and covenants in the Interest Purchase Agreement, including, among others, covenants with respect to the conduct of the Acquired Companies during the period between execution of the Interest Purchase Agreement and the completion of the Transactions. In addition, the parties to the Interest Purchase Agreement have each agreed to use commercially reasonable efforts to cause the Transactions to be consummated. Subject to certain limitations and conditions set forth in the Interest Purchase Agreement, the Company and the Sellers have agreed to indemnify each other for, among other things, breaches of representations, warranties and covenants contained in the Interest Purchase Agreement, and certain tax and other pre- and post-closing liabilities. To supplement the indemnification provided by the Sellers, the Company has obtained a standard representation and warranty insurance policy.

Conditions to Closing

The closing of the Transactions is subject to certain conditions, including, among others, (i) the Company (or its subsidiaries) obtaining sufficient financing to pay the consideration under the Interest Purchase Agreement, (ii) approval by the owners of the Acquired Companies to enter into the Transactions, and (iii) the expiration or termination of the applicable waiting periods under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

Termination

The Interest Purchase Agreement may be terminated by the Buyers and the Sellers under certain circumstances, including, among others, (i) by mutual written consent of the Buyers and the Sellers, (ii) by the Buyers if the closing of the Transactions has not occurred by July 16, 2019; and (iii) by the Sellers if the closing of the Transactions has not occurred prior to June 17, 2019, subject to extension to July 1, 2019 in certain circumstances. In the event that the Sellers exercise their termination rights pursuant to clause (iii) of the preceding sentence, the Company has agreed to pay the Sellers \$5.0 million in cash as liquidated damages.

The foregoing description of the Interest Purchase Agreement and the Transactions does not purport to be complete and is qualified in its entirety by the terms and conditions of the Interest Purchase Agreement, a copy of which is attached hereto as Exhibit 2.1 and is incorporated herein by reference. The Interest Purchase Agreement contains representations, warranties and covenants that the respective parties made to each other as of the date of such agreement or other specific dates. The assertions embodied in those representations, warranties and covenants were made for purposes of the contract among the respective parties and are subject to important qualifications and limitations agreed to by the parties in connection with negotiating such agreement. The Interest Purchase Agreement has been attached to provide investors with information regarding its terms. It is not intended to provide any other factual information about the Company, the Acquired Companies or any other party to the Interest Purchase Agreement. In particular, the representations, warranties, covenants and agreements contained in the Interest Purchase Agreement, which were made only for purposes of such agreement and as of specific dates, were solely for the benefit of the parties to the Interest Purchase Agreement, may be subject to limitations agreed upon by the contracting parties (including being qualified by confidential disclosures made for the purposes of allocating contractual risk between the parties to the Interest Purchase Agreement instead of establishing these matters as facts) and may be subject to standards of materiality applicable to the contracting parties that differ from those applicable to investors and reports and documents filed with the U.S. Securities and Exchange Commission (the "SEC"). Investors should not rely on the representations, warranties, covenants and agreements, or any descriptions thereof, as characterizations of the actual state of facts or condition of any party to the Interest Purchase Agreement. In addition, the representations, warranties, covenants and agreements and other terms of the Interest Purchase Agreement may be subject to subsequent waiver or modification. Moreover, information concerning the subject matter of the representations and warranties and other terms may change after the date of the Interest Purchase Agreement, which subsequent information may or may not be fully reflected in the Company's public disclosures.

Item 7.01 Regulation FD Disclosure.

On March 18, 2019, the Company announced the execution of the Interest Purchase Agreement. A copy of the press release is furnished herewith as Exhibit 99.1. In addition, furnished herewith as Exhibit 99.2 is an investor presentation that the Company will use when discussing the Transactions with certain existing and potential investors.

The information in this Item 7.01 and Exhibit 99.1 attached hereto shall not be deemed "filed" for purposes of Section 18 of the Exchange Act of 1934, as amended (the "Exchange Act"), or otherwise subject to the liabilities of that section, nor shall it be deemed incorporated by reference in any filing under the Securities Act or the Exchange Act, except as expressly set forth by specific reference in such filing.

Forward-Looking Statements

This Current Report on Form 8-K includes "forward-looking statements" within the meaning of the "safe harbor" provisions of the Private Securities Litigation Reform Act of 1995. The Company's, the Buyers' and the Acquired Companies' actual results may differ from their expectations, estimates and projections and consequently, you should not rely on these forward looking statements as predictions of future events. Words such as "expect," "estimate," "project," "budget," "forecast," "anticipate," "intend," "plan," "may," "will," "could," "should," "believes," "predicts," "potential," "continue," and similar expressions are intended to identify such forwardlooking statements. These forward-looking statements include, without limitation, the Company's, the Buyers' and the Acquired Companies' expectations with respect to future performance and anticipated financial impacts of the Transactions, the satisfaction of the closing conditions to the Transactions and the timing of the completion of the Transactions. These forward-looking statements involve significant risks and uncertainties that could cause the actual results to differ materially from the expected results. Most of these factors are outside the Company's, the Buyers' and the Acquired Companies' control and are difficult to predict. Factors that may cause such differences include, but are not limited to: (1) the occurrence of any event, change or other circumstances that could give rise to the termination of the Interest Purchase Agreement, (2) the outcome of any legal proceedings that may be instituted against the Company, the Buyers and the Acquired Companies following the announcement of the Interest Purchase Agreement and the transactions contemplated therein; (3) the inability to complete the Transactions, including due to failure to obtain the financing required to complete the Transactions; (4) the occurrence of any event, change or other circumstance that could give rise to the termination of the Interest Purchase Agreement or could otherwise cause the Transactions to fail to close; (5) the risk that the Transactions disrupt current plans and operations; (6) the ability to recognize the anticipated benefits of the Transactions, which may be affected by, among other things, competition, the ability of the combined company to grow and manage growth profitably and retain its key employees; (7) changes in applicable laws or regulations; (8) the possibility that the Acquired Companies or the combined company may be adversely affected by other economic, business, and/or competitive factors; and (9) other risks and uncertainties indicated from time to time in the Company's filings with the SEC. The Company cautions that the foregoing list of factors is not exclusive. The Company cautions readers not to place undue reliance upon any forward-looking statements, which speak only as of the date made. The Company does not undertake or accept any obligation or undertaking to release publicly any updates or revisions to any forward-looking statements to reflect any change in its expectations or any change in events, conditions or circumstances on which any such statement is based.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits

Exhibit No.	Description
<u>2.1*</u>	Interest Purchase Agreement, dated as of March 18, 2019, by and between the Company, Brundage-Bone Concrete Pumping, Inc., CPH Acquisition, LLC, ASC Equipment, LP, Capital Pumping, LP, MC Services, LLC, Capital Rentals, LLC, Central Texas Concrete Services, LLC, A. Keith Crawford and Melinda Crawford.
99.1	Press Release, dated March 18, 2019.
99.2	Investor Presentation, dated March 18, 2019

* The exhibits and schedules to this Exhibit have been omitted in accordance with Regulation S-K Item 601(b)(2). The Company agrees to furnish supplementally a copy of any omitted exhibit or schedule to the SEC upon its request.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

CONCRETE PUMPING HOLDINGS, INC.

By: /s/ Iain Humphries

Name: Iain Humphries Title: Chief Financial Officer and Secretary

Dated: March 18, 2019

INTEREST PURCHASE AGREEMENT

DATED MARCH 18, 2019,

BY AND AMONG

CAPITAL PUMPING, LP; ASC EQUIPMENT, LP; MC SERVICES, LLC; CAPITAL RENTALS, LLC; CENTRAL TEXAS CONCRETE SERVICES, LLC; A. KEITH CRAWFORD; and MELINDA CRAWFORD (collectively, the Company Parties,

and

BRUNDAGE-BONE CONCRETE PUMPING, INC., and CPH ACQUISITION, LLC, as Buyers

and

CONCRETE PUMPING HOLDINGS, INC., solely for purposes of <u>Section 8.3</u>

TABLE OF CONTENTS

		Page
ARTICLE	E 1	1
1.1	Definitions	1
1.2	Construction	14
ARTICLE	E 2	15
2.1	Purchase and Sale of Purchased Interests	15
2.2	Transferred Assets; Distribution of Cash	15
2.3	Partner, Member and Manager Acknowledgment	16
ARTICLE	23	16
3.1	Purchase Price	16
3.2	Payoff Letters; Lien Releases	16
3.3	Payment of Purchase Price	16
3.4	Equipment Inspection	17
3.5	Withholding	18
ARTICLE		19
4.1	Closing	19
4.2	Buyers' Conditions to Closing	19
4.3	Company Parties' Conditions to Close	21
ARTICLE	25	22
5.1	Organization	22
5.2	Due Authorization	22
5.3	No Conflict	23
5.4	Compliance with Laws	23
5.5	Governmental Consents	23
5.6	Capitalization	23
5.7	Assets Used in the Business	24
5.8	Sufficiency of Assets; No Third Party Rights	25
	- i -	

	5.9	Financial Statements	25
	5.10	Undisclosed Liabilities; Indebtedness	25
	5.11	Litigation and Actions	26
	5.12	Contracts	26
	5.13	Employee Benefit Plans	27
	5.14	Employment Matters	29
	5.15	Taxes	31
	5.16	Insurance	33
	5.17	Licenses and Permits	34
	5.18	Owned Real Property	34
	5.19	Leased Real Property	35
	5.20	Environmental	35
	5.21	Intellectual Property	36
	5.22	Conduct of the Business	38
	5.23	Affiliate Matters	38
	5.24	Creditors, Solvency and Bankruptcy	39
	5.25	Bank Accounts	39
	5.26	Accounts Receivable; Warranty Claims; Accounts Payable; Excluded Insurance Proceeds	39
	5.27	Information Supplied	39
	5.28	Full Disclosure	39
	5.29	Brokerage	40
	5.30	No Additional Representations or Warranties	40
ARTI	ICLE 6		40
	6.1	Organization	40
	6.2	Due Authorization	40
	6.3	No Conflict	41
	6.4	Brokerage	41
		- ii -	

	6.5	Financing	41
	6.6	No Additional Representations or Warranties	41
ART	ICLE 7	1	41
	7.1	Conduct of Business	41
	7.2	Pending Insurance Claims	43
	7.3	HSR Act	43
	7.4	Financing Cooperation	43
	7.5	Termination of Affiliate Agreements	44
	7.6	Authority	44
	7.7	Dispute Cooperation	44
	7.8	Updated Information	44
	7.9	Exclusivity	45
	7.10	Release	45
	7.11	Environmental Reports	45
ARTICLE 8			46
	8.1	HSR Act	46
	8.2	R&W Insurance Policy	47
	8.3	Financing	47
	8.4	Dispute Cooperation	47
ART	ICLE 9		47
	9.1	Support of Transaction	47
	9.2	Further Assurances	47
	9.3	Transfer Taxes	48
	9.4	Allocation of Taxes	48
	9.5	Notification of Certain Matters	51
	9.6	Run-Off Period	51
		- iii -	

ARTI	ICLE 1	0	52
	10.1	Acknowledgment	52
	10.2	Non-Competition	52
	10.3	Non Solicitation	52
	10.4	Interference with Relationships	52
	10.5	Confidential Information	53
	10.6	Enforceability; Blue Pencil	53
	10.7	Remedies	53
ARTI	ICLE 1	1	53
	11.1	Termination	53
	11.2	Effect of Termination	55
	11.3	Liquidated Damages	55
ARTI	ICLE 1	2	55
	12.1	Survival	55
	12.2	Indemnification by the Sellers	56
	12.3	Indemnification by Buyers	56
	12.4	Limitations on Indemnification	56
	12.5	Indemnification Claim Process	57
	12.6	Indemnification Procedures for Non-Third Party Claims	59
	12.7	Exclusive Remedy	59
	12.8	Tax Treatment of Indemnity Payments	60
	12.9	Determination of Breaches and Losses	60
	12.10	Effect of Investigation	60
	12.11	Resolution of Disputes Under Article 12	61
ARTI	ICLE 1	3	61
	13.1	Waiver	61
	13.2	Notices	61
	13.3	Assignment	62
		- iv -	

13.4	Successors and Assigns	62
13.5	Rights of Third Parties	62
13.6	Joint and Several Obligations	62
13.7	Expenses	63
13.8	Governing Law; Jurisdiction; Service of Process	63
13.9	Captions; Counterparts	63
13.10	Schedules, Annexes and Exhibits	63
13.11	Entire Agreement	64
13.12	Amendments	64
13.13	Publicity	64
13.14	Severability	64
13.15	Representation by Counsel	64
13.16	Conflict of Terms	65
13.17	Certain Matters Regarding Debt Lenders	65
	- V -	

INTEREST PURCHASE AGREEMENT

THIS INTEREST PURCHASE AGREEMENT, dated as of March 18, 2019, is by and among ASC EQUIPMENT, LP, a Texas limited partnership ("<u>ASC</u>"), CAPITAL PUMPING, LP, a Texas limited partnership ("<u>CP</u>"), MC SERVICES, LLC, a Texas limited liability company ("<u>MCS</u>"), CAPITAL RENTALS, LLC, a Texas limited liability company ("<u>CR LLC</u>"), CENTRAL TEXAS CONCRETE SERVICES, LLC, a Texas limited liability company ("<u>CTCS</u>"), A. KEITH CRAWFORD, MELINDA CRAWFORD, BRUNDAGE-BONE CONCRETE PUMPING, INC., a Colorado corporation ("<u>BBCP</u>"), and CPH ACQUISITION, LLC, a Delaware limited liability company ("<u>CPHA LLC</u>"), and solely for purposes of <u>Section 8.3</u>, CONCRETE PUMPING HOLDINGS, INC., a Delaware corporation ("<u>CPHI</u>").

WHEREAS, the Companies are engaged in the business of concrete pumping within the State of Texas (the "Business"); and

WHEREAS, (a) CP's sole limited partner is A. Keith Crawford, and CP's sole general partner is CR LLC; (b) ASC's sole limited partner is A. Keith Crawford, and ASC's sole general partner is CTCS; and (c) MCS's sole manager is Melinda Crawford and its sole member is Melinda Crawford; and

WHEREAS, (a) CR LLC's only members are A. Keith Crawford and Melinda Crawford; and (b) CTCS' only members are A. Keith Crawford and Melinda Crawford; and

WHEREAS, on the terms and conditions set forth in this Agreement, (A) A. Keith Crawford desires to sell, assign, transfer and convey his limited partnership interests in CP ("<u>CP Limited Partnership Interest</u>") to BBCP, (B) A. Keith Crawford desires to sell, assign, transfer and convey his limited partnership interests in ASC ("<u>ASC Limited Partnership Interest</u>") to BBCP, (C) CR LLC desires to sell its general partnership interest in CP ("<u>CP General Partnership Interest</u>") to CPHA LLC, (D) CTCS desires to sell its general partnership interest in ASC ("<u>ASC General Partnership Interest</u>") to CPHA LLC, (D) Melinda Crawford desires to sell her limited liability company interests in MCS ("<u>MCS Membership Interest</u>") to BBCP.

NOW, THEREFORE, in consideration of the premises and mutual covenants contained in this Agreement and of other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties, intending to be legally bound hereby, agree as follows:

ARTICLE 1 CERTAIN DEFINITIONS

1.1 Definitions. As used herein, the following terms shall have the following meanings:

"<u>ABL Agreement</u>" means that certain Credit Agreement, dated as of December 6, 2018, by and among Wells Fargo Bank, National Association, as agent, sole lead arranger and sole bookrunner, the lenders that are parties thereto, as lenders, Wells Fargo Capital Finance (UK) Limited, as UK security agent, CPHI, Concrete Pumping Intermediate Acquisition Corp., Brundage-Bone Concrete Pumping Holdings Inc., BBCP, Eco-Pan, Inc., Camfaud Concrete Pumps Limited and Premier Concrete Pumping Limited, as it may be amended, modified or supplemented in accordance with its terms.

"Acquisition Proposal" has the meaning set forth in Section 7.9.

"<u>Action</u>" means any civil or criminal claim, action, cause of action, suit, proceeding, hearing, investigation, inquiry administrative action, lawsuit, demand or litigation, whether at law or in equity, or before any court, arbitrator, arbitration panel or Governmental Authority.

"<u>Affiliate</u>" means, with respect to any specified Person, any Person that, directly or indirectly, controls, is controlled by, or is under common control with, such specified Person, through one or more intermediaries or otherwise. For purposes of this definition, the term "control" (including the terms "controlling," "controlled by" and "under common control with") means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by Contract or otherwise.

"<u>Affiliate Agreement</u>" means any Contract, other than this Agreement, between or among any Company Party or Affiliate of a Company Party, on the one hand, and any other Company Party or Affiliate of a Company Party, on the other hand.

"<u>Affiliated Group</u>" means any affiliated group of corporations or other entities that files a combined, consolidated, unitary or similar Tax Return that includes (or was required to include) any Company, including any affiliated group within the meaning of Section 1504(a) of the Code or any similar group defined under a similar provision of applicable Law.

"Agreement" means this Interest Purchase Agreement, together with all exhibits and schedules hereto, as the same may be amended or restated from time to time.

"Antitrust Authority" means the Antitrust Division of the United States Department of Justice and the United States Federal Trade Commission.

"<u>Antitrust Information or Document Request</u>" means any request or demand for the production, delivery or disclosure of documents or other evidence, or any request or demand for the production of witnesses for interviews or depositions or other oral or written testimony, by any Antitrust Authority relating to the transactions contemplated hereby or by any third party challenging the transactions contemplated hereby, including any so called "second request" for additional information or documentary material or any civil investigative demand made or issued by the Antitrust Division of the United States Department of Justice or the United States Federal Trade Commission or any subpoena, interrogatory or deposition by any Antitrust Authority.

"<u>ASC</u>" has the meaning set forth in the introductory paragraph.

"ASC General Partnership Interest" has the meaning set forth in the Recitals.

"ASC Limited Partnership Interest" has the meaning set forth in the Recitals.

"<u>Assignments</u>" means the assignments of the Purchased Interests made by the Sellers in favor of Buyers, each in form and substance reasonably satisfactory to Buyers.

"Audited Financial Statements" has the meaning set forth in Section 5.9.

"Bill of Sale" means a quit claim bill of sale and assignment made by the Companies in favor of Sellers or their designees, dated as of the Closing Date, in form and substance reasonably acceptable to the parties.

"Business" has the meaning set forth in the Recitals.

"Business Day" means any day that is not a Saturday, a Sunday or other day on which the Federal Reserve Bank of New York is closed.

2

"Business Personnel" means the employees, consultants and other independent contractors of any Company.

"<u>Buyer Cure Period</u>" has the meaning set forth in <u>Section 11.1.3(a)</u>.

"Buyer Indemnitee" has the meaning set forth in Section 12.2.

"<u>Buyers</u>" means BBCP and CPHA LLC, and "<u>Buyer</u>" means either of them.

"<u>Cap</u>" has the meaning set forth in <u>Section 12.4</u>.

"<u>Claims Notice</u>" has the meaning set forth in <u>Section 12.5.1</u>.

"<u>Closing</u>" has the meaning set forth in <u>Section 4.1</u>.

"Closing Date" has the meaning set forth in Section 4.1.

"COBRA" means the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended.

"Code" means the Internal Revenue Code of 1986, as amended, and the Treasury Regulations promulgated thereunder.

"Companies" mean ASC, CP and MCS, and "Company" means any one of them.

"Companies' Audit Firm" has the meaning set forth in the definition of "Required Financial Information."

"Company Cure Period" has the meaning set forth in Section 11.1.2(b).

"Company Indemnitee" has the meaning set forth in Section 12.3.

"<u>Company Parties</u>" means (a) prior to the Closing, the Companies, CR LLC, CTCS, A. Keith Crawford and Melinda Crawford, and "<u>Company Party</u>" means any one of them, and (b) after the Closing, CR LLC, CTCS, A. Keith Crawford and Melinda Crawford.

"<u>Company Parties' Knowledge</u>" or "<u>Knowledge of the Company Parties</u>" means the knowledge of A. Keith Crawford or Gary Shake following reasonable inquiry of their direct reports.

"Company Return" has the meaning set forth in Section 9.4.3(b).

"Compliance with Laws Claims" has the meaning set forth in Section 12.4.3.

"Confidential Information" has the meaning set forth in Section 10.5.

"<u>Contract</u>" means any agreement, contract, instrument, commitment, lease, guaranty, indenture, license, sales order and purchase order, mortgage, commitment, arrangement, of every kind, or other legally binding arrangement or understanding, whether verbal or written.

"<u>CP</u>" has the meaning set forth in the introductory paragraph.

"CP General Partnership Interest" has the meaning set forth in the Recitals.

3

"CP Limited Partnership Interest" has the meaning set forth in the Recitals.

"CPHI" has the meaning set forth in the introductory paragraph.

"<u>CPHI Stockholder Meeting</u>" has the meaning set forth in <u>Section 5.27</u>.

"<u>CR LLC</u>" has the meaning set forth in the introductory paragraph.

"CTCS" has the meaning set forth in the introductory paragraph.

"Customer Lists" means all customer lists, customer files, customer sales records and other customer records and information maintained by the Companies.

"Cut-Off Date" has the meaning set forth in Section 12.1.1.

"<u>Debt Documents</u>" means appropriate and customary lender presentations, business and financial projections, and similar documents and materials (including assistance in creating usual and customary "public versions" of the foregoing), in connection with the Debt Financing.

"Debt Financing" means the debt financing contemplated to be consummated by the Incremental Amendment.

"Debt Lenders" means one or more lenders and the the administrative agent and collateral agent that have executed and delivered an Incremental Amendment (as herein defined), together with their respective Affiliates, and their respective Affiliates' officers, directors, employees, accountants, attorneys, agents and representatives and their respective successors and assigns.

"Deductible" means \$250,000.

"Defense Assumption Notice" has the meaning set forth in Section 12.5.2.

"Deposit Amount" has the meaning set forth in Section 5.7.2.

"Employee Benefit Plan" means each pension, retirement, profit sharing, deferred compensation, bonus, employee stock ownership, stock option, stock appreciation right, restricted stock, profits interest, phantom equity, other equity-based compensation, stock purchase, change in control, severance pay, salary contribution, retention agreement, commission, benefits continuation, severance, salary continuation, tuition reimbursement, dependent care assistance, legal assistance, fringe benefit (cash or non-cash), vacation, holiday, sick leave, cafeteria, health, medical, dental, vision, hearing, accident, disability, life, incentive, insurance, death benefit, unemployment, termination, bonus, deferred compensation, profit sharing, pension, retirement, group insurance, any employee benefit plan" as that term is defined in Section 3(3) of ERISA, and any "nonqualified deferred compensation plan" within the meaning Section 409A of the Code, but excluding government programs and statutory benefits in which Employees or Former Employees participate or under which any Company could have any liability, including, without limitation, joint and several liability in connection with a plan sponsored or maintained by an ERISA Affiliate, or to which an ERISA Affiliate contributes or has an obligation to contribute.

"Employee Bonus Amount" has the meaning set forth in Section 3.3.3.

4

"<u>Employees</u>" means all individuals (including common law employees), at any time of determination, who are employed by any Company in connection with the Business.

"Entity Company Parties" means CP, ASC, MCS, CR LLC and CTCS, and "Entity Company Party" means any of them.

"Environmental Laws" means all environmental, public health and safety Laws (including but not limited to the Occupational Safety and Health Act), codes and ordinances, whether foreign or domestic and whether national, federal, provincial, state or local, and all rules and regulations promulgated thereunder, including, without limitation, the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. § 9601 et seq.; the Resource Conservation and Recovery Act, 42 U.S.C. § 6901 et seq.; the Federal Water Pollution Control Act, 33 U.S.C. § 1251 et seq.; the Clean Air Act, 42 U.S.C. § 7401 et seq.; the Hazardous Materials Transportation Act, 49 U.S.C. § 1471 et seq.; the Toxic Substances Control Act, 15 U.S.C. § 2601 through 2629; the Oil Pollution Act, 33 U.S.C. § 2701 et seq.; the Emergency Planning and Community Right to Know Act, 42 U.S.C. § 11001 et seq.; and the Safe Drinking Water Act, 42 U.S.C. § 300f through 300j, and all Laws relating to emissions, discharges, releases or threatened releases of Hazardous Substances into the environment (including, without limitation, air, surface water, ground water, land surface or subsurface strata) or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of Hazardous Substances. The term "Environmental Laws" includes all state, local and municipal Laws dealing with the subject matter of the above listed federal statutes or promulgated by any governmental or quasi-governmental agency thereunder in order to carry out the purposes of any federal, state, local or municipal Law.

"Environmental Permits" means all permits, licenses, authorizations, certificates, approvals, consents, registrations, entitlements and other authorizations required by any Governmental Authority pursuant to any Environmental Law.

"Equipment Adjustment Amount" has the meaning set forth in Section 3.4.4.

"Equity Financing" means a public offering or private placement of equity or equity-linked securities of CPHI.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended and the rules and regulations promulgated thereunder.

"<u>ERISA Affiliate</u>" means, with respect to any entity, any corporation or trade or business (whether or not incorporated) which is or was, at the relevant time, treated with such entity as a single employer within the meaning of Section 414(b), (c), (m) or (o) of the Code.

"Escrow Agent" means JPMorgan Chase.

"Escrow Agreement" has the meaning set forth in Section 3.3.1.

"Escrow Amount" means \$1,292,000.00.

"Escrow Fund" has the meaning set forth in Section 12.7.3.

"Estimated Gross Employee Bonus Amount" has the meaning set forth in Section 3.3.3.

"Excluded Equipment" has the meaning set forth in Section 3.4.4.

5

"<u>Excluded Insurance Proceeds</u>" means proceeds actually received by Buyers or their Affiliates (including the Companies) as payment in respect of claims for coverage made by the Companies to Markel prior to date hereof in respect of damage to pumping equipment unit numbers 125 and 175.

"<u>Excluded Liabilities</u>" means all Liabilities arising out of or relating to the conduct of the Business, the ownership, lease or use of the properties or assets of the Companies, or the employment or engagement or use of Employees or Former Employees prior to the Closing Date. Without limiting the generality of the foregoing, "<u>Excluded Liabilities</u>" includes:

(a) all Liabilities in respect of (i) Pre-Closing Accounts Payable and (ii) credits and trade credits, if any, provided to customers, suppliers or vendors prior to the Closing Date for goods sold or services performed by the Companies prior to the Closing Date, in each case, only to the extent such Liabilities are not repaid or reimbursed by the Sellers in accordance with Section 9.6;

(b) all Liabilities arising out of, relating to or with respect to (i) the employment or performance of services, or termination of employment or services by any Company of any individual before the Closing Date, including, without limitation, the payment of any and all severance, salary, benefits in lieu of notice requirements, bonuses (including sale bonuses and change of control bonuses, but excluding amounts payable in pursuant to the Employee Bonus Plan), severance or termination obligations, and perquisites, and any payroll or other Taxes payable with respect to such payments, and (ii) workers' compensation claims against the Companies that relate to the period before the Closing Date, irrespective of whether such claims are made prior to or after the Closing;

(c) to the extent incurred prior to the Closing Date, Liabilities arising out of, under or in connection with all Employee Benefit Plans sponsored by any Company or their respective predecessors or with respect to which a Company or its respective predecessors has made or is required to make payments, transfers or contributions in respect of any Employees or Former Employees, directors, officers, equityholders, managers, consultants or independent contractors of any Company or their respective predecessors and all insurance policies, fiduciary liability policies, benefit administration contracts, actuarial contracts, trusts, escrows, surety bonds, letters of credit and other contracts primarily relating to any Employee Benefit Plan and any acts or omissions of any Company, any of their Affiliates, or any fiduciaries or trustees of any Employee Benefit Plan in connection with the operation or administration of any Employee Benefit Plan;

(d) all Liabilities owed by the Companies to Sellers and their respective Affiliates, other than pursuant to the Lease Agreements or the other Transaction Documents;

(e) all Liabilities owed by the Companies to any Person with any interest (whether past, present, future or contingent) in securities of any Company;

(f) all Liabilities arising out of any violation of Environmental Law by any Person (other than Buyers or their Affiliates) with respect to, or the presence of any Hazardous Materials on or under, the property located at 16113 North IH-35 Frontage Road, Salado, TX (the "Salado Property"); and

(g) all Liabilities arising at any time, out of, or in connection with, the Transferred Assets.

"Final Gross Employee Bonus Amount" has the meaning set forth in Section 3.7.1.

6

"Final Equipment Adjustment Amount" has the meaning set forth in Section 3.4.5.

"Final Equipment Adjustment Decision" has the meaning set forth in Section 3.4.5.

"Financial Statements" has the meaning set forth in Section 5.9.

"Former Employees" means all individuals (including common law employees) who have at any time previously been employed by any Company in connection with the Business.

"Fundamental Representations" has the meaning set forth in Section 12.1.1.

"GAAP" means United States generally accepted accounting principles, consistently applied, as in effect on the date of this Agreement.

"<u>Governmental Authority</u>" means any (a) federal, state, local, municipal, foreign or other government, (b) governmental or quasi-governmental authority or political subdivision of any nature (including any governmental agency, branch, division, department, official, or entity and any court or other tribunal) or (c) body exercising, or entitled to exercise any administrative, executive, judicial, legislative, police, regulatory, or taxing authority or power of any nature, including any arbitral tribunal, departments, courts, commissions, boards, bureaus, ministries, agencies or other instrumentalities of any of them.

"Governmental Order" means any order, judgment, injunction, decree, writ, stipulation, determination or award, in each case, entered by or with any Governmental Authority.

"<u>Hazardous Material</u>" means any (a) element, material, compound, mixture, solution, chemical, substance, or pollutant within the definition of "hazardous substance" pursuant to Section 101(14) of CERCLA, (b) toxic or hazardous materials or substance, (c) solid wastes, including asbestos, polychlorinated biphenyls, mercury, flammable or explosive materials, (d) hydrocarbons, petroleum, petroleum products or any fraction, byproduct or distillation product thereof, (e) asbestos, polychlorinated biphenyls, or any radioactive substances, (f) material, chemical, pollutant, contaminant, substance or waste regulated as hazardous waste by any jurisdiction or (g) any other material, chemical, pollutant, contaminant, substance or waste that is regulated or for which liability or standards of care are imposed pursuant to any Environmental Law.

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"Imputed Underpayment" means an "imputed underpayment" within the meaning of Section 6225 of the Code.

"Incremental Amendment" means an amendment to the Term Loan Agreement by and among CPHI, Concrete Pumping Intermediate Acquisition Corp., Buyers, certain subsidiaries of CPHI, the administrative agent and collateral agent named therein, and the lenders party thereto, and/or any other agreement, commitment, or instrument entered into by CPHI, certain of its subsidiaries, or any of their Affiliates and certain lenders or other financial institutions party thereto to provide for debt financing to consummate the transactions contemplated hereby, as the same may be amended, supplemented or otherwise modified from time to time in accordance with its terms.

7

"Indebtedness" means, without duplication, (a) all obligations of the Companies for borrowed money and any accrued interest or prepayment premiums related thereto, (b) all obligations as lessee or lessees under leases that have been recorded as capital leases in accordance with GAAP, (c) all obligations represented by notes payable or drafts accepted that evidence extensions of credit or evidenced by bonds, debentures, notes or similar instruments, (d) all reimbursement obligations (including without limitation all contingent reimbursement obligations) in connection with letters of credit, standby letters of credit (other than the standby letter of credit referred to in <u>Section 3.2</u>) or guaranties of letters of credit, (e) all guarantees, performance bonds, sureties and/or similar obligations of any kind or nature issued by or on behalf of a Company in connection with any customer Contracts, proposals or otherwise, and (f) all intercompany accounts, payables or loans of any kind or nature.

"Indemnified Covenants" has the meaning set forth in Section 12.1.2.

"Indemnified Taxes" means (a) any and all Taxes of any Seller (and if any Seller is disregarded for U.S. federal income tax purposes, such Seller's regarded owner) for any Tax period, (b) any and all Taxes imposed on or payable by any Company, or for which any Company otherwise may be liable, for any Pre Closing Period; (c) any and all Taxes imposed on or payable by any Company, or for which any Company otherwise may be liable, by reason of being (or ceasing to be) or having been a member of an Affiliated Group prior to the Closing Date (or being included (or required to be included)) in any Tax Return relating thereto; (d) any and all liabilities of any Company for any Taxes by reason of a Tax sharing, Tax indemnity or similar agreement entered into prior to the Closing Date or by reason of transferee or successor liability arising in respect of a transaction undertaken prior to the Closing Date; (e) with respect to any Straddle Period, Taxes imposed on or payable by any Company that are allocable, pursuant to <u>Section 9.4.2</u>, to the portion of such period prior to the Closing Date; (f) the employer's portion of any payroll Taxes arising from any compensatory payments made prior to or in connection with the transactions contemplated by this Agreement; and (g) all costs and expenses of contesting any audit or Tax Contest that could result in the imposition of Indemnified Taxes.

"Indemnitee" has the meaning set forth in Section 12.3.

"Indemnitor" means the party required to provide indemnification pursuant to Article 12.

"Inspected Equipment" has the meaning set forth in Section 3.4.1.

"Inspection Buyer Representative" has the meaning set forth in Section 3.4.4.

"Inspection Event" has the meaning set forth in Section 3.4.1.

"Inspection" has the meaning set forth in Section 3.4.1.

"Inspection Report" has the meaning set forth in Section 3.4.2.

"Inspection Seller Representative" has the meaning set forth in Section 3.4.4.

"<u>Insured Value</u>" means the value of an item of Inspected Equipment, as designated under the insurance policies insuring such Inspected Equipment as of April 1, 2019, a true, correct and complete list of Insured Values of Inspected Equipment has been delivered to Buyers.

8

"Intellectual Property" means any of the following worldwide: (a) patents and patent applications (including utility and design patents), (b) registered and unregistered trademarks, service marks trade dress and trade names, pending trademark and service mark registration applications, and intent-to-use registrations or similar reservations of marks, (c) registered and unregistered copyrights, and applications for registration of copyright, slogans, and all applicable moral rights associated with any of the foregoing, (d) internet domain names and URLs, (e) trade secrets (including business and technical documents), know-how, inventions (patentable or not), and other proprietary rights and (f) rights in computer software.

"Interim Financial Statements" has the meaning set forth in Section 5.9.

"IT Assets" has the meaning set forth in Section 5.21.3.

"Law" means all provisions of constitutions, statutes, rules, regulations, and orders of all Governmental Authorities applicable to such Person, and all orders, decrees, judgments, injunctions and writs of all courts, arbitrators and Governmental Authorities.

"<u>Lease Agreement</u>" means, with respect to each real property location described on <u>Annex A</u>, a lease agreement, between a Buyer or an Affiliate of a Buyer, as lessee, and the Real Property Owner of such premises, as lessor, for the lease, occupancy and use of such premises, with rental rates matching current rental rates for such real properties or as otherwise agreed to by the parties, and in form and substance acceptable to Buyers and Sellers, to be executed and delivered at Closing.

"Lease Termination and Release" means that certain Lease Termination and Release, to be dated as of the Closing Date, in form and substance acceptable to Buyers.

"Leased Real Property" means all real property, land, buildings, fixtures, improvements and structures leased or subleased by any Company and all rights and interests of any Company, which is more particularly described on <u>Schedule 5.19</u>.

"<u>Liabilities</u>" means any and all debts, liabilities, obligations, duties and responsibilities of any kind and description, whether absolute or contingent, monetary or non-monetary, direct or indirect, known or unknown or matured, unmatured, or of any other nature (whether or not known or capable of being known, disclosed, matured accrued, asserted, contingent, direct, conditional, implied, vacarious, derivative, joint, several or secondary).

"Licenses and Permits" means, with respect to any Person, all of the permits, licenses, certificates, approvals, consents, waivers, exemptions and authorizations of, and registrations with all Governmental Authorities, and under, all federal, state, local and foreign Laws, as are required for the operation of such Person's business, or the ownership and use of such Person's assets or Real Property, or the employment of such Person's employees.

"<u>Liens</u>" means any mortgages, security interests, servitudes, deeds of trust, liens, charges, pledges, equitable interest, options, conditional sales contracts, rights of purchase, claims, restrictions, covenants, easements, rights of way, title defects or other lien or encumbrances or restrictions of any nature (whether absolute or contingent).

"Limited Representations" has the meaning set forth in Section 12.1.1.

"Liquidated Damages" has the meaning set forth in Section 11.3.

"Losses" means any and all losses, liabilities, damages, deficiencies, obligations, fines, expenses, claims, demands, actions, suits, proceedings, judgments or settlements, including interest and penalties recovered by a third party with respect thereto and out-of-pocket expenses and attorneys' and accountants' fees and expenses incurred in the investigation or defense of any of the same or in asserting, preserving or enforcing any of the Indemnitee's rights hereunder, suffered by an Indemnitee; provided, however, Losses shall not include diminution in value, punitive damages or opportunity costs, unless such punitive damages or opportunity costs are actually awarded in connection with a Third Party Claim.

9

"<u>Material Adverse Effect</u>" means, any event, change, development, effect, occurrence that has, or would reasonably be expected to have, individually or in the aggregate, (a) a material adverse effect on the business, assets, properties, results of operations, liabilities or condition (financial or otherwise) of the Companies or (b) have a material adverse effect on the ability of any Company Party to enter into or perform its obligations under, or to consummate the transactions contemplated by, this Agreement or any other Transaction Document; provided, however, that in no event will any of the following (or the effect of any of the following), alone or in combination, be deemed to constitute, or be taken into account in determining whether there has been or will be, a Material Adverse Effect: (i) any change in Law, regulatory policies, accounting standards or principles (including GAAP) or any guidance relating thereto or interpretation thereof, (ii) any change in the economic, political, business or financial market conditions generally, (iii) any change generally affecting any of the industries in which the Companies operate or the economy as a whole, (iv) any natural disaster, (v) any acts of terrorism, sabotage, war, the outbreak or escalation of hostilities, weather conditions, change in geopolitical conditions or other force majeure events or; except, in the case of <u>clauses (i)</u>, (ii) and (iii) above, to the extent that any such change, condition, event or effect has a materially disproportionate and adverse effect on (A) business, assets, properties, results of operations, liabilities or condition (financial or otherwise) of the Companies relative to other businesses in the industry in which the Companies operate, taken as a whole, or (B) on the ability of the Company Parties to enter into, to perform their obligations under, or to consummate the transactions contemplated by, this Agreement and the other Transaction Documents.

"<u>MCS</u>" has the meaning set forth in the introductory paragraph.

"<u>MCS Membership Interest</u>" has the meaning set forth in the Recitals.

"<u>Mini Basket</u>" has the meaning set forth in <u>Section 12.4.3</u>.

"<u>Organizational Documents</u>" means (a) in the case of a Person that is a corporation, its articles or certificate of incorporation and its by-laws, regulations or similar governing instruments required by the laws of its jurisdiction of formation or organization; (b) in the case of a Person that is a partnership, its articles or certificate of partnership, formation or association, and its partnership agreement (in each case, limited, limited liability, general or otherwise); (c) in the case of a Person that is a limited liability company, its articles or certificate of formation or organization, and its limited liability company agreement or operating agreement; and (d) in the case of a Person that is none of a corporation, partnership (limited, limited liability, general or otherwise), limited liability company or natural person, its governing instruments as required or contemplated by the laws of its jurisdiction of organization.

"Outstanding Claims" has the meaning set forth in Section 12.7.3.

"Owned Intellectual Property" means all Intellectual Property owned or purported to be owned by the Companies.

"<u>Owned Real Property</u>" means all real property, land, buildings, fixtures, improvements and structures owned by any Company Party or Real Property Owner, and all rights and interests of such Company Party or Real Property Owner related thereto, which is more particularly described on <u>Schedule 5.18</u>.

10

"<u>Partnership Tax Audit Rules</u>" means Sections 6221 through 6241 of the Code, as amended by the Bipartisan Budget Act of 2015, together with any guidance issued thereunder or successor provisions and any similar provision of state or local Law.

"Payoff Amounts" has the meaning set forth in Section 3.3.3.

"Payoff Letters" has the meaning set forth in Section3.2.

"Payor" has the meaning set forth in Section 9.3.

"<u>Permitted Lien</u>" means (a) Liens for Taxes not yet due and payable and (b) mechanics, carriers', workmen's, repairmen's or other like Liens arising or incurred in the ordinary course of business consistent with past practices or amounts that are not delinquent and which are not, individually or in the aggregate, material to the business of the Companies.

"<u>Person</u>" means any individual, firm, corporation (including any not-for-profit corporation), general or limited partnership, limited liability partnership, limited liability company, incorporated or unincorporated association, joint venture, joint stock company, governmental agency, trust, firm, organization, instrumentality or other entity of any kind.

"<u>Personnel Records</u>" means all personnel and other records (to the extent permitted by Law) for Employees, and all manuals, books and records of the Companies, including, without limitation, personnel policies and files and manuals and accounting records.

"<u>Post-Closing Period</u>" means any taxable period beginning on the Closing Date and, with respect to a Straddle Period, the portion of such Straddle Period beginning on the Closing Date.

"<u>Pre-Closing Accounts Payable</u>" means any account payable of any Company attributable to the period prior to the Closing Date, including, but not limited to, (a) accounts payable arising in connection with the ordinary course of the Business, and (b) the fees and expenses payable to advisors and other professional service firms incurred by the Company Parties in connection with the preparation, negotiation and execution of the Transaction Documents and the consummation of the transactions contemplated thereby.

"<u>Pre-Closing Accounts Receivable</u>" means any account receivable of any Company attributable to the period prior the Closing Date for which a Company has issued an invoice as of the Closing Date.

"<u>Pre-Closing Period</u>" means any taxable period ending before the Closing Date and, with respect to a Straddle Period, the portion of such Straddle Period ending on the day immediately prior to the Closing Date.

"Pre-Closing Warranty Claims" means warranty claims made by Companies prior to the Closing Date.

"<u>Prepaid Insurance Amount</u>" means that portion of the insurance premiums paid in advance by the Companies with respect to periods of insurance coverage on and after the Closing Date.

"Property Taxes" means real, personal and intangible property Taxes and any other Taxes imposed on a periodic basis and measured by the value of any item.

"PTO and Payroll/Commission Amount" has the meaning set forth in Section 5.14.12.

11

"<u>Pumping Equipment</u>" means all trucks, concrete pumps and equipment of the Companies, together with all maintenance records, spare parts, spare and replacement parts and accessories related thereto, in each case, whether new or used.

"Purchase Price" means \$129,200,000.

"Purchased Assets" means the properties and assets of the Companies, but excluding the Transferred Assets.

"<u>Purchased Interests</u>" means the CP Limited Partnership Interests, the ASC Limited Partnership Interests, the CP General Partnership Interests, the ASC General Partnership Interests and the MCS Membership Interests.

"R&W Insurance Policy" has the meaning set forth in Section 8.2.

"<u>Real Property</u>" means all interests in real property, land, buildings, fixtures, improvements and structures, owned, leased licensed or used in connection with the ownership, use or operation of the Business, or any assets used in the Business, of the Companies, including the real property described in the Lease Agreements.

"<u>Real Property Owners</u>" means CLC 31, LLC, a Texas limited liability company, and A. Keith Crawford, and "<u>Real Property</u> <u>Owner</u>" means either of them.

"<u>Required Financial Information</u>" means the financial information regarding the Companies that is required to be included in the Proxy Statement or other SEC filing by CPHI in connection with the Equity Financing, including (a) all audited financial statements (and notes thereto) of the Companies for the preceding two fiscal years, prepared in compliance with Regulation S-X and accompanied by an unqualified audit report of Maxwell Locke & Ritter LLP ("<u>Companies' Audit Firm</u>") with respect thereto (including all required audited financial statements of businesses acquired by the Companies accompanied by an unqualified audit report from the Companies Audit Firm with respect thereto), (b) all unaudited interim financial statements (and notes thereto) of the Companies prepared in compliance with Regulation S-X and having been reviewed by the Companies' Audit Firm which shall have advised the Companies in writing (e-mail being sufficient) that it has no further comments thereon, including, without limitation (i) the interim financial statements for the fiscal quarters ended March 31, 2019 and 2018 and (ii) the Companies' Mudit Firm providing such firm's consent to the inclusion of such firm's audit report within the Proxy Statement or other SEC filing by CPHI in connection with the Equity Financing and (d) all other financial and business information regarding the Companies reasonably required by the CPHI and Buyers in connection with the Debt Financing, the Equity Financing and the Proxy Statement.

"<u>Restricted Period</u>" has the meaning set forth in <u>Section 10.2</u>.

"<u>Restrictive Covenant Agreements</u>" means restrictive covenant agreements, between CP, on the one hand, and each of Mike Schultz, Ariel Fuentes and Troy A. Casada, on the other hand, on terms acceptable to Buyers.

"Retained Escrow Amount" has the meaning set forth in Section 12.7.3.

12

"Run-off Period" has the meaning set forth in Section 9.6.

"SEC" means the United States Securities and Exchange Commission.

"Seller Indemnitee" has the meaning set forth in Section 12.3.

"Seller Returns" has the meaning set forth in Section 9.4.3(a).

"Sellers" mean CRLLC, CTCS, A. Keith Crawford and Melinda Crawford, and "Seller" means any one of them.

"Straddle Period" means any taxable year or Tax period beginning before and ending on or after the Closing Date.

"Subsidiary" means, with respect to a Person, a corporation or other entity of which more than 50% of the voting power of the equity securities or equity interests is owned, directly or indirectly, by such Person.

"Tax" or "Taxes" means any and all taxes, however denominated, the liability for which is imposed by Law, which taxes shall include, but not be limited to, all net income, gross income, gross receipts, franchise, excise, occupation, estimated, alternative minimum, add on minimum, premium, windfall profit, profits, gains, net worth, paid up capital, capital stock, greenmail, sales, use, ad valorem, value added, motor vehicle sales, retailers' occupation, stamp, natural resources, environmental, real property, personal property, custom, duty, transfer, recording, escheat, license, registration, documentation, leasing, insurance, social security, employment, severance, workers' compensation, impact, hospital, health, unemployment, disability, payroll, license, service, service use, employee or other withholding, or other tax or governmental charge, of any kind whatsoever, including any interest, penalties, fees, charges, levies, assessments, duties, tariffs, imposts or additions to tax that may become payable in respect thereof, and any liability in respect of such amounts arising as a result of being a member of any affiliated, consolidated, combined, unitary or similar group, as a successor to or transferee of another Person or by Contract.

"Tax Arbiter" has the meaning set forth in Section 9.4.3(a).

"Tax Contest" means any audit, investigation, claim, litigation, assessment, reassessment, dispute or controversy relating to Taxes.

"<u>Tax Return</u>" means any report, return, document, declaration or other information or filing required to be supplied to any taxing authority or jurisdiction (foreign or domestic) with respect to Taxes including, without limitation, information returns, any documents with respect to or accompanying payments of estimated Taxes, or with respect to or accompanying requests for the extension of time in which to file any such report, return, document, declaration or other information, or any amendment to any of the foregoing.

"Taxing Authority" means any Governmental Authority responsible for the administration, assessment, determination, collection or the imposition of any Tax.

"<u>Term Loan Agreement</u>" means that certain Term Loan Agreement, dated as of December 6, 2018, by and among Credit Suisse AG, Cayman Islands Branch, as administrative agent, CPHI, Concrete Pumping Intermediate Acquisition Corp., BBCP, certain subsidiaries of CPHI, Credit Suisse Loan Funding, LLC, Jefferies Finance, LLC, and Stifel Nicolaus & Company Incorporated LLC, as Joint Lead Arrangers and Joint Bookrunners, as the same may be amended, modified, or supplemented in accordance with its terms.

13

"Terminating Buyer Breach" has the meaning set forth in Section 11.1.3(a).

"Terminating Company Breach" has the meaning set forth in Section 11.1.2(b).

"Termination Date" has the meaning set forth in Section 11.1.2(c).

"Territory" has the meaning set forth in Section 10.2.

"Third Party Claim" has the meaning set forth in Section 12.5.2.

"Third Party Lease" means any Contract pursuant to which any Company leases, licenses or has the right to use real property and the lessor is a Person other than a Real Property Owner.

"<u>Transaction Documents</u>" means (a) this Agreement, (b) the Bill of Sale, (c) the Escrow Agreement, (d) the Restrictive Covenant Agreements, (e) the Transition Services Agreement, (f) the Lease Agreements and (g) the Lease Termination and Releases.

"<u>Transfer Taxes</u>" means all federal, state, local and non-U.S. transfer, excise, sales, use, ad valorem, value added, motor vehicle sales, license, registration, stamp, recording, real and personal property transfer and similar Taxes or fees applicable to, imposed upon, or arising out of the transfer of the transactions contemplated herein.

"Transferred Assets" has the meaning set forth in Section 2.2.

"<u>Transition Services Agreement</u>" means that certain Transition Services Agreement, to be entered into between Buyers and the Companies, on the one hand, and Sellers, on the other hand, in form and substance acceptable to Buyers and the Sellers.

"Vehicles" means all vehicles, and service vehicles of the Companies, together with all maintenance records, spare parts, replacement parts and accessories related thereto.

"WARN Act" has the meaning set forth in Section 5.14.7.

"Workers Compensation Claims" has the meaning set forth in Section 5.14.6.

"Warrant Exchange" means an offer by CPHI to exchange shares of CPHI common stock for outstanding warrants to purchase shares of CPHI common stock.

1.2 Construction.

1.2.1 Unless the context of this Agreement otherwise requires, (a) words of any gender include each other gender, (b) words using the singular or plural number also include the plural or singular number, respectively, (c) the terms "hereof," "herein," "hereby," "hereto" and derivative or similar words refer to this entire Agreement, (d) the terms "Article", "Section", "Schedule" or "Annex" refer to the specified Article or Section of, or Schedule or Annex to, this Agreement, (e) the word "including" shall mean "including, without limitation," (f) the word "or" shall be disjunctive but not exclusive, and (g) for purposes of <u>ARTICLE 5</u> only, the word "delivered" shall mean delivered to Buyer or Buyers' attorneys or uploaded to the Firmex VDR established for this transaction at 11:59 p.m. (CT) on the Business Day prior to the date hereof.

14

1.2.2 Unless the context of this Agreement otherwise requires, references to agreements and other documents shall be deemed to include all subsequent amendments and other modifications thereto.

1.2.3 Unless the context of this Agreement otherwise requires, references to statutes shall include all rules and regulations promulgated thereunder.

1.2.4 The language used in this Agreement shall be deemed to be the language chosen jointly by the parties to express their mutual intent and no rule of strict construction shall be applied against any party.

1.2.5 Whenever this Agreement refers to a number of days, such number shall refer to calendar days unless Business Days are specified.

1.2.6 The phrase "to the extent" shall mean the degree to which a subject or other thing extends, and such phrase shall not mean simply "if".

GAAP.

1.2.7 All accounting terms used herein and not expressly defined herein shall have the meanings given to them under

1.2.8 All amounts payable pursuant to this Agreement shall be paid in U.S. dollars, and all references to "\$" or "dollars" shall mean the lawful currency of the United States of America.

ARTICLE 2 PURCHASE AND SALE

2.1 Purchase and Sale of Purchased Interests. In accordance with and subject to the terms and conditions set forth in this Agreement, at the Closing, (a) CR LLC shall sell, assign, transfer, convey to CPHA LLC, and CPHA LLC shall purchase and accept, the CP General Partnership Interest (b) A. Keith Crawford shall sell, assign, transfer, convey to BBCP, and BBCP shall purchase and accept, the CP Limited Partnership Interest, (c) CTCS shall sell, assign, transfer, convey to CPHA LLC, and CPHA LLC shall purchase and accept, the ASC General Partnership Interest, (d) A. Keith Crawford shall sell, assign, transfer, convey to BBCP, and BBCP shall purchase and accept, the ASC General Partnership Interest, (d) A. Keith Crawford shall sell, assign, transfer, convey to BBCP, and BBCP shall purchase and accept, the ASC Limited Partnership Interest, and (e) Melinda Crawford shall sell, assign, transfer, convey to BBCP, and BBCP shall BBCP shall purchase and accept, the MCS Membership Interest, in each case, free and clear of all Liens, for the consideration specified in Section 3.1.

2.2 Transferred Assets; Distribution of Cash. Immediately prior to the Closing, the Companies shall transfer, assign, convey and deliver to CTCS, and CTCS shall accept from the Companies, all of the Companies' right, title, or interest in to, and under (a) all personal tangible property (including computers, printers, software, copiers, fax machines, office supplies and other similar tangible property and software licenses, to the extent such license are applicable only to such personal property) used in the operation of the Business that is ordinarily located at 320 Steck Avenue, Unit 210 and 220, Austin, TX, (b) that certain Contract 003901 and Mediation Agreement between CP and Globequest Travel Club represented by Hotel Reservations Center Limited, (c) the mobile phone numbers of Sellers and their immediate family members; (d) the Excluded Equipment and (e) the two scanners associated with the global search system and currently located at the 9907 Iota Drive and 9501 McKenzie Road offices (collectively, the "Transferred Assets"). The Companies shall convey the Transferred Assets on an AS-IS WHERE-IS basis with all faults pursuant to a Bill of Sale. Immediately prior to the Closing, the Companies shall make a distribution of all of the Companies' cash held as of such date to the Sellers in accordance with each Company's Organizational Documents.

15

2.3 Partner, Member and Manager Acknowledgment. Upon the consummation of the transactions contemplated herein, (a) Melinda Crawford hereby resigns as the manager of MCS, and acknowledges and agrees that she will cease to be a party to the MCS Operating Agreement, and shall cease to be the "tax matters partner" of MCS (and, if applicable, an officer of MCS), (b) A. Keith Crawford and CR LLC hereby withdraw as partners of CP, and (c) A. Keith Crawford and CTCS hereby withdraw as partners of ASC.

ARTICLE 3 PURCHASE PRICE

3.1 Purchase Price. Subject to adjustment in accordance with this <u>Article 3</u> and <u>Article 12</u>, the aggregate consideration for the Purchased Interests shall be equal to the Purchase Price.

3.2 Payoff Letters; Lien Releases. At least two Business Days prior to the Closing Date, (a) an authorized officer of the Companies shall deliver to Buyers the pay-off letters in form reasonably satisfactory to Buyers to be executed at or prior to the Closing by the lenders of all of the Indebtedness of the Companies, including, without limitation, as set forth on <u>Schedule 3.2</u>, which shall include an undertaking by such lenders to discharge, terminate and release in full at the Closing any Lien securing such Indebtedness (such letters, the "<u>Payoff Letters</u>"), and (b) the Company Parties shall have made arrangements satisfactory to Buyers for the full and final payment and satisfaction of all Indebtedness set forth on <u>Schedule 3.2</u> and the release of Liens securing such Indebtedness. Notwithstanding the foregoing, the Companies shall not pay-off or terminate the Irrevocable Standby Letter of Credit No. CTCS-965184, issued in favor of Texas Mutual Insurance Company unless otherwise agreed to by the parties prior to the Closing.

3.3 Payment of Purchase Price. At the Closing, Buyers shall pay and satisfy the Purchase Price, subject to adjustment in accordance with this <u>Article 3</u>, as follows:

3.3.1 Buyers shall pay or cause to be paid, by wire transfer of immediately available funds, the Escrow Amount to the Escrow Agent to be held as security in respect of the Sellers' indemnification obligations pursuant to <u>Article 12</u> in accordance with the escrow agreement (the "<u>Escrow Agreement</u>") to be entered into by and among the Sellers, Buyers and the Escrow Agent at the Closing on terms mutually acceptable to them.

3.3.2 Buyers shall pay or cause to be paid, on behalf of the Company Parties, by wire transfer of immediately available funds, the amounts set forth in each Payoff Letter to the accounts set forth in such Payoff Letters (such amounts, in the aggregate, are the "Payoff Amounts").

3.3.3 After Closing, Buyers shall withhold and retain from the Purchase Price an amount equal to \$3,700,000 (the "<u>Employee Bonus Amount</u>") <u>plus</u> an amount equal to the employer portion of withholding taxes (including FICA and FUTA) and the employer portion of any employer contribution under Employee Benefit Plans in respect thereof (all of the foregoing being the "<u>Estimated Gross Employee Bonus Amount</u>"), and Buyers shall cause the Companies to pay and remit the Employee Bonus Amount in accordance with <u>Section 3.7</u>.

16

3.3.4 Buyers shall pay or cause to be paid, by wire transfer of immediately available funds, to the Sellers to the account set forth on <u>Annex B</u>, the Purchase Price <u>plus</u> (a) the Deposit Amount <u>minus</u> (b) the Escrow Amount, <u>minus</u> (c) the Payoff Amounts, <u>minus</u> (d) the Estimated Gross Employee Bonus Amount, <u>minus</u> (e) the PTO and Payroll/Commission Amount, <u>minus</u> (f) the Equipment Adjustment Amount and <u>minus</u> (g) 50% of the fees of the Escrow Agent, which Buyers shall cause to be paid to the Escrow Agent pursuant to the Escrow Agreement. The Sellers (i) shall agree among themselves as to the allocation and distribution of the such payment, and shall be solely responsible for distributing such amount to Persons entitled to a portion thereof, and (ii) acknowledge that upon Buyer's payment of such amount to the account set forth on <u>Annex B</u>, Buyer's obligations pursuant to this <u>Section 3.3.4</u> shall be fully satisfied and release the Buyers from any obligation or responsibility with respect to the payment of the Purchase Price to the Sellers pursuant to this <u>Section 3.3.4</u>. Notwithstanding anything contained herein to the contrary, for purposes of this <u>Section 3.3.4</u>, the Deposit Amount shall not exceed \$40,000.

3.4 Equipment Inspection.

3.4.1 Between the date hereof and Closing, Buyer will complete physical inspections of any Pumping Equipment and Vehicles with an Insured Value in excess of \$100,000 (together, the "Inspected Equipment") to confirm the compliance and conformity of such Inspected Equipment with Section 5.8 (the "Inspection"). The Inspection will occur during on weekends (each, an "Inspection Event") at such time(s) as mutually agreed to by the parties. All parties hereby agreeing to act reasonably and in good faith in scheduling and performing such Inspections.

3.4.2 Within three (3) days following each Inspection Event, Buyer will deliver to Sellers a report (each, an "Inspection Report") (a) identifying in reasonable detail any Significant Defects discovered during such Inspection Event and (b) providing a good faith estimate of the cost of repairing such Significant Defects. For purposes of this Section 3.4, "Significant Defects" shall mean functional items which are in excess of \$5,000 per piece of Inspected Equipment that materially impact the Inspected Equipment's ability to safely and effectively perform its primary operational functions. Significant Defects excludes normal wear and tear items for similar Inspected Equipment of equivalent age, and any minor cosmetic items that do not adversely affect the safety and functionality of such Inspected Equipment.

3.4.3 The Sellers will review the Inspection Report and accept or reject the proposed repairs of the Significant Defects as to each item of Inspected Equipment. If Sellers accept the proposed repairs, Sellers will repair and resolve the identified Significant Defects prior to the Closing Date, at the Sellers' sole cost and expense. Sellers shall deliver a written summary and certification covering any such repairs to Buyer at least three Business Days prior to Closing, which shall complete Sellers' obligation with respect to such repairs and the Inspected Equipment under this <u>Section 3.4</u>. In the event any accepted repairs are unable to be completed prior to the Closing Date, the cost of such repairs, as reflected on the Inspection Report, shall be deducted from the Purchase Price at Closing, and Buyer shall thereafter be responsible for completing such repairs on or after the Closing Date.

3.4.4 Prior to Closing, Keith Crawford, for and on behalf of the Sellers (the "Inspection Seller Representative") and Bruce Young, for and on behalf of Buyer (as "Inspection Buyer Representative"), shall work diligently and in good faith to resolve any of the Significant Defects or proposed repairs of such Significant Defects rejected by Sellers. If Inspection Seller Representative and Inspection Buyer Representative are unable to achieve such resolution at least three Business Days prior to Closing, the Insured Value of the item of Inspected Equipment subject to dispute shall be deducted from the Purchase Price (such amount is the "Equipment Adjustment Amount") and Sellers shall retain any such Inspected Equipment (as Transferred Assets, the "Excluded Equipment") and may subsequently sell such Excluded Equipment without restriction within 270 days following the Closing Date.

17

3.4.5 The Equipment Adjustment Amount shall reduce the Purchase Price on the Closing Date, dollar for dollar.

3.5 Allocation of Purchase Price. The parties agree that the Purchase Price shall be allocated as set forth on <u>Schedule 3.5</u>. The Sellers shall agree, among themselves, as to the allocation and distribution of the Purchase Price, and the Sellers shall be solely responsible for distributing the Purchase Price to Persons entitled to a portion thereof. Subject to the foregoing, Buyers shall prepare and deliver Internal Revenue Service Form 8594 to the Sellers for their review within 45 calendar days after the Closing Date, with a separate Form 8594 for each Company which Forms 8594 shall include the allocations on <u>Schedule 3.5</u> and take into account any adjustments to the Purchase Price as set forth in <u>Article 3</u>, and <u>Article 12</u>. Upon the agreement of all parties as to the Form 8594, it shall be timely filed with the Internal Revenue Service by all parties and the parties shall file all Tax Returns in a manner consistent with such allocation.

3.6 Withholding. Notwithstanding anything in this Agreement to the contrary, Buyers may deduct and withhold from any consideration otherwise payable to any Person pursuant to this Agreement such amounts as it is required to deduct and withhold with respect to the making of such payment under the Code. Any such amounts will be treated for all purposes of this Agreement as having been paid to the applicable Person in respect to which such deduction and withholding was made.

3.7 Employee Bonus Amount.

3.7.1 After Closing, subject to appropriate withholding, Buyers shall cause the Companies to pay with the first payroll immediately following the Closing Date, in accordance with normal payroll practices of the Companies, the Employee Bonus Amount to the Employees identified on <u>Schedule 3.7</u> (which schedule shall be delivered to Buyers at least five Business Days prior to Closing), in the amounts set forth opposite their names; <u>provided</u>, <u>however</u>, that the aggregate amount of such payments shall not exceed \$3,700,000. Buyer shall cause the Companies to (a) withhold and remit to the appropriate taxing authorities the employee's portion of all applicable withholding taxes related thereto, (b) withhold and remit to the appropriate taxing authorities the employer's portion of all applicable withholding taxes related thereto, (c) withhold the employee's contributions to Employee Benefit Plans with respect to such payments and (d) remit to the Employee Benefit Plans the employer's contributions with respect to such payments. The sum of the Employee Bonus Amount and the Employer's amounts in <u>clauses (b)</u> and (d) of the immediately preceding sentence is the "<u>Final Gross</u> Employee Bonus Amount."

3.7.2 If the Estimated Gross Employee Bonus Amount is *greater than* the Final Gross Employee Bonus Amount, Buyers shall pay to the Sellers, not later than 10 Business Days after the determination of such amount, such excess to the account set forth on <u>Annex B</u>. The Sellers (a) shall agree among themselves as to the allocation and distribution of the such payment, and shall be solely responsible for distributing such amount to Persons entitled to a portion thereof, and (b) acknowledge that upon the Companies' payment of such amount to the account set forth on <u>Annex B</u>, Buyers and the Companies' obligations pursuant to this <u>Section 3.7.2</u> shall be fully satisfied.

3.7.3 If the Estimated Gross Employee Bonus Amount is *less than* the Final Gross Employee Bonus Amount, the Sellers shall pay to Buyers, not later than 10 Business Days after determination of such amount, such shortfall to an account designated in writing by the Buyers.

18

ARTICLE 4 CLOSING

4.1 Closing. Subject to the terms and conditions of this Agreement, the closing of the transactions contemplated by this Agreement (the "<u>Closing</u>") shall take place at the offices of Ballard Spahr LLP, 1225 17th Street, Suite 2300, Denver, Colorado 80202, at 10:00 a.m. (Denver time) two Business Days after the date on which all conditions set forth in this <u>Article 4</u> have been satisfied or waived in writing (other than those conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction or waiver of such conditions) or such other time and place as the parties may mutually agree. The date on which the Closing actually occurs is the "<u>Closing Date</u>."

4.2 Buyers' Conditions to Closing. The obligation of Buyers to consummate the transactions contemplated by this Agreement is subject to the satisfaction (or Buyers' written waiver) of the following conditions as of the Closing Date:

4.2.1 The representations and warranties set forth in <u>Article 5</u> shall be true and correct at and as of the Closing Date as though then made (other than those representations and warranties that address matters as of particular dates which shall be true and correct at and as of such particular dates), except for inaccuracies that are de minimis in amount and effect, disregarding, solely for such purpose, all qualifications contained herein regarding materiality or Material Adverse Effect;

4.2.2 The Company Parties shall have performed in all material respects all of the covenants and agreements required to be performed by them under this Agreement at or prior to the Closing;

4.2.3 The consents and approvals required to be obtained by the Company Parties to effect the transactions contemplated hereby, including those set forth on <u>Schedule 5.3</u>, shall have been obtained and delivered to Buyers;

4.2.4 Since the date of this Agreement, there shall not have occurred a Material Adverse Effect on the Business or any Company;

4.2.5 All of the conditions to closing the Equity Financing shall have occurred or been waived by the applicable parties, and the funding of the Equity Financing shall have occurred or shall occur substantially concurrently with the consummation of the transactions contemplated by this Agreement;

4.2.6 All of the conditions to each of (i) closing the Debt Financing and (ii) consummating the transactions contemplated by this Agreement as a "Permitted Acquisition" under and as defined in the ABL Agreement and the Term Loan Agreement shall have occurred or been waived by the applicable parties; and the funding under the Incremental Amendment shall have occurred or shall occur substantially concurrently with the consummation of the transactions contemplated by this Agreement;

4.2.7 The transactions contemplated in the Bill of Sale shall have occurred, and a duly executed copy of the Bill of Sale shall have been delivered to Buyers;

4.2.8 The applicable waiting periods, if any, under the HSR Act shall have expired or been terminated;

19

- 4.2.9 No judgment, decree or order shall have been entered which would prevent the consummation of the Closing;
- 4.2.10 The Company Parties shall have delivered to Buyers each of the following:

(a) a certificate of each Company Party in the form reasonably satisfactory to Buyers, dated as of the Closing Date, stating that the preconditions specified in <u>Sections 4.2.1</u> and <u>4.2.2</u> have been satisfied, duly executed by, (i) in the case of Company Parties that are limited partnerships, the general partner, (ii) in the case of Company Parties that are limited liability companies, the manager(s) and (iii) in the case of Company Parties that are natural persons, such natural person;

(b) with respect to each Entity Company Party, (i) a certificate of the general partner or manager of such Entity Company Party, as the case may be, certifying to and attaching a copy of such Entity Company Party's Organizational Documents, a certificate of fact of good standing and (ii) Texas tax clearance certificate(s) with respect to such Entity Company Party that covers all franchise and sales and use Taxes and personal property Taxes, each dated as of a date no more than 20 Business Days prior to the Closing Date, (iii) resolutions adopted by such Entity Company Party's governing body and partners or members, as the case may be, authorizing the Transaction Documents and the transactions contemplated herein and therein, and (iv) incumbency certificates, each in the form reasonably satisfactory to Buyers and dated as of the Closing Date;

(c) (i) a Lease Termination and Release, duly executed by the lessor and the lessee under that certain Lease Agreement, dated April 1, 2016, between A. Keith Crawford and CP, for the real property located at 320 Steck Avenue, Unit 210 and 220, Austin, TX, as amended or restated through the Closing Date, and (ii) evidence acceptable to Buyers of the termination of each real property lease (other than the lease referenced in <u>clause (i)</u> above) pursuant to which any Real Property Owner is the lessor and a Company is a lessee; provided, however, that in connection with the termination of each such real property Owner shall forever release and discharge each Company, its successors, assigns, affiliates, officers, directors, members, managers, employees, agents and attorneys, from any and all actions, causes of action, suits, claims, and demands, of every kind and nature whatsoever whether known or unknown, mutual mistake of fact notwithstanding, that any Real Property Owner ever had, now has, or may have had, against any Company or its successors, assigns, affiliates, officers, members, managers, partners, employees, agents or attorneys, arising out of such real property leases, expressly including those obligations and indemnifications that otherwise would have survived the expiration or earlier termination of the relevant real property leases as provided therein or by applicable Laws;

(d) an executed landlord estoppel certificate in form and substance acceptable to Buyers for each of the Third Party Leases, but excluding any Third Party Lease which is month-to-month;

(e) a certificate duly executed by the Company Parties (i) certifying that the employees of the Companies, which have been requested in writing by Buyers at least five Business Days prior to the Closing, have resigned or been terminated prior to or concurrently with the Closing;

(f) Assignments, duly executed by the Sellers;

20

(g) a Lease Agreement, with respect to each real property location described on <u>Annex A</u>, in a form reasonably satisfactory to the parties, duly executed by the Real Property Owner party thereto;

(h) the Escrow Agreement, duly executed by the Sellers, the Companies and the Escrow Agent;

(i) the Restrictive Covenant Agreements, in a form reasonably satisfactory to Buyers, each duly executed by CP and the Employees named therein;

(j) the Transition Services Agreement, duly executed by the Sellers;

(k) (i) the Payoff Letters with respect to all Indebtedness set forth on <u>Schedule 3.2</u>, and (ii) releases of the Liens set forth on <u>Schedule 3.2</u>;

(1) a certificate of the Sellers' non-foreign status that complies with the requirements of Sections 1445 and 1446(f) of the Code, and the regulations promulgated thereunder;

(m) an opinion of counsel to the Company Parties, dated the Closing Date, addressed to Buyers, and in form and substance acceptable to Buyers; and

(n) resolutions duly adopted by the partners of CP and ASC and by the manager and member of MCS ratifying all prior company actions and the appointment of persons authorized on their behalf to act, in form and substance acceptable to the Sellers and Buyers.

4.3 Company Parties' Conditions to Close. The obligations of the Company Parties to consummate the transactions contemplated by this Agreement are subject to the satisfaction (or the written waiver by the Company Parties) of the following conditions as of the Closing Date:

4.3.1 The representations and warranties set forth in <u>Article 6</u> shall be true and correct at and as of the Closing as though then made (other than those representations and warranties that address matters as of particular dates which shall be true and correct at and as of such particular dates), except for inaccuracies that are de minimis in amount and effect, disregarding all qualifications contained herein regarding materiality or Material Adverse Effect;

4.3.2 Buyers shall have performed in all material respects all the covenants and agreements required to be performed by it under this Agreement at or prior to the Closing;

4.3.3 All consents and approvals set forth on <u>Schedule 4.3.3</u> shall have been obtained;

4.3.4 The applicable waiting periods, if any, under the HSR Act shall have expired or been terminated;

4.3.5 No judgment, decree or order shall have been entered which would prevent the Closing;

21

4.3.6 Buyers shall have delivered to the Sellers each of the following:

(a) a certificate in the form reasonably satisfactory to Sellers, dated as of the Closing Date, stating that the preconditions specified in Sections 4.3.1 and 4.3.2 have been satisfied, duly executed by an officer of each Buyer;

(b) for each Buyer, a certificate of an authorized officer certifying to and attaching a copy of such Buyer's Organizational Documents, a certificate of fact of good standing dated as of a date no later than five Business Days prior to the Closing Date, and resolutions adopted by such Buyer's governing body authorizing the Transaction Documents and the transactions contemplated herein and therein, and incumbency certificates;

(c) the Lease Agreements, in a form reasonably satisfactory to Buyers, each duly executed by a Buyer or an Affiliate of such Buyer;

- (d) the Escrow Agreement, duly executed by Buyers and the Companies; and
- (e) the Transition Services Agreement, duly executed by BBCP.

ARTICLE 5

REPRESENTATIONS AND WARRANTIES OF THE COMPANY PARTIES

The Company Parties hereby represent and warrant, jointly and severally, to Buyers as of the date hereof and as of the Closing Date, as follows:

5.1 Organization. CP and ASC are each limited partnerships duly organized, validly existing and in good standing under the Laws of the State of Texas, and have all limited partnership power and authority to own or lease their properties and assets and to operate and carry on their Business as it is now being conducted. MCS, CR LLC and CTCS are each limited liability companies duly organized, validly existing and in good standing under the Laws of the State of Texas, and have all limited iability company power and authority to own or lease their properties and assets and to operate and carry on their Business as it is now being conducted. MCS, CR LLC and CTCS are each limited liability company power and authority to own or lease their properties and assets and to operate and carry on their Business as it is now being conducted. The copies of the Organizational Documents of CP and ASC previously made available by them in the electronic data room to Buyers are true, correct and complete. The copies of the Organizational Documents of MCS, CR LLC and CTCS previously made available by them in the electronic data room to Buyers are true, correct and complete. Each Entity Company Party is duly licensed or qualified to do business and (where applicable) is in good standing as a foreign corporation in each jurisdiction in which the ownership, operation or lease of its property or the character of its activities is such as to require it to be so licensed or qualified or in good standing, as applicable, except where the failure to be so licensed or qualified or in good standing would not reasonably be expected to have a Material Adverse Effect.

5.2 Due Authorization. Each Entity Company Party has all requisite power and authority to execute and deliver this Agreement and each other Transaction Document to which it is a party (subject to the consents, approvals, authorizations and other requirements described in Section 5.2) and to consummate the transactions contemplated hereby and thereby and to perform all obligations to be performed by them hereunder and thereunder. The execution and delivery of this Agreement and the other Transaction Documents to which it is a party by each Entity Company Party and consummation of the transactions contemplated hereby and thereby have been duly and validly authorized and approved by all general partner, limited partner and limited liability company proceedings on the part of the Entity Company Parties necessary to authorize this Agreement and the Transaction Documents. This Agreement has been, and each of the Transaction Documents to which each Company Party is a party has been or will be at or prior to Closing, duly and validly executed and delivered by such Company Party and (assuming this Agreement constitutes a legal, valid and binding obligation of Buyers) constitutes a legal, valid and binding obligation of the Company Parties, enforceable against them in accordance with its terms, subject to applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and similar Laws affecting creditors' rights generally and subject, as to enforceability, to general principles of equity. A. Keith Crawford and Melinda Crawford are natural Persons and have the capacity to execute and deliver this Agreement and each other Transaction Document to which he or he is a party (subject to the consents, approvals, authorizations and other requirements described in Section 5.2) and to consummate the transactions contemplated hereby and thereby and to perform all obligations to be performed by them hereunder and thereunder.

22

5.3 No Conflict. Subject to the receipt of the consents, approvals, authorizations and other requirements set forth on <u>Schedule 5.3</u>, the execution, delivery and performance of this Agreement and the Transaction Documents by the Company Parties and the consummation by the Company Parties of the transactions contemplated hereby and thereby do not and will not indirectly or directly, (a) violate any provision of, or result in the breach of, any applicable Law to which any Company Party is subject or by which any property or asset of the Companies is bound, (b) conflict with, result in a breach or violation of or constitute a default under any of the provisions of the Organizational Documents of the Entity Company Parties, (c) violate any provision of, result in a breach of, require a consent under, terminate or result in the termination of, or give rise to any right of revocation, withdrawal, suspension, acceleration, cancellation, modification, imposition of additional obligations or loss of rights or payment becoming due under any Contract or result in the creation of any Lien upon any of the properties or assets of any Company Party, or constitute an event which, with or without notice or lapse of time or both, would result in any such violation, breach, termination, right or creation of a Lien or (d) result in a violation or revocation of any required license, permit or approval from any Governmental Authority.

5.4 Compliance with Laws. Each Company Party is in compliance, in all material respects, with and, during the three year period prior to the date hereof, have complied in all material respects with, all Laws and judgments applicable to the ownership, occupancy and use of the Purchased Assets and the Real Property, and the operation of the Business, and the employment of the Employees. The Purchased Assets and Real Property and the use, operation and maintenance thereof and the Business are in material compliance with all Contracts applicable with respect thereto.

5.5 Governmental Consents. Except as may result from any facts or circumstances relating solely to Buyers or any of their Affiliates, no consent, approval or authorization of, or designation, declaration or filing with, any Governmental Authority is required on the part of any Company Party with respect to the Company Parties' execution or delivery of this Agreement or the consummation by the Company Parties of the transactions contemplated hereby, except for applicable requirements of the HSR Act.

5.6 Capitalization.

5.6.1 <u>Schedule 5.6</u> sets forth (a) the full legal name and legal address of each Company Party, (b) the Purchased Interests owned, beneficially and of record, by each Seller and (c) the names and addresses of the legal owners of the Entity Company Parties.

5.6.2 Each Seller is the sole record and beneficial owner of the Purchased Interests set forth next to such Seller's name on <u>Schedule 5.6</u>, such Seller has good and marketable title to such Purchased Interests and, at the Closing, the Purchased Interests shall be free and clear of all Liens.

23

5.6.3 The Purchased Interests constitute, in the aggregate, all of the issued and outstanding partnership and membership interests of each of the Companies, and except for the Purchased Interests, there are no (a) other issued and outstanding equity or economic interests or voting securities of any nature, type of kind of any Company (or contracts or agreements obligating any Company to issue, deliver or sell or cause to be issued, delivered or sold any of the foregoing) or (b) voting trusts, proxies or other agreements or understandings with respect to the voting of any equity interest of any Company.

5.6.4 There are no outstanding obligations of any Company (contingent or otherwise) to repurchase, redeem or otherwise acquire, directly or indirectly, any partnership or membership interests (or options or warrants to acquire any such interests) of such Company, and there are no outstanding rights to cause any Company to register its securities or which otherwise relate to the registration of any securities of such Company.

5.6.5 The Purchased Interests have been duly authorized and validly issued, and such Purchased Interests are not subject to, nor were they issued in violation of, any purchase option, call option, right of first refusal or offer, co-sale or participation right, preemptive right, subscription right or similar right.

5.6.6 Each Seller has all requisite power and authority to vote his, her or its Purchased Interests, and there are no proxies, voting trusts, powers of attorney, divorce or marital decrees, Contracts, Laws or other restrictions or limitations binding on such Seller with respect to his, her or its Purchased Interests.

5.6.7 There are no declared or accrued but unpaid distributions with respect to the Purchased Interests.

5.6.8 There are no outstanding equity-appreciation rights, equity-based performance units, "phantom" equity rights or other contracts or obligations of any character (contingent or otherwise) pursuant to which any Person is or may be entitled to receive any payment or other value based on the revenues, earnings or financial performance, equity price performance or other attribute of such Company or its business or assets or calculated in accordance therewith (other than payments or commissions to sales representatives of such Company based upon revenues generated by them without augmentation as a result of the transactions contemplated hereby, in each case in the ordinary course of business).

5.6.9 Following the consummation of the transactions contemplated herein (a) BBCP will be the sole owner of 100% of CP Limited Partnership Interests, 100% of the ASC Limited Partnership Interests and 100% of the MCS Membership Interests free and clear of all Liens, and (b) CPHA LLC will be the sole owner of 100% of the CP General Partnership Interests and 100% of the ASC General Partnership Interests, ASC Limited Partnership Interests, MCS Membership Interests, CP General Partnership Interests or ASC General Partnership Interests are certificated.

5.6.10 Melinda Crawford is the sole manager of MCS.

5.6.11 None of the Companies owns any equity interest of any Person.

5.7 Assets Used in the Business.

5.7.1 The Companies have the right to own, operate, use and maintain their properties and assets in the manner in which they are now operated, used and maintained and to conduct the business of the Companies as currently conducted. <u>Schedule 5.7.1</u> sets forth a complete and accurate list of Pumping Equipment and Vehicles owned or used by the Companies in the operation of the Business. The Companies have good and marketable title to, or have a valid and enforceable leasehold or license to, its properties and assets used in the business. As of the Closing, all properties and assets used in the Business are free and clear of all Liens (including protective Lien filings) other than Permitted Liens, capital leases, financing arrangements, and free of any restrictions on use.

24

5.7.2 <u>Schedule 5.7.2</u> sets forth a list of all deposits (the sum of such deposits being the "<u>Deposit Amount</u>"), including the amounts and a description of each deposit, the name of the Person holding such deposit, all Persons with any right, title or interest in or to such deposit, and any claim that may be made against such deposit.

5.7.3 The Customer Lists maintained by the Companies are true, correct and complete in all material respects, and are the only lists of the customers of the Business, and the Sellers have no other or separate list of the customers of the Business.

5.8 Sufficiency of Assets; No Third Party Rights. The Companies own or lease all of the assets, properties, titles, rights and interests necessary for the conduct of the Business as presently conducted. All of the Vehicles, Pumping Equipment, other equipment, machinery, fixtures, improvements and other tangible assets (whether owned or leased) used in the Business, are fully operational and will pass a boom inspection and Department of Transportation inspection, and have been maintained and are in good operating condition and repair (ordinary wear and tear excepted) and are fit for use in the ordinary course of the Business as presently conducted. The Purchased Assets, together with the rights and obligations of the parties set forth in this Agreement and the other Transaction Documents, are all of the properties and assets (tangible and intangible) necessary for the operation and conduct of the Business as presently conducted. To the extent used in the Business, the Companies' assets and inventory consist of a quantity and quality usable and salable in the ordinary course of business, are not obsolete, defective, or damaged (ordinary wear and tear excepted), are merchantable and are fit for their intended use.

5.9 Financial Statements. Attached as <u>Schedule 5.9</u> are (a) the audited consolidated balance sheets and statements of income, cash flows and equity holders' equity of each Company, as of and for the fiscal years ended December 31, 2017 and December 31, 2018, together with the auditor's report thereon, including, in each case, any notes thereto, (the "<u>Audited Financial Statements</u>") and (b) an unaudited consolidated balance sheet and statements of income and cash flow of each Company as of and for the one month period ended January 31, 2019 (the "<u>Interim Financial Statements</u>" and, together with the Audited Financial Statements, the "<u>Financial Statements</u>"). The Financial Statements (including the notes thereto) are true, correct and complete, are consistent with the books and records of the Companies and present fairly, in all material respects, the consolidated financial position and results of operations of the Companies as of the dates and for the periods indicated in such Financial Statements in conformity with GAAP (except in the case of the Interim Financial Statements, for the absence of footnotes and other presentation items and for normal year-end adjustments).

5.10 Undisclosed Liabilities; Indebtedness. There is no liability, debt or obligation (whether accrued, absolute, contingent, direct, indirect, perfected, inchoate, unliquidated or otherwise and whether due or to become due) of any Company, except for liabilities and obligations (a) reflected or reserved for on the Financial Statements or disclosed in the notes thereto, (b) that have arisen since the date of the most recent balance sheet included in the Financial Statements in the ordinary course of business consistent with past practices (none of which relate to a breach of Contract, breach of warranty, tort, infringement, violation of Law, Governmental Order, permit or any Action), (c) incurred in connection with the transactions contemplated by this Agreement or (d) set forth on <u>Schedule 5.11</u>. <u>Schedule 5.10</u> sets forth a true, correct and complete accounting of the outstanding Indebtedness of the Companies. The Companies have performed in all material respects all of its obligations required to be performed by it under each document evidencing their Indebtedness. No Company has any outstanding liability, including, without limitation, amounts owing as deferred purchase price, including all seller notes and "earn-out" payments, in respect of the acquisition of any business or division, equity interests or all or a material portion of the assets of any Person.

25

5.11 Litigation and Actions. Except as set forth on <u>Schedule 5.11</u>, there are no Actions pending, or to the Company Parties' Knowledge, threatened, and there exists no basis for any Action, (a) against or by any Company affecting its business or any of its properties or assets (or by or against Seller or any Affiliate thereof and relating to the Company) or (b) against or by any Company Party or any Affiliate of any Company Party that challenges or seeks to prevent, enjoin or otherwise delay the transactions contemplated by this Agreement. No event has occurred or circumstances exist that may give rise to, or serve as a basis for, any such Action. There are no investigations pending, or to the Company Parties' Knowledge, threatened, against any Company Party or any officer, general or limited partner, director, employee, or manager of any Entity Company Party with respect to such Person's capacity as such. To the Company Parties' Knowledge, all liabilities or losses suffered or incurred in connection with matters set forth on <u>Schedule 5.11</u> are fully covered by insurance policies maintained by the Companies, subject to applicable deductibles, retentions and other policy limitations, and no claims in respect thereof have been denied or disputed by any applicable insurer. There is no material unsatisfied judgment, or any material obligation pursuant to any settlement agreement or any open injunction binding upon any Company. <u>Schedule 5.11</u> sets forth a list of all closed or resolved Actions during the preceding three years applicable to the Companies (including their predecessors) or the conduct of their businesses or the ownership or use of any of their properties or assets, including the date such Action was commenced and concluded, and the nature of the resolution thereof.

5.12 Contracts.

5.12.1 <u>Schedule 5.12</u> sets forth a true, correct and complete list and, if oral, a complete description of:

(a) all Contracts (including all Affiliate Agreements) that any Company Party is a party to or is bound by that affect or relate to the Companies or their Business or the ownership of use of any of their properties or assets, and that provide for, or is likely to result in, payments in any 12-month period by any party thereto in excess of \$150,000;

(b) any written Contract for the employment of any officer, individual employee or other Person on a full time, part time, consulting or other basis;

(c) any Contract under which any Company has advanced or loaned money to, guaranteed an amount for the benefit of or made an investment in any other Person;

(d) any Contract, promissory note or indenture relating to borrowed money or other Indebtedness or the mortgaging, pledging or otherwise placing a Lien on any asset or group of assets of the Companies;

(e) any Contract regarding any material indemnification provided by any Company against third party claims;

26
(f) any Contract prohibiting any Company from freely engaging in the Business anywhere in the United States, granting most favored nation pricing or exclusive rights to a counterparty or requiring it to purchase all or substantially all of its requirements for a product or service from a particular Person;

(g) any material non-disclosure or confidentiality agreements, other than those entered into in contemplation of a sale of the Companies or assets of the Companies (outside of the ordinary course of business); or

(h) any settlement, conciliation or similar agreement with any Governmental Authority or other Person containing obligations yet to be performed or completed by either or both parties.

5.12.2 True, correct and complete copies of the written Contracts and descriptions of oral Contracts listed on <u>Schedule</u> <u>5.12</u> (including all Affiliate Agreements) have been delivered to Buyers. To the Company Parties' Knowledge, no Person who is a party to any such Contract is in default, nor has any event occurred which, with the giving of notice or the passage of time or both, would constitute a material default, under any such Contract or any other material obligation owed by any Company Party or Affiliate of any Company Party, and to the Company Parties' Knowledge, no event has occurred which, with the giving of notice or the passage of time or both, would constitute a material default by any other party to any such Contract.

5.12.3 The Companies maintain all work in process within the PumpMagic software system in the ordinary course of business consistent with industry standards. The PumpMagic software system maintained by the Companies includes a true, correct and complete list of all work in process as of the Closing Date.

5.13 Employee Benefit Plans.

5.13.1 <u>Schedule 5.13.1</u> sets forth a true, correct and complete list of each Employee Benefit Plan. With respect to each Employee Benefit Plan set forth on <u>Schedule 5.13.1</u>, the Companies have made available in the electronic data room to Buyers true, correct and complete copies of (a) such Employee Benefit Plan document and any amendments thereto and, with respect to any unwritten Employee Benefit Plan, a written description of the material terms of such Employee Benefit Plan, (b) the most recent summary plan description (if any), (c) the most recent annual report on Form 5500s and all attachments thereto filed with the Internal Revenue Service with respect to such Employee Benefit Plan (if applicable), (d) the most recent nondiscrimination testing results, if applicable, (e) the most recent determination, advisory, or opinion letter, if any, issued by the Internal Revenue Service with respect to such Employee, and (f) any material correspondence with any Governmental Authority within the past 12 months.

5.13.2 With respect to each Employee Benefit Plan:

(a) (i) such Employee Benefit Plan has been maintained, funded, operated, and administered in all material respects in accordance with its terms and all applicable Laws, including ERISA and the Code, (ii) all contributions and premiums required to be made with respect to such Employee Benefit Plan have been made or, to the extent not yet due, accrued on the Companies' Financial Statements and (iii) if such Employee Benefit Plan is intended to be qualified within the meaning of Section 401(a) of the Code, such Employee Benefit Plan has received a favorable determination, advisory or opinion letter as to the form of such Plan and, to the Company Parties' Knowledge, nothing has occurred that would reasonably be expected to adversely affect the qualified status of such Employee Benefit Plan.

27

(b) No Employee Benefit Plan is, and no Company or Affiliate of any Company, and none of their respective ERISA Affiliates sponsors or contributes to, or has, within the past six years, sponsored, contributed to or been required to contribute to, (i) a multiemployer plan (within the meaning of Section 4001(a)(3) of ERISA), any other pension plan subject to Title IV of ERISA, Section 302 of ERISA or Section 412 of the Code, a "multiple employer plan" within the meaning of Section 210(a) of ERISA or Section 413(c) of the Code, (ii) a "multiple employer welfare arrangement" within the meaning of Section 3(40) of ERISA or (iii) a voluntary employee's beneficiary association that is exempt from taxation under Section 501(c)(9) of the Code.

(c) No Employee Benefit Plan provides health, life, death or disability benefits to any officer, director or employee of any Employee following retirement or other termination of employment, other than as required by Section 4980B of the Code, or similar applicable Law, or for continued coverage through the end of the month in which such retirement or termination occurs.

(d) With respect to the Employee Benefit Plans, no Actions (other than routine claims for benefits in the ordinary course) are pending or, to the Company Parties' Knowledge, threatened against any Company. The Companies have, for purposes of each Employee Benefit Plan and for purposes of Tax withholding, correctly classified all individuals performing services for any such entity as employees and independent contractors, as applicable.

(e) With respect to each Company Employee Benefit Plan, no Company or any ERISA Affiliates is currently liable for any tax arising under sections 4971, 4972, 4975, 4979, 4980 or 4980B of the Code, and no fact or event exists that would give rise to any such tax liability. No Company has any obligation or commitment to "gross up" any Person with respect to Taxes under Section 409A or 4999 of the Code.

(f) No Company or any ERISA Affiliate is under any obligation (express or implied) to modify any Employee Benefit Plan, or to establish any new Employee Benefit Plan for the benefit of the Company's employees. Each Employed Benefit Plan has expressly reserved to the Company the right to amend, modify or terminate each Employee Benefit Plan, or any portion of any Employee Benefit Plan, without liability to the Company.

(g) Except as set forth in <u>Schedule 5.13.2</u>, neither the Companies' execution of, nor the performance of the transactions contemplated by, this Agreement will, either alone or in connection with any other event, (i) result in any payment, severance or benefit becoming due to any Employee or Former Employee, director, officer or independent contractor of any Company, (ii) increase the amount of any compensation, severance, or benefits payable under any Employee Benefit Plan, or (iii) result in the acceleration of the time of payment, funding, or vesting of any compensation, severance, or benefit due to any Employee or Former Employee, director of any Company.

(h) Each Employee Benefit Plan that is a nonqualified deferred compensation plan (as defined in Section 409A(d)(1) of the Code) has been maintained in all material respects in documentary and operational compliance with Section 409A of the Code or an available exemption therefrom, and no participant has incurred or would reasonably be expected to incur income acceleration or Taxes under Section 409A of the Code with respect to any payment to be made under any such Employee Benefit Plan.

28

5.14 Employment Matters.

5.14.1 No Company is a party to any collective bargaining agreement, work council agreement, trade union agreement, or other similar agreement for the representation of Employees and there is no labor union or other organizations representing any Employees of the Companies. There has not been, there is not presently pending or existing, and there is not threatened (a) any labor strike, slowdown, work stoppage, picketing or other labor disruption pending with respect to Employees or (b) any unfair labor practice. No union or labor representative organizing activities are taking place or have taken place with respect to any Employee in the three years prior to Closing at any of the locations operated by any Company. To the Company Parties' Knowledge, no petition for representation concerning Employees has been filed with the National Labor Relations Board or any other Governmental Authority seeking certification as the collective bargaining representative of any Employee.

5.14.2 To the extent permitted by Law, <u>Schedule 5.14.2-A</u> attached hereto sets forth a true, correct and complete list of all Persons who are employed by any Company, or otherwise providing services to any Company in connection with the conduct of the Business as of the date hereof. Schedule <u>5.14.2-B</u> sets forth a true, correct and complete list of all Persons who have been employed by any Company, or otherwise providing services to any Company in connection with the conduct of the Business at any time during the 45 days prior to Closing. <u>Schedule 5.14.2-A</u>, as of the date hereof, and <u>Schedule 5.14.2-B</u>, as of the Closing Date, correctly states, with respect to each Person identified on such schedule, such Person's title and job function, location, employing entity, current annual rate of compensation or hourly wage, current target bonus opportunity, other material compensation arrangements, employee loans, employee status (full-time or part-time, temporary, exempt or non-exempt, and active or on leave or disability), and date of hire. To the Company Parties' Knowledge, no Person designated as an "exempt" employee on <u>Schedule 5.14.2-A</u>, as of the date hereof, or <u>Schedule 5.14.2-B</u>, as of the Closing Date, will terminate his or her employment as a result of the consummation of the transactions contemplated by this Agreement or otherwise, has not notified any Company Party of such Person's intention to terminate, nor given any Company Party any express indication that such Person will terminated such employment before or after the Closing.

5.14.3 Except as set forth on <u>Schedule 5.14.3</u>, no Company has entered into any employment agreement with any Employee, and all Employees may be terminated at will. Except as set forth on <u>Schedule 5.14.3</u>, there is no contractual obligation or special termination or severance arrangement in respect of any of the Employees (including any Former Employee) and there is no provision of any agreement or arrangement with any of the Employees (including any Former Employee), or any other legal or contractual requirement, which would obligate any Company to employ any of the Employees (or any time in the two years prior to the Closing Date was a party to, had an obligation or other arrangement with respect to, or was bound by). The Company Parties have provided Buyers with true, correct and complete copies of each such employment agreement (or a complete written description thereof if the employee arrangement is not set forth in writing). All Employees classified as exempt under the Fair Labor Standards Act and state and local wage and hour Laws are properly classified.

5.14.4 The Companies have paid in full all wages, bonuses, commissions and other benefits and sums due (and all required Taxes, insurance, social security and withholding thereon), including all accrued vacation, accrued sick leave, accrued benefits and accrued payments (and pro rata accruals for a portion of a year) to its Employees and Former Employees due and owing. The Companies have maintained in effect all insurance policies and other employee benefits covering any employee claims incurred through and including the Closing Date.

29

5.14.5 To the extent permitted by Law, <u>Schedule 5.14.5</u> sets forth a true, correct and complete list of all Persons who have been engaged by any Company as an independent contractor and consultant by any Company at any time during the 45 days prior to Closing, including a description of the services provided and the most recent rate of compensation payable to such Person. The Company Parties have provided Buyers with true, correct and complete copies of each such consulting or independent contractor agreement (or a complete written description thereof if the consulting or independent contractor arrangement is not set forth in writing). All independent contractors and consultants are properly classified. To the Company Parties' Knowledge, no Company has any liability, whether absolute or contingent, direct or indirect, including for the purposes of Tax withholding Laws or any under any benefit plan, with respect to any misclassification of a Person as an independent contractor rather than as an employee or with respect to any employees "leased" from another entity. No Person who is not an Employee of a Company has made or, to the Company Parties Knowledge, threatened any claim that such Person is or should be classified as an Employee of a Company under applicable Law, and there is no basis for any liability of the Companies arising from the classification of a natural person as an independent contractor or other non-employee status.

5.14.6 <u>Schedule 5.14.6</u> sets forth a true and complete list of all open, unresolved, reserved for, and incurred but not yet reported Workers Compensation Claims, and other claims arising out of injuries or occupational safety matters involving Employees (or Former Employees, to the extent the same have not been fully resolved) or sustained by Employees or Former Employees (or Former Employees, to the extent the same have not been fully resolved) ("<u>Workers Compensation Claims</u>"), along with the nature of the claim, date of injury, date of claim, amount paid or claimed to date and amount reserved for, arising or with a date of occurrence on or before the last day of employment of such Employee by any Company (the "<u>Employees' Termination Date</u>"). The Companies have been in compliance, in all material respects, with the filing requirements with appropriate workers' compensation or other Governmental Authorities on all claims arising out of injuries or occupational safety matters based on an injury or event occurring on or prior to Closing.

5.14.7 The Companies are and have been in compliance with all applicable Laws relating to employment, wages and hours, immigration, plant closings and layoffs under the Worker Adjustment and Retraining Notification Act of 1988 (the "<u>WARN</u> <u>Act</u>") and other similar applicable Laws, unemployment insurance, workers' compensation, pay equity, discrimination in employment, wrongful discharge, collective bargaining, fair labor standards wages and hours, affirmative action, civil rights, background checks, hiring practices, and occupational health and safety, the payment of social security and similar Taxes, income Tax withholding and privacy rights of Employees. There are no Actions pending, or to the Company Parties' Knowledge, threatened against the Companies by or on behalf of any Employee or Former Employee related to any labor or employment matter. No Company, and no predecessor to any Company, has taken any action that constitutes a "mass layoff," "mass termination," or "plant closing" within the meaning of the WARN Act or similar state, local, or foreign Laws, affecting any site of employment or one or more facilities of any of the Companies.

5.14.8 The Companies have delivered to Buyers true, correct and complete copies of forms of the non-competition and non-solicitation Contracts entered into between the Companies and all Business Personnel. The Companies have not sought to enforce any noncompetition or non-solicitation Contract covering any former Business Personnel.

5.14.9 To the Company Parties' Knowledge, no Employee or Former Employee has suffered or is suffering from any illness or disease caused directly or indirectly by any employment related condition or by contact with any Hazardous Materials within the scope of such Person's employment.

5.14.10 No Company is a covered federal contractor or subcontractor under Executive Order 11246, as amended (including, without limitation, Section 503 of the Rehabilitation Act of 1973 and the Vietnam Era Veterans Readjustment Act of 1974). No Company has any affirmative action plans with any Governmental Authority.

5.14.11 None of the Companies' Business Personnel is an undocumented alien, and a properly completed Form I-9 is on file with the Companies for each such Person. The Companies have complied with the U.S. Immigration and Nationality Act, as amended from time to time, and the rules and regulations promulgated thereunder, and with all and there is no basis for any claim that any Company is not in compliance with the terms thereof. No Company Party has received notice or other communication from any Governmental Authority regarding any unresolved violation or alleged violation of any applicable Law relating to the hiring, recruiting, employing of (or continuing to employ) anyone who is not legally authorized to work in the United States. The Company Parties have provided Buyers with true, correct and complete copies of all Form I-9s with respect to the Companies' Business Personnel.

5.14.12 <u>Schedule 5.14.12-B</u> as of the Closing Date, sets forth a true, correct and complete list of the amount and value of all accrued but unpaid salary, earned but unpaid bonuses, unpaid reimbursable expenses, earned (whether by acceleration or otherwise) but unpaid commission and the net amount and value of all vacation, paid time off and related payments to which the Employees are entitled as of such date (the "<u>PTO and Payroll/Commission Amount</u>").

5.15 Taxes.

5.15.1 Each Company has duly and timely filed (taking into account any valid extensions of time within which to file) all Tax Returns that were required to have been filed by it on or prior to the Closing Date, all such Tax Returns are true, complete and correct in all material respects, and all Taxes that are due and payable by each such Company (whether or not shown as due on such Tax Returns) have been timely paid in full.

5.15.2 Each Company has withheld and timely remitted to the appropriate taxing authorities all Taxes required to have been withheld and remitted in connection with amounts paid or owing to any Business Personnel, creditor, equityholder or other Person.

5.15.3 There is no Tax Contest pending or, to the Company Parties' Knowledge, threatened against any Company with respect to any Taxes.

5.15.4 Other than Permitted Liens, there are no Liens for Taxes on any of the properties or assets used in the business, nor is any taxing authority in the process of imposing any Liens for Taxes on any of the properties or assets used in the business.

5.15.5 All deficiencies asserted, or assessments made, against a Company as a result of any examinations by any taxing authority have been fully paid.

31

5.15.6 No Company Party has waived or extended or been requested to waive or extend any statute of limitations for the assessment, collection or imposition of any Taxes, which waiver or extension is currently in effect.

5.15.7 No claim or nexus inquiry has been made by any Governmental Authority in a jurisdiction where a Company does not file Tax Returns or pay Taxes (including withholding Taxes) that a Company is or may be subject to taxation by, or have a duty to deduct, withhold or collect and pay over Taxes to, that jurisdiction (including federal, state, local or non-U.S. jurisdictions). No Company has commenced a voluntary disclosure proceeding in any state or local or non-U.S. jurisdiction relating to any material Taxes that has not been fully resolved or settled.

5.15.8 Each Company (a) is not, and never has been, a party to or otherwise bound by any Tax sharing, allocation or indemnification agreement under which a Company would reasonably be expected to be liable after the Closing Date for a material Tax liability of any Person that is not a Company or Company Party, other than commercial agreements or arrangements that do not relate primarily to Taxes, (b) is not and never has been a member of an Affiliated Group or (c) does not have any liability for the Taxes of any other Person under U.S. Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or foreign Law), as a transferee or successor, by Contract or otherwise.

5.15.9 No Company Party is a "foreign person" within the meaning of Section 1445 of the Code. Neither the Code nor any other applicable Law requires Buyers to withhold any portion of the Purchase Price.

5.15.10 No Company is, and never has been, a party to, or a promoter of, a "reportable transaction" within the meaning of Section 6707A(c)(1) of the Code and U.S. Treasury Regulation Section 1.6011-4(b).

5.15.11 None of the properties or assets used in the business constitutes stock in a corporate Subsidiary or a joint venture, partnership interest, limited liability company interest or other equity interest in any Person or an arrangement or Contract which is taxed as a partnership for U.S. federal income tax purposes (or the right to acquire any of the foregoing interests).

5.15.12 Each Company does not have (and has not previously had) any permanent establishment in any foreign country, and such Company does not engage (and has never previously engaged) in a trade or business in any foreign country.

5.15.13 For U.S. federal and applicable state and local income Tax purposes, no Buyer or any of its Affiliates will be required to recognize any income on or after the Closing Date on account of any deferred revenue or prepaid income received by a Company prior to the Closing Date.

5.15.14 Each Company has complied with all Laws relating to sales and use Taxes, including receipt and retention of any required exemption certificates. The Companies have complied with all Laws relating to unclaimed property and escheat, including the filing of all required Tax Returns and reports with respect thereto.

5.15.15 None of (a) the goodwill, (b) going concern value or (c) other intangible assets of a Company that would not be amortizable prior to the enactment of Section 197 of the Code was held by a Company (or any "related person" (within the meaning of Section 197(f)(9)(C) of the Code)) on or before August 10, 1993 or could constitute anti-churning property under Section 197(f)(9) (A) of the Code.

32

5.15.16 None of the properties or assets used in the business is (a) required to be treated as being owned by another person pursuant to the so-called "safe harbor lease" provisions of former Section 168(f)(8) of the Internal Revenue Code of 1954, as amended, (b) subject to Section 168(g)(1)(A) of the Code, or (c) subject to a disqualified leaseback or long-term agreement as defined in Section 467 of the Code.

5.15.17 Any related party transaction conducted by a Company has been on an arms-length basis in accordance with applicable Law, and all documentation and reporting requirements under applicable transfer pricing Laws have been timely complied with and, if necessary, all relevant documentation has been retained in accordance with applicable transfer pricing Laws.

5.15.18 Each Company is (and has been for its entire existence) classified as a partnership for all Income Tax purposes. No election has ever been filed to treat Company as an association taxable as a corporation for income tax purposes.

5.15.19 Since the date of the most recent balance sheet, no Company has (a) incurred any material Taxes outside in the ordinary course of business consistent with past practices, (b) changed a material method of accounting for Tax purposes, (c) entered into any agreement with any Governmental Authority (including a "closing agreement" under Code Section 7121) with respect to any material Tax matter, (d) surrendered any right to a material Tax refund, (e) changed an accounting period with respect to Taxes, (f) filed an amended material Tax Return or (g) changed or revoked any material Tax election.

5.15.20 No Company is required to include any material item of income in, or exclude any material item of deduction from, its taxable income for any period (or portion thereof) beginning after the Closing Date as a result of (a) an installment sale transaction occurring on or before the Closing governed by Code Section 453 (or any similar provision of state, local or non-U.S. Laws), (b) a transaction occurring on or before the Closing reported as an open transaction for U.S. federal income Tax purposes (or any similar doctrine under state, local, or non-U.S. Laws), (c) any prepaid amounts received or paid on or prior to the Closing or deferred revenue realized on or prior to the Closing, (d) a change in method of accounting made before the Closing Date with respect to a pre-Closing tax period or (e) a Tax agreement entered into with any Governmental Authority (including a "closing agreement" under Code Section 7121 or any "gain recognition agreements" entered into under Code Section 367) on or prior to the Closing Date.

5.15.21 No Company has made an election (including a protective election) pursuant to Code Section 108(i).

5.15.22 No Company currently uses the cash method of accounting.

5.15.23 No Company has made an election under Section 965(h) or Section 965(n) of the Code.

5.15.24 All of the property of each Company that is subject to property Tax has been properly listed and described on the property tax rolls of the appropriate taxing authorities.

5.15.25 No Company has made an election to apply the Partnership Tax Audit Rules for any taxable period beginning prior to January 1, 2018.

33

5.16 Insurance.

5.16.1 <u>Schedule 5.16</u> contains a true, correct and complete list of all certificates, binders and policies of property, fire and casualty, product liability, workers' compensation, and other forms of insurance held by, or for the benefit of, the Companies as of the date of this Agreement. For each such certificate of insurance, binder and policy so listed, <u>Schedule 5.16</u> also sets forth: (a) all applicable binder or policy numbers, (b) the name and address of the insurance carrier, (c) the identity of the names insured, (d) the annual premium and the policy or binder period, and (e) the coverable liability limits (including any deductible or retention, if any). The Companies have delivered to Buyers true, correct and complete copies of such insurance policies. The historical insurance programs of the Companies are commercially reasonable and there are no historical gaps in coverage. Except as set forth in <u>Schedule 5.16</u>, (i) no Company has received any written notice from any insurer under any such insurance policies, canceling or materially adversely amending any such policy or denying renewal of coverage thereunder and (ii) all premiums on such insurance policies due and payable have been paid in full.

5.16.2 There are no outstanding claims which have been denied or disputed by the insurer. The Companies maintain, and at all times during the past three years have maintained, in full force and effect, certificates of insurance, binders and policies of such types and in such amounts and for such risks, casualties and contingencies as is reasonably adequate to insure the Companies against insurable losses, damages, claims and risks to or in connection with or relating to their respective businesses, properties, assets and operations. Except as set forth on <u>Schedule 5.16</u>, no Company has ever maintained, established, sponsored, participated in or contributed to any self-insurance program, retrospective premium program or captive insurance program.

5.17 Licenses and Permits. The Companies have obtained and are in material compliance with, all Licenses and Permits necessary under applicable Laws to permit the Companies to own, operate, use and maintain their assets in the manner in which they are now operated, used and maintained and to conduct the Business as currently conducted. All applications required to have been filed for the renewal of such Licenses and Permits have been duly filed on a timely basis with the appropriate Governmental Authority, and all other filings required to have been made with respect to such Licenses and Permits have been duly made on a timely basis with the appropriate Governmental Authority. All such Licenses and Permits are renewable by their terms or in the ordinary course of business consistent with past practices. Since January 1, 2016, (a) there has not occurred any default under any of the Licenses and Permits by any Company, (b) no Company Party has received any written notice from any Governmental Authority relating to the revocation or modification of any of the Licenses and Permits or with respect to any failure by any Company to have any License and Permit required in connection with the operation of the Business, the ownership of the properties and assets used in the Business, the use and occupancy of the Real Property or the employment of the Employees, and no material violations have been recorded in respect of any of the Licenses and Permits, and (c) to the Company Parties' Knowledge, there have been no threatened claims, actions, suits or other proceedings or investigations before or by any Governmental Authority that would reasonably be expected to result in the revocation or termination of any of the Licenses and Permits.

5.18 Owned Real Property. No Company owns any Real Property. <u>Schedule 5.18</u> sets forth the address of the Owned Real Property. With respect to each parcel of Owned Real Property, there are no outstanding options, rights of first offer or rights of first refusal or other options to purchase any Owned Real Property or any portion thereof or interest therein. No condemnation, eminent domain or similar proceeding is pending with respect to any Owned Real Property or any portion thereof, and to the Company Parties' Knowledge no condemnation or rezoning proceedings are threatened or contemplated with respect to any Owned Real Property.

34

5.19 Leased Real Property. Schedule 5.19 sets forth the address of each parcel of Leased Real Property, and describes each Third Party Lease. The Companies have delivered to Buyers true, correct and complete copies of the Third Party Leases, together with all other Contracts affecting or in connection with the Leased Real Property or the Third Party Leases (including but not limited to any estoppels, subordination agreements, commitment letters, etc.). No Company Party has leased or otherwise granted to any Person, other than the Company lessee, the right to use or occupy any portion of the Leased Real Property. There are no outstanding options, rights of first offer or rights of first refusal to purchase all or any portion of the Leased Real Property. With respect to each Third Party Lease: (a) such Third Party Lease is valid, binding, enforceable and in full force and effect, and the Company party as lessee to such Third Party Lease enjoys peaceful and undisturbed possession of the Real Property leased thereby, (b) the Company party as lessee to such Third Party Lease is not in breach or default under such lease, and no event has occurred or circumstance exists which, with the delivery of notice, passage of time or both, could result in a breach or default, and the Company party thereto has paid all rent and other amounts due and payable under such Third Party Lease (including Taxes) through the Closing Date, (c) no Company Party has received nor given any written notice of any breach or default under such lease, no other party to such Third Party Lease is in breach or default thereof, and no other party to such Third Party Lease has exercised or threatened to exercise any termination rights with respect thereto and (d) the Company party to such Third Party Lease has not subleased, assigned or otherwise granted to any Person the right to use or occupy all or any portion of the Real Property leased thereby. Except as set forth on Schedule 5.19, no Company Party has received any notice of any, and to the Company Parties' Knowledge there are no: (i) material violations of building codes and/or zoning ordinances or other governmental or regulatory Laws affecting any Leased Real Property, (ii) existing, pending or threatened condemnation proceedings affecting the Leased Real Property, (iii) existing, pending or threatened zoning, building code or other moratorium proceedings, or similar matters which would reasonably be expected to affect the ability to operate the Leased Real Property as currently operated by the Company party to the applicable Assigned Lease. The Leased Real Property is in an operating condition and state of maintenance and repair, normal wear and tear excepted, that is sufficient to conduct the business of the Companies and the ownership or use of any of their properties or assets in substantially the manner currently conducted.

5.20 Environmental.

Except as set forth on Schedule 5.20, the Company Parties' and, to the Knowledge of Company Parties, the 5.20.1 Real Property Owners', ownership, use and operation of their Real Property are in compliance in all material respects with the requirements of all applicable Environmental Laws. All Environmental Permits required for the ownership, use and operation of the properties, assets and Real Property have been timely obtained and are in full force and effect, and the Companies' ownership, use and operation of the properties, assets and Real Property are in material compliance with, all such Environmental Permits. There are no Actions pending or, to the Company Parties' Knowledge, threatened relating to an actual or alleged violation of Environmental Laws, any remediation obligations pursuant to Environmental Laws, any Environmental Permits (or the failure to have an Environmental Permit), personal injury or property damage claims relating to a release or threatened release of Hazardous Materials, or response, removal or remedial costs pursuant to CERCLA or any similar state Law, in each case on or with respect to the properties, assets and Real Property. No Company is operating or required to be operating its properties, assets or Real Property pursuant to any compliance order, schedule, decree or agreement, any consent decree, order or agreement, or corrective action decree, order or agreement issued or entered into as a result of a violation or an alleged violation of any Environmental Law. To the Company Parties' Knowledge, there are and have been no Hazardous Materials that have been disposed of or released in violation of Environmental Laws on, in, from or under the properties, assets and Real Property of the Companies, or with respect to the operations of the Business, that have resulted in or could reasonably be expected to result in a violation of any Environmental Law, any claim of exposure to or damage from any Hazardous Material, or in a liability under Environmental Law to perform any remediation, removal, response, restoration, abatement, investigation or monitoring. Except as identified in the environmental reports set forth on Schedule 5.20.2, no underground or above ground storage tanks, landfills, waste disposal areas, wastewater surface impoundments or lagoons, asbestos, or polychlorinated biphenyls septic systems, septic tanks or leachfields, drywells, monitoring wells, lead paint, or treatment systems of any kind, on, at, or underneath any of the Real Property. No Company is a party to any Contract pursuant to which it is obligated to indemnify any other Person with respect to, or be responsible for any claims or violations, obligations or liabilities pursuant to, or in violation of, Environmental Law.

Capital Pumping

5.20.2 True, correct and complete copies of all environmental assessments, audit reports and similar studies or analyses relating to any Real Property owned or leased by any Company Party that are in the possession of any Company Party have been delivered to Buyers. <u>Schedule 5.20.2</u> sets forth a list of all such assessments, audit reports and similar studies or analyses.

5.20.3 The Company Parties acknowledge and agree that prior to the Closing Date, no Buyer or any of its Affiliates (a) has ever been in control of any Company or any Affiliate of any Company, the Purchased Assets, any Real Property, or the operations or activities related to the Business, or any Company's affairs or the affairs of any Affiliate of any Company, and (b) has the capacity through the provisions of this Agreement or any other document contemplated hereunder or otherwise to influence any Company's conduct, or the conduct of or any Affiliate of any Company, with respect to the ownership, operation or management of the properties, assets and Real Property of the Companies, or the operations or activities related to the Business, or the affairs of any Company or of any Affiliate of any Company, or compliance (or not) with Environmental Laws.

5.21 Intellectual Property.

5.21.1 <u>Schedule 5.21</u> contains a true, correct and complete list of all of the trademarks, service marks trade dress and trade names, pending trademark and service mark registration applications, and intent-to-use registrations or similar reservations of marks, internet domain names and URLs and rights in computer software (other than software that is off the shelf or non proprietary) owned or used by the Companies in the conduct of the Business. Except as set forth on <u>Schedule 5.21</u>: (a) the Companies own and possess all right, title and interest in and to, or possesses the valid and enforceable right to use, the Intellectual Property used in the conduct of the Business, (b) during the six-year period prior to the date of this Agreement, no Company has received any written notices of material infringement or misappropriation from any third party with respect to any Company's use of any Intellectual Property, (c) to the Company Parties' Knowledge, no third party is materially infringing or misappropriating any registered Intellectual Property owned by any Company and (d) to the Company Parties' Knowledge, no Company is infringing on any registered Intellectual Property owned by a third party.

5.21.2 Except as set forth on <u>Schedule 5.21</u>, the Companies, the conduct of the Business and all products and services of the Companies and the use thereof have not infringed or otherwise violated, and do not infringe or otherwise violate, any Intellectual Property rights of any Person in any material respect. No Company is the subject of any pending Action that (a) alleges a claim of infringement, misappropriation, dilution or violation of any Intellectual Property rights of any Person, and no such claim has been asserted or threatened against any Company at any time during the past six years or (b) challenges the ownership, use, patentability, registration, validity or enforceability of any Owned Intellectual Property. Within the six years prior to the date of this Agreement, no Person has notified any Company Party that any of such Person's Intellectual Property rights are infringed, misappropriated or otherwise violated by any Company in any material respect or that any Company requires a license to any of such Person's Intellectual Property rights in order for such Company to continue activities that are material to the Business as currently conducted or as proposed to be conducted.

36

5.21.3 All software, systems, servers, computers, hardware, firmware, middleware, networks, data communications lines, routers, hubs, switches and all other information technology equipment, and all associated documentation ("<u>IT Assets</u>"), in each case, used by any Company in connection with the operation of the Business are operational, fulfill the purposes for which they were acquired or developed, have security, back-ups and disaster recovery arrangements in place and hardware and software support, and maintenance which are at a commercially reasonable level for the current needs of the Business. The Companies have taken commercially reasonable measures to implement disaster recovery and security plans, and have taken reasonable steps consistent with or exceeding industry standards for the field of the Business to safeguard the availability, security and integrity of the IT Assets and all data and information stored thereon, including from unauthorized access and code such as from material bugs, errors, disabling codes, spyware, Trojan horses, worms or other malicious code intended to cause any unauthorized disrupting or disabling of the operation of, or provision of unauthorized access to, a computer system or network or other device on which such code is stored or installed. The Companies have maintained all required licenses and service Contracts, including the purchase of a sufficient number of license seats for all software, with respect to the material IT Assets in the ordinary course of business consistent with past practices. Except as set forth on <u>Schedule 5.21.3</u>, to the Company Parties' Knowledge, the IT Assets have not suffered any security breach or material failure within the past three years.

5.21.4 To the Company Parties' Knowledge, no third party is infringing upon, misappropriating or otherwise violating any Intellectual Property owned by any Company. Within the past six years, no Company Party has sent any written notice, charge, complaint, claim or other written assertion asserting or threatening to assert any Action against any Person involving or relating to any Intellectual Property of any Company.

5.21.5 The Companies have taken commercially reasonable measures to protect the confidentiality of all trade secrets and any other material confidential information of the Companies and the Business (and any confidential information owned by any Person to whom any Company has a confidentiality obligation). No trade secrets or any other material confidential information of any Company or of any Person to whom any Company owes a duty of confidentiality has been disclosed by any Company Party to any Person other than pursuant to a written agreement restricting the disclosure and use of such trade secrets or any other material confidential information by such Person. No current or former founder, officer, director, shareholder, general partner, limited partner, member, employee, contractor, or consultant of any Company has any right, title or interest, directly or indirectly, in whole or in part, in any Owned Intellectual Property. The Companies have obtained from all Persons (including all current and former founders, officers, directors, shareholders, general partners, limited partners, members, employees, contractors, consultants and agents) involved in the development of any Intellectual Property for the Companies valid and enforceable written assignments of any such Intellectual Property to the Companies or such Intellectual Property has been assigned to the Companies by operation of law, except in each case where the failure to do so would not adversely affect the Companies or the Business in any material respect. To the Company Parties' Knowledge, no such Person has contested the ownership of or any right, title, or interest to, any material Intellectual Property by the Companies.

Capital Pumping

5.21.6 The Companies have a privacy policy regarding the collection, use, and disclosure of personal information in connection with the operation of the Business and in the Business's possession, custody, or control, or otherwise held or processed on their behalf and is and has been in compliance with such privacy policy. True, correct and complete copies of such privacy policies have been provided to Buyers. The Companies have complied at all times with all applicable Laws regarding the collection, use, storage, transfer, or disposal of personal information. The Companies are in compliance with the terms of all Contracts to which any Company is a party relating to data privacy, security, or breach notification (including provisions that impose conditions or restrictions on the collection, use, storage, transfer, or disposal of personal information). The Companies do and have posted a privacy policy in a clear and conspicuous location on all websites and applicable mobile applications owned or operated by the any Company. No Company has experienced any loss, damage, or unauthorized access, disclosure, use, or breach of security of any personal information in its possession, custody, or control, or otherwise held or processed on its behalf.

5.22 Conduct of the Business.

5.22.1 Since January 1, 2019 and except as set forth on <u>Schedule 5.22</u>, (a) the Companies have conducted the Business only in the ordinary course of business consistent with past practices, (b) none of the Companies have materially changed their collections practices, (c) no rights to receive payment in respect of work in process have been waived by any Company, other than in the ordinary course of business consistent with past practices, (d) except as listed on <u>Schedule 5.22</u>, there has been no damage, destruction or loss (whether or not covered by insurance) which exceed \$100,000 in the aggregate to the Business or any assets or properties used in the Business, the Real Property, or rights of any Company, (e) there has been no material adverse change, individually or in the aggregate, in the assets, liabilities, financial condition, operations or results of operations, employee, client or customer relations, or the Business, (f) none of the Companies has amended its Organizational Documents, (g) none of the Companies has entered into or amended any Contract that is or that would be required to be included in <u>Schedule 5.12</u>, (h) none of the Companies has (i) acquired (by merger, consolidation, acquisition of stock or assets or otherwise) or organized any Person, (ii) acquired any rights, assets or properties other than in the ordinary course of business or (iii) acquired any equity interest or other securities of any Person, or (i) none of the Companies has incurred any Indebtedness, guaranteed any liability of any other Person, or mortgaged or encumbered any of its assets to become subject to any Liens. To the Company Parties' Knowledge, there have been no developments since January 1, 2019 which are reasonably likely to constitute a Material Adversely Affect.

5.22.2 To the Company Parties' Knowledge, no customer or group of affiliated customers who accounts for more than \$250,000 of annual revenue to the Business intends to cease doing business with the Companies or intends to substantially reduce the amount of business such customer or customers have done with the Companies, other than dates of completion of a project. To the Company Parties' Knowledge, the Companies' relationships with customers of the Business have been conducted consistent with industry standards. No supplier or vendor has canceled or otherwise terminated, or to the Company Parties' Knowledge, threatened to cancel or otherwise terminate its relationship with any Company, or to decrease its services or its supply of goods to any Company.

5.23 Affiliate Matters. Except as set forth on <u>Schedule 5.23</u>, no Company, or, to the Company Parties' Knowledge, no officer, Employee, Former Employee, shareholder, manager, member, former manager or member, general partner, limited partner or Affiliate of any Company, nor any individual in the immediate family of any of the foregoing, is a party to any Contract with any Entity Company Party or Affiliate of any Entity Company Party, or has any interest in any Real Property or personal property used by any Company or in the Business.

38

5.24 Creditors, Solvency and Bankruptcy. No Company Party has the intent to hinder, delay, defraud, or avoid any obligation to any past, present, or future creditor in the transactions contemplated by Transaction Documents. No Company Party is insolvent and no Company will be rendered insolvent as a result of the transactions contemplated in the Transaction Documents. No Company Party has initiated, and no Company Party intends to initiate with respect to itself or any other Company Party as debtor, nor has it had initiated against it nor does it expect to have initiated against it as debtor, any proceeding under federal or any state's bankruptcy, insolvency or similar Laws. As a result of the transactions contemplated in the Transaction Documents, no Company Party will become unable to pay in full all of its debts as they become due.

5.25 Bank Accounts. <u>Schedule 5.25</u> sets forth an accurate and complete list of the names of all banks and financial institutions in which each Company has an account, deposit, safe deposit box, line of credit or other loan facility or relationship, or lock box or other arrangement for the collection of accounts receivable, the corresponding number of each such account or safe deposit box and the names of all persons holding check-signing or withdrawal powers or other authority with respect thereto. Also set forth on <u>Schedule 5.25</u> are the names of all persons, if any, holding powers of attorney from any Company and a summary statement of the terms thereof.

5.26 Accounts Receivable; Warranty Claims; Accounts Payable; Excluded Insurance Proceeds. <u>Schedule 5.26</u>, to be delivered at the Closing, will set forth a true, correct and complete list of all of Pre-Closing Accounts Receivable, Pre-Closing Warranty Claims, Pre-Closing Accounts Payable, and Excluded Insurance Proceeds as of the Closing Date.

Information Supplied. None of the information supplied by the Company Parties for inclusion in the Proxy Statement 5.27 will, in the case of the definitive proxy statement (and any amendment or supplement thereto), at the date of mailing of such definitive proxy statement (and any amendment or supplement thereto) and at the time of the meeting of CPHI's stockholders convened for the purpose of soliciting stockholder approval of the Equity Financing and other applicable matters (the "CPHI Stockholders Meeting"), contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein in light of the circumstances under which they are made, not misleading; provided, however, that CPHI shall have provided the Company Parties with the opportunity to review and comment on a draft of the definitive proxy statement (and any amendment or supplement thereto) at least three days prior to the date on which such proxy statement (or amendment or supplement thereto, as applicable) is filed with the SEC, or in the case of a registration statement, at the time such registration statement becomes effective. None of the information supplied by the Company Parties expressly for inclusion in any of the filings made by CPHI with the SEC will, at the time filed with the SEC, or in the case of a registration statement, at the time such registration statement becomes effective, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading; provided, however, that CPHI shall have provided the Company Parties with the opportunity to review and comment on a draft of any such filings with the SEC (and any amendments or supplements thereto) at least three days prior to the date on which such filings with the SEC (or amendments or supplements thereto, as applicable) are made.

5.28 Full Disclosure. With respect to the Company Parties or the Business, neither this Agreement nor any other Transaction Document (including the exhibits and schedules hereto and thereto) contains any untrue statement of a fact or omits to state a material fact necessary to make the statements contained therein, in light of the circumstances in which they are made, not misleading. The representations and warranties of the Company Parties shall not be affected or deemed waived by reason of any investigation made by or on behalf of any Buyers (including but not limited to by any of its advisors, consultants, or representatives) or by reason of the fact that such Buyers or any of such advisors, consultants or representatives knew or should have known that any such representation or warranty is or might be inaccurate.

39

5.29 Brokerage. Except for WoodRock Securities, LP, no broker, finder, investment banker or other Person is entitled to any brokerage fee, finders' fee or other similar commission, for which Buyers or the Companies could be liable in connection with the transactions contemplated by this Agreement.

5.30 No Additional Representations or Warranties . Except as provided in this <u>Article 5</u>, no Company Party, nor any of their directors, officers, employees, partners, members, managers or representatives has made, or is making, any representation or warranty whatsoever to Buyers or their Affiliates, or their directors, officers, employees, stockholders, partners, members, managers or representatives.

ARTICLE 6

REPRESENTATIONS AND WARRANTIES OF BUYERS

The Buyers hereby represent and warrant to the Company Parties, jointly and severally, as of the date hereof and as of the Closing Date, as follows:

6.1 Organization. BBCP is a corporation duly organized, validly existing and in good standing under the Laws of the State of Colorado. CPHA LLC is a limited liability company duly organized, validly existing and in good standing under the Laws of the State of Delaware. Each Buyer has all requisite power and authority to own or lease its properties and assets and to operate and carry on its business as it is now being conducted. Each Buyers is duly licensed or qualified to do business and (where applicable) is in good standing as a foreign corporation in each jurisdiction in which the ownership, operation or lease of its property or the character of its activities is such as to require it to be so licensed or qualified or in good standing, as applicable, except where the failure to be so licensed or qualified or have a material adverse effect on such Buyer or its business.

6.2 Due Authorization. Each Buyer has all requisite power and authority to execute and deliver this Agreement and each other Transaction Document to which it is a party to consummate the transactions contemplated hereby and thereby and to perform all obligations to be performed by them hereunder and thereunder. The execution and delivery of this Agreement and the other Transaction Documents to which it is a party by each Buyer and consummation of the transactions contemplated hereby and thereby have been duly and validly authorized and approved by the Board of Directors of BBCP and the Managing Member of CPHA LLC and no other organizational proceeding on the part of either Buyer is necessary to authorize this Agreement and the Transaction Documents to which such Buyer is a party. This Agreement has been, and each of the Transaction Documents to which each Buyer is a party has been or will be at or prior to Closing, duly and validly executed and delivered by such Buyer and (assuming this Agreement constitutes a legal, valid and binding obligation of such Buyer, enforceable against it in accordance with its terms, subject to applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and similar Laws affecting creditors' rights generally and subject, as to enforceability, to general principles of equity.

40

6.3 No Conflict. Neither the execution, delivery and performance of this Agreement nor any other Transaction Document to which each Buyer is a party, and the consummation by such Buyer of the transactions contemplated hereby and thereby do not and will not, indirectly or directly, (a) violate any provision of, or result in the breach of, any applicable Law to which such Buyer is subject or by which any property or asset of such Buyer is bound or (b) conflict with, result in a breach or violation of or constitute a default under any of the provisions of the Organizational Documents of such Buyer.

6.4 Brokerage. Except for UBS Securities, LLC, Stifel, Nicholas and Company, Incorporated, and Baird Global Investment Banking, no broker, finder, investment banker or other Person is entitled to any brokerage fee, finders' fee or other similar commission, for which the Company Parties could be liable in connection with the transactions contemplated by this Agreement.

6.5 No Additional Representations or Warranties . Except as provided in this <u>Article 6</u>, neither the Buyers nor any of their Affiliates (including CPHI), nor any of their respective directors, officers, employees, stockholders, partners, members, managers or representatives has made, or is making, any representation or warranty whatsoever to any Company Party or their Affiliates, or its or their directors, officers, employees, stockholders, partners, members, managers or representatives.

ARTICLE 7 COVENANTS OF THE COMPANY PARTIES

7.1 Conduct of Business. From the date hereof through the Closing, the Company Parties shall (a) operate the Business in the ordinary course and substantially in accordance with past practice, (b) pay all accounts payable as such amounts become due in the ordinary course of business consistent with past practices, (c) preserve and protect its business organization, employment relationships, and relationships with customers, strategic partners, suppliers, distributors, landlords and others having dealings with it in the ordinary course of business consistent with past practices, (d) maintain its assets, properties, books of account and records in the ordinary course of business consistent with past practices, (e) maintain all assets used in the Business, Owned Real Property and Leased Real Property in their present condition, except for ordinary wear and tear and damage by unavoidable casualty, (f) keep in full force and effect insurance comparable in amount and scope of coverage to insurance carried with respect to the Business as of the date hereof, (g) comply in all material respects with all Licenses and Permits and all applicable Laws, (h) perform in all material respects all obligations under Contracts relating to or affecting the Business so as to prevent any default thereunder, (i) maintain its books and records in the ordinary course of business consistent with its past practices, (j) otherwise preserve the goodwill and ongoing operations of its Business and (k) not take any action or fail to take any action that could result in any of the conditions set forth in <u>Section 4.2</u> not being satisfied or that would otherwise be reasonably expected to prevent or delay the consummation of the transactions contemplated by this Agreement. Without limiting the generality of the foregoing, the Company Parties shall not:

7.1.1 change or amend the Organization Documents of CP, ASC or MCS or waive any material provision thereof;

7.1.2 materially amend, materially modify, terminate (excluding any expiration in accordance with its terms), waive any material provision in or, except in the ordinary course of business consistent with past practices, enter into any Contract of a type required to be listed on <u>Schedule 5.12</u> or any material insurance policy required to be listed on <u>Schedule 5.16</u>;

41

7.1.3 become legally committed to make any capital expenditures in excess of \$250,000, in the aggregate;

7.1.4 grant or allow any Lien on any properties or assets used in the Business (except for Permitted Liens);

7.1.5 sell, assign, transfer, convey, lease or otherwise dispose of any properties or assets used in the Business, or any interest in any Real Property, other than in the ordinary course of business consistent with past practices;

7.1.6 make any material changes or modifications to any insurance policies; <u>provided</u>, <u>however</u>, that the Company Parties may and shall use commercially reasonable efforts to extend their existing insurance policies for a period of not more than 90 days past their expiration date;

7.1.7 except as required by Law or pursuant to existing Employee Benefit Plans or Contracts, (a) other than in the ordinary course of business consistent with past practices, increase or accelerate the compensation or fees payable to any Employee or independent contractor who receives, or in the current calendar year is likely to receive, total compensation in excess of \$100,000, (b) other than in the ordinary course of business consistent with past practices, hire or terminate the employment or engagement of any Employee or independent contractor who receives, or in the current calendar year is likely to receive, total compensation in excess of \$100,000, (b) other than in the ordinary course of business consistent with past practices, hire or terminate the employment or engagement of any Employee or independent contractor who receives, or in the current calendar year is likely to receive, total compensation in excess of \$100,000, (c) adopt, enter into, terminate or materially amend any Employee Benefit Plan (or plan or arrangement that would be an Employee Benefit Plan if in effect on the date hereof), (d) enter into, terminate or amend any collective bargaining agreement, works council agreement, or other agreement for the labor representation of Employees or (e) pay or agree to pay any Employee or independent contractor any change of control or similar transaction bonuses in connection with the transactions contemplated herein, other than the Employee Bonus Plan;

7.1.8 acquire by merger or consolidation with, or merge or consolidate with, or purchase all or substantially all of the assets of, any Person or division thereof in a single transaction or a series of related transactions;

7.1.9 incur any Indebtedness, except in the ordinary course of business consistent with past practices;

7.1.10 make any loans or advances of money to any Person (other than the Companies), except for advances to Employees for expenses incurred in the ordinary course of business consistent with past practices;

7.1.11 sell, license, transfer or otherwise dispose of, any material Intellectual Property of the Companies, except in the ordinary course of business consistent with past practices;

7.1.12 change or modify in any material respect any Company's ordinary course credit, collection or payment policies, procedures or practices, including acceleration of billing, invoicing or collections of amounts due or to become due from any customer, fail to pay or delay payment of payables or other liabilities in any material respect, or otherwise materially change the manner in which any Company deals with customers, suppliers or vendors in the ordinary course of business consistent with past practices; or

7.1.13 enter into any agreement, or otherwise become obligated, to do any action prohibited under this Section 7.1.

42

7.2 Pending Insurance Claims. From and after the date hereof, to the extent any of the properties or assets used in the Business are not in good working order or are damaged prior to the Closing Date, the Company Parties shall cooperate with Buyers in processing claims with respect to such damage under the Company Parties' insurance policies, and the proceeds of such claims shall be paid to Buyers upon receipt, excepting the Excluded Insurance Proceeds. The Company Parties agree and covenant that they will use commercially reasonable efforts to dutifully prosecute the defense of and pursue any and all available insurance claims relating to the Business, the properties or assets used in the Business, the Real Property or the Employees.

7.3 HSR Act.

7.3.1 In connection with the transactions contemplated by this Agreement, the Companies shall (and, to the extent required, shall cause its Affiliates to) (a) comply promptly, but in no event later than five Business Days after the date hereof, with the notification and reporting requirements of the HSR Act and timely make all further filings pursuant thereto that may be necessary (including resubmit filings that are rejected for any reason whatsoever by the relevant Governmental Authority) and (b) request early termination of the waiting period under the HSR Act. The Companies shall promptly comply with any Antitrust Information or Document Request made of the Companies.

7.3.2 The Companies shall exercise their commercially reasonable efforts to (a) obtain termination or expiration of the waiting period under the HSR Act and such other approvals, consents and clearances as may be necessary, proper or advisable under any foreign antitrust or competition Laws, in each case, as soon as practicable, (b) furnish to Buyers all information reasonably required for any application or other filing to be made pursuant to any Law in connection with the transactions contemplated by this Agreement (including, to the extent permitted by Law, responding to any reasonable requests for copies of documents filed with the Companies' prior filings), and (c) otherwise reasonably cooperate with Buyers in connection with any filing and in connection with resolving any investigation or other inquiry of any Governmental Authority.

7.3.3 The Company Parties shall promptly furnish to Buyers copies of any notices or written communications received or given by them or any of their Affiliates from or to any third party or any Governmental Authority with respect to the transactions contemplated by this Agreement, and the Company Parties shall permit counsel to Buyers an opportunity to review in advance, and the Companies shall consider in good faith the views of such counsel in connection with, any proposed written communications by the Companies or their Affiliates to any third party or any Governmental Authority concerning the transactions contemplated by this Agreement. The Companies shall provide Buyers and their counsel the opportunity, on reasonable advance written notice, to participate in any substantive meetings or discussions, either in person or by telephone, between the Companies and any of their respective Affiliates, agents or advisors, on the one hand, and any third party or any Governmental Authority, on the other hand, concerning or in connection with the transactions contemplated hereby.

7.4 Financing Cooperation. The Company Parties shall provide such assistance with the Proxy Statement, any SEC filings necessary for the Equity Financing and any Debt Documents, if applicable, as is reasonably requested by CPHI, including but not limited to the following: (a) as promptly as reasonably practicable, furnishing CPHI with financial and other pertinent information regarding the Companies as may be reasonably requested by CPHI, (b) providing and executing customary consents with respect to the Proxy Statement and the Debt Documents, and (c) preparing and furnishing to CPHI, as promptly as practicable, all Required Financial Information. The Company Parties will, upon reasonable written request of CPHI, use their best efforts to update any Required Financial Information (to the extent it is available) to ensure that such Required Financial Information, when taken as a whole, does not contain as of the time provided, giving effect to any supplements, any untrue statement of material fact or omit to state any material fact necessary in order to make the statements contained therein not materially misleading. As soon as practicable but in any event no later than May 14, 2019, but only if the Closing shall not have occurred on or prior to such date, the Company Parties shall provide CPHI with the information specified in clause (b) of the definition of

"Required Financial Information." On the Closing Date, Buyers shall reimburse the Sellers for their reasonable out-of-pocket expenses incurred in connection with the preparation of such information and the review of such information by the Companies' Audit Firm. The Company Parties will consent to the use of all of their logos in connection with the Equity Financing and Debt Financing; <u>provided</u>, <u>however</u>, that such logos are used solely in a manner that is not reasonably likely to harm or disparage the Company Parties or the reputation or goodwill of the Company Parties.

7.5 Termination of Affiliate Agreements. On or prior to the Closing Date, the Company Parties shall take all actions necessary to terminate, and shall cause to be terminated, all Affiliate Agreements (other than the indemnification and other provisions thereof that expressly survive any such termination), other than Contracts for the lease, occupancy and use of the real property described on <u>Annex A</u>.

7.6 Authority. From and after the Closing, the Sellers shall not represent that they have any affiliation with or authority to bind any Company or Buyer.

7.7 **Dispute Cooperation**. From and after the date hereof, the Company Parties agree to provide such cooperation as Buyers or their counsel may reasonably request in connection with any Action arising out of this Agreement or the other Transaction Documents and the transactions contemplated hereby and thereby which are hereafter pending or threatened with respect to the period prior to Closing and to which Buyers is a party. Such cooperation shall include, but not be limited to, making the Sellers and their Affiliates available upon the reasonable request and at the expense of Buyers for any out-of-pocket costs incurred in connection therewith to consult with and assist Buyers and their counsel in connection with any such Action and to prepare for and testify in any such Action, including depositions, trials and arbitration proceedings.

7.8 Updated Information. Without limiting Buyers' rights hereunder, the Company Parties shall promptly notify Buyers in writing of (a) any event, condition or circumstance occurring, or failing to occur, from the date hereof through the Closing Date, which occurrence or failure to occur would constitute, or would reasonably be expected to result in a violation or breach of this Agreement, and (b) any event, occurrence, transaction or other item (i) which would have been required to have been disclosed on any schedule or statement delivered hereunder had such event, occurrence, transaction or item existed on the date hereof, other than items arising in the ordinary course of business consistent with past practices or (ii) that is necessary to correct any information in the Company Parties' disclosure schedules. No later than two Business Days prior to Closing, the Company Parties shall deliver to Buyers Schedule 5.14.2-B which schedule shall be incorporated into <u>Sections 5.14.2</u> by reference and made a part hereof. On the Closing Date, the Company Parties shall deliver to Buyers Schedule 5.14.12-B and Schedule 5.26, which shall be incorporated into Section 5.14.12 and Section 5.26, respectively, by reference and made a part hereof. No supplement or amendment to any schedule of the Company Parties (including the delivery of Schedule 5.14.2-B, Schedule 5.14.12-B and Schedule 5.26 to Buyers) nor any notifications or disclosure made to Buyers pursuant to the foregoing requirements shall have any effect for purposes of determining satisfaction of the conditions set forth in Section 4.2.1. For purposes of determining whether there is any misrepresentation or breach of warranty, covenant or agreement under Article 11 or Article 12, any schedules, supplements or amendment to any schedule of the Company Parties (other than Schedule 5.14.2-B and Schedule 5.14.12-B) and any notifications or disclosures delivered by the Company Parties to Buyers shall be deemed to include only the information contained therein on the date of this Agreement.

44

Exclusivity. Until the first to occur of the Closing or the earlier termination of this Agreement pursuant to Article 10, 7.9 the Company Parties will not, and will cause their respective Affiliates, directors, officers, partners, members, managers, employees, agents, consultants and other advisors and representatives not to, directly or indirectly: (a) solicit, initiate, encourage, knowingly facilitate any inquiry or the making of any proposal or offer, (b) enter into, continue or otherwise participate in any discussions or negotiations, (c) furnish to any Person any non-public information or grant any Person access to its properties, assets, books, contracts, personnel or records, (d) approve or recommend, or propose to approve or recommend, or execute or enter into, any letter of intent, agreement in principal, merger agreement, acquisition agreement, option agreement or other Contract, or (e) propose, whether publicly or to any director or stockholder, or agree to do any of the foregoing for the purpose of encouraging or facilitating any proposal, offer, discussions or negotiations; in each case relating to an Acquisition Proposal. "Acquisition Proposal" means any offer or proposal regarding a business combination transaction involving any Company Party or any other transaction to acquire all or any material part of the Business or any properties or assets used in the Business, other than in the ordinary course of business, or any amount of the equity securities of any Entity Company Party (whether or not outstanding), whether by merger, acquisition of assets, purchase of equity, tender offer or other similar transactions, other than with Buyers. The Company Parties will immediately cease and cause to be terminated any such negotiations, discussion or other communication, or Contracts (to the extent unilaterally terminable by the Company Parties without the counterparty's consent and without penalty) (other than with Buyers) with respect to the foregoing and will immediately (but in any event within five Business Days after the date of this Agreement) terminate any access of the type referenced in clause (c) above.

7.10 Release. In consideration for the transactions contemplated in this Agreement, effective as of the Closing, each Seller (for and on behalf of itself, himself or herself, and such Seller's family members, heirs, estates, representatives, trustees, executors, successors and assigns) hereby unconditionally and irrevocably acquits, remises, discharges and forever releases and discharges the Buyers and the Companies and their respective Affiliates, equityholders, partners, managers, members, employees, officers, directors, insurers, agents, successors and assigns (collectively, "<u>Releases</u>") from any and all claims, demands, rights, liabilities and causes of action of any kind or nature, known or unknown, arising prior to or through the date hereof, including, but not limited to, any claims, demands, rights, liabilities and causes of action arising or having arisen out of or in connection with the Purchased Interests. Notwithstanding the foregoing, nothing in this <u>Section 7.10</u> shall constitute a release of any Person from its obligations, covenants and agreements hereunder or under any Transaction Document or the enforceability of this Agreement, any Transaction Document or any of their respective terms or conditions.

7.11 Environmental Reports. The Company Parties shall undertake commercially reasonable efforts deliver to Buyers, no later than 5 days prior to Closing, a phase II environmental report covering the scope of soil and groundwater sampling as recommended by Terracon in that certain Phase 1 Environmental Site Assessment, dated December 5, 2018, project number 96187992, with respect to the real property located at 16113 North IH-35 Frontage Road, Salado, TX. After reviewing such environmental report, if more than five days prior to the Closing, Buyers may, in their sole and absolute discretion, elect to amend and replace <u>Annex A</u> hereto to delete such parcel from <u>Annex A</u>, and to require the Company Parties to terminate that certain Industrial Lease, dated January 1, 2017, between CLC 31, LLC and CP, as amended and restated through the Closing Date, and deliver to Buyers at or prior to the Closing, a Lease Termination and Release with respect thereto.

7.12 Leased Real Property Actions. On or prior to the Closing Date, with respect to each real property location described on <u>Annex A</u>, the Company Parties shall service (a) all grease traps, (b) oil/water separators, and (c) sewers, septic tanks, pipes and systems, in each case consistent with best practices for premises used in the manner the Company Parties have historically used such premises.

7.13 Deliveries Prior to Closing. At least five Business Days prior to Closing, the Company Parties shall deliver to Buyers <u>Schedule 3.7</u> and <u>Annex B</u>.

ARTICLE 8 COVENANTS OF BUYERS

8.1 HSR Act.

8.1.1 In connection with the transactions contemplated by this Agreement, Buyers shall (and, to the extent required, shall cause its Affiliates to) (a) comply promptly, but in no event later than five Business Days after the date hereof, with the notification and reporting requirements of the HSR Act and timely make all further filings pursuant thereto that may be necessary (including resubmit filings that are rejected for any reason whatsoever by the relevant Governmental Authority) and (b) request early termination of the waiting period under the HSR Act. Buyers shall promptly comply with any Antitrust Information or Document Request made of Buyers.

8.1.2 Buyers shall exercise their commercially reasonable efforts to (a) obtain termination or expiration of the waiting period under the HSR Act and such other approvals, consents and clearances as may be necessary, proper or advisable under any foreign antitrust or competition Laws, in each case, as soon as practicable, (b) furnish to the Companies all information reasonably required for any application or other filing to be made pursuant to any Law in connection with the transactions contemplated by this Agreement (including, to the extent permitted by Law, responding to any reasonable requests for copies of documents filed with Buyers' prior filings), and (c) otherwise reasonably cooperate with the Companies in connection with any filing and in connection with resolving any investigation or other inquiry of any Governmental Authority. In connection therewith Buyers shall not be required to make any divestitures, enter into any license or other agreements, or otherwise dispose of any of its businesses or assets, or the businesses or assets being acquired hereunder, in order to obtain HSR clearance and/or the expiration of any applicable waiting period pursuant to the HSR Act.

8.1.3 Buyers shall promptly furnish to the Companies copies of any notices or written communications received or given by Buyers or any of their Affiliates from or to any third party or any Governmental Authority with respect to the transactions contemplated by this Agreement, and Buyers shall permit counsel to the Companies an opportunity to review in advance, and Buyers shall consider in good faith the views of such counsel in connection with, any proposed written communications by Buyers and their Affiliates to any third party or any Governmental Authority concerning the transactions contemplated by this Agreement. Buyers agree to provide the Companies and their counsel the opportunity, on reasonable advance notice, to participate in any substantive meetings or discussions, either in person or by telephone, between Buyers and any of their Affiliates, agents or advisors, on the one hand, and any third party or any Governmental Authority, on the other hand, concerning or in connection with the transactions contemplated hereby.

46

8.1.4 Buyers shall advance and pay 100% of all fees payable to the Antitrust Authorities in connection with the transactions contemplated by this Agreement. Each party shall bear its own legal or advisor fees in connection with any filings, Actions or litigation under Section 7.3 or this Section 8.1.

8.2 R&W Insurance Policy. Buyers or their Affiliate have obtained and bound coverage under a representations and warranties insurance in the form attached hereto as <u>Annex C</u> (such policy, the "<u>R&W Insurance Policy</u>"). All premiums, underwriting fees and brokers' commissions related to such R&W Insurance Policy shall be borne by Buyers or such Affiliates. Prior to the Closing, the Companies shall provide commercially reasonable assistance to Buyers, if applicable, to the extent required to maintain bound coverage under the R&W Insurance Policy.

8.3 Financing. In connection with the transactions contemplated under this Agreement and the Equity Financing, CPHI will file with the SEC, if required, a definitive proxy statement to be sent to the stockholders of CPHI relating to the CPHI Stockholders Meeting (such proxy statement, together with any amendments or supplements thereto, the "Proxy Statement") for the purpose of soliciting proxies from CPHI stockholders to vote at the CPHI Stockholders Meeting to approve (a) the Equity Financing and (b) any other proposals, the approval of which CPHI deems necessary or desirable to consummate the Equity Financing. Buyers shall work diligently and undertake commercially reasonable efforts to consummate the Equity Financing and Debt Financing prior to Closing.

8.4 Dispute Cooperation. From and after the date hereof, the Buyers agree to provide such cooperation as Sellers or their counsel may reasonably request in connection with any Action arising out of this Agreement or the other Transaction Documents and the transactions contemplated hereby and thereby which are hereafter pending or threatened with respect to the period prior to Closing and to which a Buyer is a party at the Sellers' sole cost and expense.

8.5 Pre Closing Insurance Premiums. Buyers shall pay over to the Sellers, within ten Business Days after the Closing, the Prepaid Insurance Amount as of the Closing Date. The Sellers acknowledge that upon Buyer's payment of such amount to the account set forth on <u>Annex B</u>, Buyer's obligations pursuant to this <u>Section 8.5</u> shall be fully satisfied, and the Sellers release Buyers from any obligation or responsibility with respect to the payment thereof.

ARTICLE 9 JOINT COVENANTS

9.1 Support of Transaction. Without limiting any covenant contained in <u>Article 7</u> or <u>Article 8</u>, the parties shall (a) use commercially reasonable efforts to assemble, prepare and file any information (and, as needed, to supplement such information) as may be reasonably necessary to obtain as promptly as practicable all governmental and regulatory consents required to be obtained in connection with the transactions contemplated hereby, (b) use commercially reasonable efforts to obtain all material consents and approvals of third parties that any of the parties are required to obtain in order to consummate the transactions contemplated herein, and (c) take such other action as may reasonably be necessary or as another party may reasonably request to satisfy the conditions set forth in <u>Article 4</u> or otherwise to comply with this Agreement and to consummate the transactions contemplated hereby as soon as practicable (but in any event prior to the Termination Date).

9.2 Further Assurances. Each party hereto agrees that, from time to time after the Closing Date, it will furnish, or cause to be furnished, upon request to each other such further information, execute and deliver, or cause its Affiliates to execute and deliver, such further instruments, and take (or cause its Affiliates to take) such other action, as may be reasonably necessary to carry out the purposes and intents of this Agreement and the transactions contemplated herein.

47

9.3 Transfer Taxes. All Transfer Taxes incurred in connection with the transfer and conveyance of the Transferred Assets shall be borne by the Sellers. If Buyers, or after Closing, the Companies shall be required to make the filings, reports, or Tax Returns, or pay Transfer Tax shown on such Tax Returns with respect to the Transferred Assets (the "<u>Payor</u>"), the Payor shall notify the Sellers in writing of the Transfer Tax shown on such filings, reports and Tax Returns and how such Transfer Tax was calculated. The Sellers shall promptly reimburse the Payor the amount of such Transfer Tax attributable to the Transferred Assets in immediately available funds within 10 Business Days of receipt of such notice.

9.4 Allocation of Taxes.

Property Taxes. All Property Taxes (for the avoidance of doubt, other than Transfer Taxes) levied with respect 9.4.1 to the Assets for the Straddle Period shall be apportioned between the Buyers and the Sellers based on the number of days of such Straddle Period included in the Pre-Closing Period and the number of days of such Straddle Period included in the Post-Closing Period. The Sellers shall be liable for the proportionate amount of such Property Taxes that is attributable to the Pre-Closing Period, and the Buyers shall be liable for the proportionate amount of such Property Taxes that is attributable to the Post-Closing Period. With respect to tangible personal property that is owned by the Companies as of the Closing Date, Sellers shall be credited with any Property Taxes paid by the Companies prior to the Closing Date in respect of Property Taxes not yet assessed by the applicable Governmental Authority. Property Taxes that relate to a fiscal year ending prior to the Closing Date shall be the responsibility of the Sellers. All installments of Property Taxes having a due date before the Closing Date shall be paid by each Company before the Closing Date, and the Buyers will cause the Companies to pay all installments of Property Taxes having a due date on or after the Closing Date. At the Closing, the net amount of all Property Tax adjustments computed according to this Section 9.4.1, based upon which party is responsible for such Property Taxes and which party is to pay such Property Taxes, shall be added to or deducted from the Purchase Price. If current tax statements for any Property Taxes for a Straddle Period are not available as of the Closing Date, the prior year's tax statements will be used for purposes of making an estimated proration at the Closing, and a final proration will be made promptly when the current tax statements for such Property Taxes are received. The Sellers shall remit to the Buyers any additional amounts due to the Buyers in such regard within 10 Business Days of receipt of written notice thereof from the Buyers (which notice shall include reasonable evidence of the Taxes owed). The Buyers shall remit to the Sellers any excess amounts previously remitted hereunder within 10 Business Days of receipt of written notice thereof from the Sellers (which notice shall include reasonable evidence of the Taxes owed). If the Sellers, from and after the Closing, receive any bill, assessment or other notice of any such Property Taxes due for any Straddle Period, the Sellers shall promptly forward a copy of such bill, assessment or other notice to the Buyers.

9.4.2 <u>Other Taxes</u>. For purposes of this Agreement, in the case of any Straddle Period, (a) Taxes of the Companies (other than Property Taxes) allocable to the Pre-Closing Period shall be computed as if such taxable period ended prior to the Closing Date, and (b) any Taxes in the form of interest, penalties, additions to tax or additional amounts that relate to Taxes for any period, or a portion of any period, ending before the Closing Date shall be treated as occurring prior to the Closing Date whether such items are incurred, accrued, assessed or similarly charged on, before or after the Closing Date. Notwithstanding any provision of this Agreement to the contrary, and for the avoidance of doubt, all amount deducted from the Purchase Price (excluding Escrow Amount and the Equipment Adjustment Amount) pursuant to <u>Section 3.3</u> shall be accrued and reported as an expense of the Companies allocated to the Pre-Closing Period, and none of the parties will take a position inconsistent therewith for Tax purposes.

48

9.4.3 Tax Returns and Payments.

(a) At the Sellers' sole expense, the Sellers will prepare or cause to be prepared all income Tax Returns of any Company for taxable periods ending prior to or on the Closing Date for which the original due date (including extensions) is after the Closing Date and any short- period returns for taxable periods ending prior to the Closing Date, which includes, for clarity, the final IRS Form 1065 of ASC and CP (the "Seller Returns"), and such Seller Returns will include the income, operations or property of the Company, for the Pre-Closing Period. For the avoidance of doubt, for federal income tax purposes, the taxable year of ASC and CP will be deemed to have ended on the day immediately before the Closing Date, and none of the parties will take a position inconsistent therewith for Tax purposes. Each such Seller Return will be prepared in a manner consistent with the Companies' past practice except as otherwise required by Law or this Agreement, and in accordance with the provisions of this Agreement. The Sellers will deliver to Buyers, for Buyers' review and comment, each such Seller Return at least 45 days prior to its due date. Buyers will provide any written comments on each such Seller Return within 20 days of receipt of such Seller Return. Buyers and the Sellers will endeavor in good faith for a period of 10 days after receipt of such comments to resolve any disputes with respect thereto. In the event that Buyers and the Sellers are not able to resolve all such disputes in such 10-day period, then either Buyers or the Sellers may at any time thereafter submit any remaining disputed items by an independent accounting firm chosen by both Buyers and the Sellers ("Tax Arbiter"). If all disputes with respect to such Seller Return are not resolved prior to the due date (including extensions) for such Seller Return, then such Seller Return will be filed incorporating Buyers' comments, and such Seller Return will be amended or otherwise refiled to reflect the determination of the Tax Arbiter. The costs, fees and expenses of the Tax Arbiter will be shared equally by Buyers and Sellers.

(b) The Companies will prepare and file, or cause to be prepared and filed, all Tax Returns of the Companies, other than the Seller Returns, for a Pre-Closing Period (including a Straddle Period) which are to be filed by any Company after the Closing Date (the "Company Returns"). Each such Company Return will be prepared in a manner consistent with the Company's past practice except as otherwise required by Law or this Agreement, and in accordance with the provisions of this Agreement, and the Sellers will timely pay to Buyers not less than five days prior to the due date for such Taxes, in the case of any such Company Return for a Tax period ending before the Closing Date, all Taxes reflected as due and payable by any Company on all such Company Returns or, in the case of any such Company Return for a Straddle Period, the portion of Taxes reflected as due and payable on all such Company Returns that are allocable to the Pre-Closing Period. Buyers will deliver to the Sellers, for the Sellers' review and comment, any Company Return that could reasonably be expected to result in an indemnity obligation of the Sellers for Taxes at least 30 days (or such shorter period as reasonably required by circumstances or the nature of the applicable Tax Return) prior to its due date. The Sellers will provide any written comments on each such Company Return within 15 days (or such shorter period as reasonably required by circumstances or the nature of the applicable Tax Return) and Buyers will consider in good faith any such comments that are reasonable.

Capital Pumping

9.4.4 <u>Tax Contests</u>.

(a) After the Closing Date, Buyers and the Sellers, respectively, will inform the other party in writing of the commencement of any Tax Contest against or with respect to any Company relating in whole or in part to a Pre-Closing Period for which Buyers may be entitled to indemnity from the Sellers under this Agreement.

(b) Subject to <u>Section 9.4.4(c)</u>, after the Closing Date, Buyers will have the exclusive right to represent the interests of the Companies in any and all such Tax Contests; <u>provided</u>, <u>however</u>, that to the extent that any such Tax Contest could reasonably be expected to result in a Tax indemnification liability of the Sellers pursuant to this Agreement, (A) Buyers will keep the Sellers reasonably informed and consult in good faith with the Sellers with respect to any such Tax Contest (and the Sellers' Tax counsel or Tax adviser, at Sellers' expense, will be permitted to attend meetings with Taxing Authorities) and (B) Buyers will provide the Sellers with copies of all material correspondence, notices and other written materials received from any Taxing Authorities and will otherwise keep the Sellers informed on a reasonable basis of significant developments in such Tax Contest and of material communications involving representatives of the Taxing Authorities.

The Sellers will have the right, at the Sellers' sole cost and expense, to control and settle the portion of (c) any Tax Contest for which the Sellers may be required to indemnify a Buyers Indemnitee; provided, however, that (A) Buyers will have the right to participate in any such Tax Contest, (B) the Sellers will keep Buyers reasonably informed and consult in good faith with Buyers with respect to any such Tax Contest (and Buyers' Tax counsel or Tax adviser, at Buyers' expense, will be permitted to attend meetings with Taxing Authorities), (C) the Sellers will provide Buyers with copies of all material correspondence, notices and other written materials received from any Taxing Authorities and will otherwise keep Buyers informed on a reasonable basis of significant developments in such Tax Contest and of material communications involving representatives of the Taxing Authorities, and (D) the Sellers will not settle, compromise or resolve any such Tax Contest without Buyer's prior written consent, which consent will not be unreasonably withheld, conditioned or delayed; provided, however, that for any Tax Contest that is subject to the Partnership Tax Audit Rules, ASC and CP will make an election under Section 6226(a) of the Code (regarding the "push out" election) with respect to any Imputed Underpayment arising with respect to any such taxable year, or portion thereof, unless the Sellers either (I) fully comply with the procedures of Section 6225(c)(2) (A) of the Code (regarding the filing of amended returns or the alternative procedure to filing amended returns, respectively), or (II) pay to Buyers the Sellers' entire allocable share of any Imputed Underpayment, which in either event will be completed within 40 days after the notice of final partnership adjustments.

(d) In the event of any inconsistency between the provisions of this <u>Section 9.4.4</u> and any provision of <u>Section 12.5</u>, this <u>Section 9.4.4</u> will control.

9.4.5 <u>Partnership Tax Audit Rules</u>. No Seller will cause ASC or CP to elect to apply the Partnership Tax Audit Rules for any taxable period beginning prior to January 1, 2018.

9.4.6 <u>Assistance and Cooperation</u>. After the Closing Date, the Sellers, on the one hand, and Buyers, on the other hand, shall (and, if requested to do so, each shall cause their respective Affiliates to) (a) assist each other in preparing any Tax Returns that any other party hereto is responsible for preparing and filing; (b) cooperate fully in preparing for or defending against any Tax Contests with taxing authorities relating to any Tax Returns of the Companies for the Pre-Closing Period; (c) make available to the other parties hereto and to any Taxing Authority as reasonably requested all information, records and documents relating to Taxes of the Companies and (d) furnish the other parties hereto with timely written notice of, and copies of all correspondence received from any taxing authority in connection with, any Tax Contest relating to Taxes of the Companies for the Pre-Closing Period.

50

9.5 Notification of Certain Matters. Prior to the Closing, Buyers, on the one hand, and the Company Parties (in addition to their obligation under <u>Section 7.8</u>), on the other hand, shall reasonably promptly notify each other in writing of (a) the occurrence or non-occurrence of any event or the existence of any fact or condition that would cause any condition set forth in <u>Article 4</u> to not be satisfied, (b) any material Actions in connection with the transactions contemplated by this Agreement commenced or threatened against Buyers or any Company Party, or (c) any written notice or other written material communication from any Governmental Authority in connection with the transactions contemplated hereby; <u>provided</u>, <u>however</u>, that a party's good-faith failure to comply with this <u>Section 9.5</u> shall not provide any other party the right not to consummate the Closing or to effect the transactions contemplated by this Agreement.

9.6 Run-Off Period.

9.6.1 From and after the Closing, the Companies may receive and, for a period of 90 days following the Closing (the "<u>Run-off Period</u>") shall use commercially reasonable efforts to collect in the ordinary course of business, funds in payment or satisfaction of (a) Pre-Closing Accounts Receivable, (b) Pre-Closing Warranty Claims and (c) Excluded Insurance Proceeds. The Sellers shall be liable for and shall pay the Companies for all Pre-Closing Accounts Payable that are paid by or on behalf of the Companies.

9.6.2 Not later than five Business Days after, first, the 45th day of the Run-off Period, and second, the last day of the Run-off Period, the Companies shall provide the Sellers with a written statement identifying in reasonable detail (i) funds actually received by the Companies in payment of Pre-Closing Accounts Receivable, Pre-Closing Warranty Claims, and Excluded Insurance Proceeds for such period (the aggregate amount received is the "<u>Pre Closing Assets Amount</u>"), and (ii) amounts paid by or on behalf of the Companies in satisfaction of Pre-Closing Accounts Payable for such period (such aggregate amount is the "<u>Pre-Closing Liabilities Amount</u>").

9.6.3 If the Pre-Closing Assets Amount is *greater* than the Pre-Closing Liabilities Amount after either measurement period described in <u>Section 9.6.2</u>, the Companies shall pay to the Sellers not later than 10 Business Days after each such determination the amount of such excess to the accounts set forth on <u>Annex B</u>. The Sellers (a) shall agree among themselves as to the allocation and distribution of the such payment, and shall be solely responsible for distributing such amount to Persons entitled to a portion thereof, and (b) acknowledge that upon the Companies' payment of such amount to the account set forth on <u>Annex B</u>, Buyers and the Companies' obligations pursuant to this <u>Section 9.6.3</u> shall be fully satisfied.

9.6.4 If the Pre-Closing Assets Amount is *less than* the Pre-Closing Liabilities Amount after either measurement period described in <u>Section 9.6.2</u>, the Sellers shall pay to the Companies not later than 10 Business Days after each such determination period the amount of such shortfall to an account designated in writing by the Buyers and the Companies.

9.6.5 If, after the last day of the Run-off Period, the Companies receive funds in payment or satisfaction of Pre-Closing Accounts Receivable, Pre-Closing Warranty Claims, or Excluded Insurance Proceeds, the Companies shall promptly notify the Sellers in writing of such receipt and shall within five Business Days thereafter turnover such funds to the Sellers, net of any thenunpaid accounts payable of the Companies attributable to the period prior to Closing. Notwithstanding anything herein to the contrary this <u>Section 9.6</u> shall not limit or reduce the obligations of the Sellers hereunder with respect to the payment of insurance proceeds to Buyers in accordance with <u>Section 7.2</u>.

51

in full.

9.6.6 The obligations and covenants of the Sellers set forth in this <u>Section 9.6</u> shall survive indefinitely until satisfied

ARTICLE 10 RESTRICTIVE COVENANTS

10.1 Acknowledgment. Each Seller agrees and acknowledges that in order to assure Buyers that the Business and the Purchased Interests will retain their value, it is necessary that the Sellers undertake not to utilize their special knowledge of the Companies' Business and their relationship with clients or customers of the Companies to compete with Buyers.

10.2 Non-Competition. Each Seller hereby agrees that from and after the Closing Date and continuing for five years from the Closing Date (the "<u>Restricted Period</u>"), they shall not, directly or indirectly, as an employee, agent, consultant, director, equityholder, manager, partner or in any other individual or representative capacity, own, operate, manage, control, engage in, invest in, be employed by or participate in any manner in, act as a consultant or advisor to, render services for (alone or in association with any Person), or otherwise assist any Person that engages in or owns, invests in, operates, manages or controls any venture or enterprise that directly or indirectly engages or proposes to engage in the State of Texas (the "<u>Territory</u>") in the Business; <u>provided</u>, <u>however</u>, that nothing herein shall be construed to prevent any Seller from investing in the stock of any competing corporation listed on a national securities exchange or traded in the over the counter market so long as such party is not involved in the business of said corporation and such party does not own more than 5% of the stock of such corporation. With respect to the Territory, each Seller specifically acknowledge that the Business has heretofore been conducted throughout the State of Texas.

10.3 Non Solicitation. Without limiting the generality of the provisions of <u>Section 10.2</u>, each Seller hereby agrees that during the Restricted Period they will not, directly or indirectly, as employee, agent, consultant, director, equityholder, manager, partner or in any other capacity (a) without the prior written consent of Buyers, solicit concrete pumping business from any Person which is or was a client or customer of any Company Party during the three year period preceding the Closing Date, or from any successor in interest to any such Person, in any case for the purpose of securing business or contracts related to the Business, or (b) solicit, encourage, initiate or participate in discussions or negotiations with, or provide any information to, any present or future acquisition target, client, customer or supplier of the Business or any Person engaged in discussions with any Company Party related to such Person becoming a client of the Business, with respect to the termination or other alteration of his, her or its relationship or potential relationship with Buyers, any Company Party or the Business.

10.4 Interference with Relationships. During the Restricted Period, each Seller shall not, directly or indirectly, as employee, agent, consultant, director, equityholder, manager, partner or in any other capacity without the prior written consent of Buyers, employ, engage, recruit or solicit for employment or engagement, any Person who is (or was within six months of the Closing Date) employed or engaged by Buyers or any Company with respect to the Business, or a client or customer of the Business in any manner that adversely affects the Companies or the Business, or otherwise seek to influence or alter any such Person's relationship with any of the foregoing; provided, however, (a) general solicitations and posting shall not violate the foregoing restrictions, (b) the Sellers shall not be restricted from employing or engaging a Person who is terminated as contemplated by Section 4.2.10(e); and (c) subject to the mutual agreement of A. Keith Crawford and the Chief Executive Officer of BBCP, the Sellers shall not be restricted from employment with the Companies was terminated without cause by the applicable Company.

52

10.5 Confidential Information. From the date hereof and thereafter, each Seller shall keep secret and retain in strictest confidence, and shall not, without the prior written consent of Buyers, furnish, make available or disclose to any third party or use for the benefit of itself or any third party, any Confidential Information. As used in this <u>Section 10.5</u>, "<u>Confidential Information</u>" means any information relating to (a) this Agreement or the Transaction Documents or the transactions contemplated hereby or thereby or (b) the Business and the business or affairs of Buyers or their Affiliates, including, without limitation, information relating to financial statements, client or customer identities, potential clients or customers, employees, suppliers, servicing methods, equipment, programs, strategies and information which is in the public domain or becomes generally known in the public domain through no wrongful act on the part of any Seller. Each Seller acknowledges that the Confidential Information is vital, sensitive, confidential and proprietary to the Business.

10.6 Enforceability; Blue Pencil. Each Seller recognizes that the territorial, time and scope limitations set forth in this <u>Article 10</u> are reasonable and are properly required for the protection of Buyers' legitimate interest in client relationships, goodwill and trade secrets of the Business. In the event that any such territorial, time or scope limitation is deemed to be unreasonable by a court of competent jurisdiction the parties agree, and each party shall submit, to the reduction of any or all of said territorial, time or scope limitations to such an area, period or scope as said court shall deem reasonable under the circumstances. If such partial enforcement is not possible, the provision shall be deemed severed, and the remaining provisions of this Agreement shall remain in full force and effect.

10.7 Remedies. Each Seller acknowledge and agree that the covenants set forth in this <u>Article 10</u> are reasonable and necessary for the protection of Buyers' business interests, that irreparable injury will result to Buyers if a Seller breaches any of the terms of this <u>Article 10</u>, and that in the event of a Seller's actual or threatened breach of any of the provisions contained in this <u>Article 10</u>, Buyers will have no adequate remedy at law. Each Seller accordingly agrees that in the event of any actual or threatened breach by it of any of the provisions contained in this <u>Article 10</u>, Buyers shall be entitled to such injunctive and other equitable relief as may be deemed necessary or appropriate by a court of competent jurisdiction. Nothing contained herein shall be construed as prohibiting Buyers from pursuing any other remedies available to it for such breach or threatened breach, including the recovery of any damages which it is able to prove.

ARTICLE 11 TERMINATION/EFFECTIVENESS

11.1 Termination. This Agreement may be terminated and the transactions contemplated hereby abandoned at any time prior to the Closing:

11.1.1 by written consent of Buyers and the Company Parties;

53

11.1.2 by written notice to the Company Parties from Buyers if:

(a) there is any material breach of any representation, warranty, covenant or agreement on the part of any Company Party set forth in this Agreement, such that the conditions specified in <u>Sections 4.2.1, 4.2.2, 4.2.3</u>, or <u>4.2.4</u> would not be satisfied at the Closing (a "<u>Terminating Company Breach</u>"), except that, if such Terminating Company Breach is curable by the Company Parties through the exercise of their commercially reasonable efforts , then, for a period of up to 30 days after receipt by the Company Parties of notice from Buyers of such breach, but only as long as the Company Parties continue to use commercially reasonable efforts to cure such Terminating Company Breach (the "<u>Company Cure Period</u>"), such termination shall not be effective and the Termination Date shall be automatically extended until the end of the Company Cure Period, and such termination shall become effective only if the Terminated pursuant to this <u>Section 11.1.2(b)</u> if, as of such time, a Buyer is in material breach of any of its covenants or other obligations hereunder. For purposes of this <u>Section 11.1.2(a)</u>, references to the terms "material," "in all material respects," Material Adverse Effect or any similar term or phrase contained in any representation or warranty shall be disregarded for purposes of determining whether there has been any breach of such representation or warranty;

(b) the Closing has not occurred on or before 11:59 pm (ET) on the day that is 120 days after the date hereof ("<u>Termination Date</u>"), unless a Buyer's willful breach is the primary reason for the Closing not occurring on or before such date; <u>provided</u>, <u>however</u>, that if the date specified above falls on a day other than a Business Day, such date shall be 11:59 ET on the next Business Day; or

(c) the consummation of any of the transactions contemplated hereby is permanently enjoined or prohibited by the terms of a final, non-appealable order or judgment of a court of competent jurisdiction unless a Buyer's willful breach is the primary reason for such injunction or prohibition; or

11.1.3 by written notice to Buyers from the Company Parties if:

(a) (i) there is any material breach of any representation, warranty, covenant or agreement on the part of Buyers set forth in this Agreement, such that the conditions specified in <u>Section 4.3.1</u> or <u>4.3.2</u> would not be satisfied at the Closing (a "<u>Terminating Buyer Breach</u>"), except that, if any such Terminating Buyer Breach is curable by Buyer through the exercise of its commercially reasonable efforts, then, for a period of up to 30 days after receipt by Buyers of notice from the Company Parties of such breach, but only as long as Buyers continue to exercise such commercially reasonable efforts to cure such Terminating Buyer Breach (the "<u>Buyer Cure Period</u>"), such termination shall not be effective and the Termination Date shall automatically be extended until the end of Buyer Cure Period, and such termination shall become effective only if the Terminating Buyer Breach is not cured within Buyer Cure Period; <u>provided</u>, <u>however</u>, this Agreement may not be terminated pursuant to this <u>Section 11.1.3(a)</u> if, as of such time, any Company Party is in material breach of any of its covenants or other obligations hereunder. For purposes of this <u>Section 11.1.3(a)</u>, references to the terms "material," "in all material respects," Material Adverse Effect or any similar term or phrase contained in any representation or warranty shall be disregarded for purposes of determining whether there has been any breach of such representation or warranty;

54

(b) the Closing has not occurred on or before 11:59 pm (ET) on the day that is the later of (i) 90 days after the date hereof, or (ii) 105 days after the date hereof if a Governmental Authority informs CPHI that it will review or request additional information with respect to any filings or submissions made to such Governmental Authority by or on behalf of CPHI or the Buyers in connection with this Agreement, the Proxy Statement, Equity Financing or the Warrant Exchange, unless a Company Party's willful breach is the primary reason for the Closing not occurring on or before such date; <u>provided</u>, <u>however</u>, that if the date specified in <u>clauses in (i)</u> or (<u>ii)</u> above falls on a day other than a Business Day, the date in such clause shall be 11:59 ET on the next Business Day; or

(c) the consummation of any of the transactions contemplated hereby is permanently enjoined or prohibited by the terms of a final, non-appealable order or judgment of a court of competent jurisdiction unless a Company Party's willful breach is the primary reason for such injunction or prohibition.

11.2 Effect of Termination. Except as otherwise set forth in this <u>Section 11.2</u> or <u>Section 11.3</u>, in the event of the termination of this Agreement pursuant to <u>Section 11.1</u>, this Agreement shall forthwith become void and have no effect, without any liability on the part of any party hereto or its respective Affiliates, officers, directors, employees, partners, managers, members or stockholders, subject to the immediately succeeding sentence and other than liability of a party for any willful and material breach of this Agreement occurring prior to such termination. The provisions of <u>Article 1</u> (Defined Terms), this <u>Section 11.2</u>, <u>Section 11.3</u> and <u>Article 13</u> (Miscellaneous).

11.3 Liquidated Damages. If this Agreement is terminated by the Company Parties under <u>Section 11.1.3(b)</u>, Buyers shall pay to the Sellers Five Million Dollars (\$5,000,000) as liquidated damages (the "Liquidated Damages"). The parties intend that the Liquidated Damages constitute compensation for Buyers' breach of this Agreement and not a penalty. The parties acknowledge and agree that the Sellers' harm caused by Buyers would be impossible or very difficult to accurately estimate, and that the Liquidated Damages are a reasonable estimate of the anticipated or actual harm that might arise from the Sellers' harm caused by Buyers. If the Company Parties terminate this Agreement pursuant to <u>Section 11.1.3(b)</u>, notwithstanding anything contained in <u>Article 13</u> or elsewhere to the contrary, Buyers' payment of the Liquidated Damages to the Company Parties is the exclusive remedy available to the Company Parties for Buyers' breach hereunder.

ARTICLE 12 SURVIVAL; INDEMNIFICATION

12.1 Survival.

12.1.1 Each of the representations and warranties of the Company Parties set forth in <u>Article 5</u> and the representations and warranties of Buyers set forth in <u>Article 6</u> shall terminate and be of no further force or effect at 6:00 p.m. (ET) on the first anniversary of the Closing Date (the "<u>Cut-Off Date</u>"); provided, however, that the representations and warranties of the Company Parties set forth in:

(a) <u>Section 5.1</u> (Organization), <u>Section 5.2</u> (Due Authorization), <u>Section 5.4</u> (Compliance With Laws), <u>Section 5.6</u> (Capitalization) and <u>Section 5.29</u> (Brokerage) shall survive indefinitely; and

55

(b) <u>Section 5.13</u> (Employee Benefit Plans), <u>Section 5.15</u> (Taxes) and <u>Section 5.20</u> (Environmental) shall survive until the statute of limitations with respect to the matters addressed therein have expired (including all waivers or extensions thereof); and

all such representations and warranties described in <u>clauses (a)</u> and <u>(b)</u> above are hereinafter referred to collectively as the "<u>Fundamental Representations</u>"; and <u>provided</u>, <u>further</u>, that any claims for indemnification made or asserted by a party within the applicable time period prescribed above shall survive such expiration until such claim is finally resolved and all obligations with respect thereto are fully satisfied. The representations and warranties of the Company Parties set forth in <u>Article 5</u> other than the Fundamental Representations are the "<u>Limited Representations</u>".

12.1.2 Each of the covenants and other agreements of the parties set forth in this Agreement required to be performed or complied with prior to the Closing shall survive indefinitely. The covenants and agreements of the parties set forth in this Agreement which contemplate performance after the Closing or that otherwise expressly by their terms survive the Closing, shall survive in accordance with their terms. Such covenants and agreements prior to their termination are referred to herein as the "Indemnified Covenants."

12.2 Indemnification by the Sellers. Subject to the limitations set forth herein, from and after the Closing Date, the Sellers shall jointly and severally defend, protect, indemnify and hold Buyers and their Affiliates and their respective officers, directors, employees, equity holders, partners, controlling Persons, agents, fiduciaries and members (each, a "Buyer Indemnitee") harmless from and against any and all Losses that may be incurred or suffered by any Buyer Indemnitee arising from (a) any breach or violation of a Limited Representation, (b) any breach or violation of a Fundamental Representation, (c) any noncompliance with or breach of any Indemnified Covenant made by a Seller or to be performed by a Seller, or made by a Company to be performed prior to Closing, (d) any and all Excluded Liabilities, (e) any Indemnified Taxes and (f) any of the items set forth on Schedule 12.2(f).

12.3 Indemnification by Buyers. Subject to the limitations set forth herein, from and after the Closing Date, Buyers shall indemnify and hold harmless the Sellers and their respective officers, directors, employees, equity holders, partners, controlling Persons, agents, fiduciaries and members (each, an "Seller Indemnitee" and, together with the Buyer Indemnitees, collectively "Indemnitees" and each an "Indemnitee"), from and against any and all Losses arising from any (a) breach or violation of any representations and warranties of Buyers set forth herein, (b) noncompliance with or breach of any Indemnified Covenant made by Buyers or to be performed by Buyers or (c) the operation of the Business on or after the Closing Date.

12.4 Limitations on Indemnification.

12.4.1 <u>Cap</u>. The maximum aggregate amount of indemnifiable Losses that may be recovered by the Buyer Indemnitees in respect of claims for indemnification under <u>Section 12.2(a)</u> and <u>Section 12.2(b)</u> solely with respect to Losses arising in connection with claims for indemnification based on breaches or violations of <u>Section 5.20</u> (Environmental), other than Losses arising in connection with claims concerning the Salado Property, which shall not be subject to or count toward the Cap, shall be the Escrow Amount (the "<u>Cap</u>"); <u>provided</u>, <u>however</u>, that this <u>Section 12.4.1</u> shall not apply to Losses arising in connection with claims for indemnification activity or willful misconduct of a Company Party. Except as set forth above, there shall be no maximum aggregate amount of indemnifiable Losses that may be recovered by the Buyer Indemnitees in respect of any other claim by the Buyer Indemnitees for indemnification.

12.4.2 <u>Escrow Fund; R&W Insurance Policy</u>. All Losses arising in connection with claims for indemnification under <u>Section 12.2(a)</u> shall be satisfied by the Sellers solely and exclusively from the then remaining amounts in the Escrow Fund and pursuant to the terms of the R&W Insurance Policy; <u>provided</u>, <u>however</u>, that this <u>Section 12.4.2</u> shall not apply to Losses arising in connection with claims for indemnification based on the fraud, criminal activity or willful misconduct of a Company Party. Losses arising in connection with claims for indemnification under <u>Section 12.2(b)</u> shall first be satisfied from the then remaining amounts in the Escrow Fund, if any.

12.4.3 <u>Deductible</u>.

(a) No indemnification payments shall be made to the Buyer Indemnitees with respect to claims for indemnification under Section 12.2(a) unless the aggregate amount of Losses under all claims for indemnification under Section 12.2(a) exceeds the Deductible, whereupon, subject to Section 12.4.1, the Buyer Indemnitees shall be entitled to recover the entire amount of such Losses and not only amounts in excess of the Deductible. Claims for indemnification under Section 12.2(b), 12.2(c), 12.2(d), 12.2(e) or 12.2(f) shall not be subject to the Deductible.

(b) Notwithstanding anything contained herein to the contrary, no indemnification payments shall be made to the Buyer Indemnitees with respect to claims for indemnification under <u>Section 12.2(b)</u> solely with respect to breaches of <u>Section 5.4</u> (Compliance with Laws) ("<u>Compliance with Laws Claims</u>") unless the aggregate amount of Losses under all Compliance with Laws Claims exceeds \$200,000 ("<u>Mini Basket</u>"), whereupon the Buyer Indemnitees shall be entitled to recover the entire amount of such Losses and not only amounts in excess of the Mini Basket.

(c) No indemnification payments shall be made to the Seller Indemnitees with respect to claims for indemnification under Section 12.3(a) unless the aggregate amount of Losses under all claims for indemnification under Section 12.3(a) exceeds the Deductible, whereupon the Seller Indemnitees shall be entitled to recover the entire amount of such Losses and not only amounts in excess of the Deductible. Claims for indemnification under Section 12.3(b) and Section 12.3(c) shall not be subject to the Deductible.

12.4.4 No party shall be obligated to indemnify any other Person with respect to any representation, warranty, covenant or condition specifically waived in writing by the other party on or prior to the Closing.

12.4.5 Any Loss under this Agreement shall be determined without duplication of recovery by reason of the state of facts giving rise to such Loss constituting a breach of more than one representation, warranty, covenant or agreement.

12.5 Indemnification Claim Process.

12.5.1 If an Indemnitee intends to seek indemnification pursuant to this <u>Article 12</u>, the Indemnitee shall promptly notify, with respect to a Buyer Indemnitee, the Sellers, and with respect to a Seller Indemnitee, Buyer, in writing of such claim, describing such claim in reasonable detail and the amount or estimated amount of Losses (a "<u>Claims Notice</u>"); provided, however, that any failure or delay on the part of the Indemnitee to deliver such a Claims Notice shall not relieve the applicable Indemnitors of their obligations under this <u>Article 12</u>, except if, and only to the extent, that such failure or delay prejudices such Indemnitors.

57

12.5.2 With respect to any action, lawsuit, proceeding, investigation, demand or other claim against an Indemnitee by a third party (a "Third Party Claim"), the Indemnitors shall have 10 Business Days after the date the Indemnitor receives the Claims Notice to notify the Indemnitee in writing that the Indemnitor desires to assume control of the defense or prosecution of the Third Party Claim and any litigation resulting therefrom (a "Defense Assumption Notice"). By delivering a Defense Assumption Notice to the Indemnitee, the Indemnitor is confirming that (a) it is obligated to indemnify, defend and hold harmless the Indemnitee under terms of their indemnification obligations hereunder (and subject to the limitations on the indemnification obligations set forth herein) and (b) that it will undertake, conduct and control, through counsel of its own choosing (which counsel shall be reasonably satisfactory to the Indemnitee) and at its own expense, the settlement or defense thereof and in which event the Indemnitee shall cooperate with it in connection therewith. If the Indemnitor delivers a Defense Assumption Notice to the Indemnitee: (i) the Indemnitee may retain separate co-counsel at the Indemnitee's sole cost and expense and participate in the defense of such Third Party Claim, but the Indemnitor shall control the investigation, defense and settlement thereof; (ii) the Indemnitee shall not file any papers or consent to the entry of any judgment or enter into any settlement with respect to such Third Party Claim without the prior written consent of the Indemnitor; and (iii) the Indemnitor shall not consent to the entry of any judgment or enter into any settlement with respect to such Third Party Claim without the prior written consent of the Indemnitee unless the judgment or settlement provides solely for the payment of money and the applicable Indemnitees receive an unconditional release with respect to such Third Party Claim. The parties shall act in good faith in responding to, defending against, settling or otherwise dealing with Third Party Claims, and cooperate in any such defense and give each other reasonable access to all information relevant thereto and shall keep each other reasonably informed with respect to the status thereof. Whether or not the Indemnitor has assumed the defense of such Third Party Claim, the Indemnitor will not be obligated to indemnify the Indemnitee hereunder with respect to any settlement entered into or any judgment consented to without the Indemnitor's prior written consent.

12.5.3 If the Indemnitor does not assume the defense of such Third Party Claim within 10 Business Days of receipt of the Claims Notice, the Indemnitee will be entitled to assume such defense, at its sole cost and expense (it being understood that the Indemnitor shall remain liable to the extent such costs and expenses constitute a Loss for which the Indemnitee is entitled to indemnification pursuant to <u>Article 12</u>), upon delivery of notice to such effect to the Indemnitor; <u>provided</u>, <u>however</u>, that the Indemnitor shall have the right to participate in the defense of the Third Party Claim at its sole cost and expense.

12.5.4 Notwithstanding the foregoing, the Indemnitor shall not have the right to assume the defense of such Third Party Claim, if (a) the claim seeks an injunction or other equitable relief, (b) the Indemnitee shall have been advised by counsel that there are one or more legal or equitable defenses available to them which are different from or in addition to those available to the Indemnitor, and, in the reasonable opinion of the Indemnitee, counsel for the Indemnitor could not adequately represent the interests of the Indemnitee because such interests could be in conflict with those of the Indemnitor, (c) the Indemnitor fails to provide the Indemnitee with reasonable evidence that the Indemnitor has the financial wherewithal to pay for such defense, (d) the Indemnitor shall not have assumed the defense of such Third Party Claim within the required time period, (e) such Third Party Claim is brought by the Indemnitee's customers, suppliers, lenders, equityholders, employees or other business relationships, (f) the Third Party Claim involves a criminal matter or (g) with respect to claims under <u>Section 12.2(a)</u> or <u>Section 12.2(b)</u>, if the reasonable potential Losses involved in such Third Party Claim exceed the amount then remaining in the Escrow Fund.

58

12.5.5 To the extent there is any inconsistency between this Section 12.5 and Section 9.3 or 9.4 as it relates to a Tax matter, the provisions of Sections 9.3 and 9.4 shall govern.

12.6 Indemnification Procedures for Non-Third Party Claims. The Indemnitee will deliver a Claims Notice to the Indemnitor promptly upon its discovery of any matter for which the Indemnitor may be liable to the Indemnitee hereunder that does not involve a Third Party Claim, which Claims Notice shall state that the Indemnitee has paid or properly accrued Losses or anticipates that it will incur liability for Losses for which such Indemnitee is entitled to indemnification pursuant to this Agreement. The Indemnitee shall reasonably cooperate and assist the Indemnitor in determining the validity of any claim for indemnity by the Indemnitee and in otherwise resolving such matters. Such assistance and cooperation shall include providing reasonable access to and copies of information, records and documents relating to such matters, furnishing employees to assist in the investigation, defense and resolution of such matters and providing legal and business assistance with respect to such matters.

12.7 Exclusive Remedy.

12.7.1 The indemnification obligations of the Sellers with respect to claims for indemnification under <u>Section</u> <u>12.2(a)</u> and <u>12.2(b)</u> shall first be satisfied through recourse to the amounts then remaining in the Escrow Fund, if any. The Sellers shall be jointly and severally liable to the Buyer Indemnitees for (i) to the extent there are no funds then remaining in the Escrow Fund, Losses arising in connection with claims for indemnification under <u>Section 12.2(b)</u> and (ii) Losses arising in connection with claims for indemnification under <u>Sections 12.2(c)</u>, <u>12.2(d)</u>, <u>12.2(e)</u> and <u>12.2(f)</u>.

12.7.2 The Escrow Fund and Buyers' rights under the R&W Insurance Policy shall be the sole and exclusive remedy of the Buyer Indemnitees in respect of Losses arising in connection with claims for indemnification under <u>Section 12.2</u>; <u>provided</u>, <u>however</u>, that this <u>Section 12.7.2</u> shall not apply to (i) Losses arising in connection with claims for indemnification under <u>Section 12.2(b)</u>, <u>12.2(c)</u>, <u>12.2(d)</u>, <u>12.2(e)</u> or <u>12.2(f)</u>, (ii) Losses arising in connection with claims for indemnification based on the fraud, criminal activity or willful misconduct of a Company Party in connection with the representations and warranties made by such Company Party in this Agreement or any certificate delivered pursuant hereto, as qualified by the Schedules, or (iii) Losses arising in connection with claims pursuant to any Transaction Document other than this Agreement.

The Escrow Amount and any interest accrued thereon (collectively, the "Escrow Fund") shall be held and 12.7.3 disbursed by the Escrow Agent in accordance with the Escrow Agreement and this Agreement. The Escrow Fund shall be held until the earlier of (a) the exhaustion of the Escrow Fund and (b) the Cut-Off Date. Any portion of the Escrow Fund remaining on the day following the Cut-Off Date shall be released to the Sellers, less the sum of the aggregate amount, if any, claimed by the Buyer Indemnitees under Section 12.2.(a) or 12(b) pursuant to claims (such claims, the "Outstanding Claims") properly made against the Escrow Fund in accordance with this Article 12 and not fully resolved prior to the Cut-Off Date, which amount shall be retained by the Escrow Agent (such amount of the retained Escrow Fund, as it may be further reduced after the Cut-Off Date by distributions to the Companies as set forth below and by recoveries by the Buyer Indemnitees pursuant to Section 12.2.(a) and 12.2(b) and the Escrow Agreement, the "Retained Escrow Amount"). In the event and to the extent that, after the Cut-Off Date, any Outstanding Claim is resolved by the Sellers and the Buyer Indemnitee or finally determined by a court of competent jurisdiction (i) the Sellers and Buyers shall jointly instruct the Escrow Agent to promptly release from the Escrow Fund to the Sellers, in accordance with payment instructions provided by the Sellers, an aggregate amount of the Retained Escrow Amount equal to any amount of the Outstanding Claim resolved against such Buyer Indemnitee, and (ii) the Sellers and Buyers shall jointly instruct the Escrow Agent to promptly release from the Escrow Fund to Buyer, in accordance with payment instructions provided by Buyers, an aggregate amount of the Retained Escrow Amount equal to any amount of the Outstanding Claim resolved in favor of such Buyer Indemnitee.

59

12.7.4 <u>Calculation of Losses; Limitations</u>. The amount of any Loss for which indemnification is provided under this <u>Article 12</u> shall be net of any amounts actually recovered by any Indemnitee under insurance policies (including any representation and warranty insurance policy maintained by or for the benefit of the Sellers prior to the Closing (other than the R&W Insurance Policy)), and from third parties or otherwise with respect to such Loss (net of direct collection expenses and any increase in premiums under insurance policies to the extent attributable to the recovery of such proceeds). With respect to any matter giving rise to a claim for indemnification in this <u>Article 12</u>, should an Indemnitee receive the amounts described in the preceding sentence after such time as the Indemnitor has made payment to the Indemnitee with respect to such matter, the Indemnitee shall promptly pay any such amounts to the Indemnitor. In the event any Indemnitee determines it is entitled to insurance proceeds (including proceeds under any representation and warranty insurance policy maintained by or for the benefit of the Sellers prior to the Closing (other than the R&W Insurance Policy)) or any other third-party recoveries in respect of any Losses (or any of the circumstances giving rise thereto) for which such Indemnitee is entitled to indemnification pursuant to this <u>Article 12</u>, such Indemnitee shall use commercially reasonable efforts to obtain, receive or realize such proceeds, payments or recoveries.

12.8 Tax Treatment of Indemnity Payments. Unless otherwise required by applicable Law, any indemnity payment made under this Agreement shall be treated by all parties as an adjustment to the Purchase Price for all federal, state, local and foreign Tax purposes, and the parties shall file their Tax Returns accordingly.

12.9 Determination of Breaches and Losses. For purposes of this <u>Article 12</u>, references to the terms "material," "in all material respects," Material Adverse Effect or any similar term or phrase contained in any representation or warranty shall be disregarded (a) for purposes of determining whether there has been any breach of any representation or warranty in this Agreement (except for the representations and warranties in <u>Sections 5.1</u> (Organization), <u>5.12</u> (Contracts), <u>5.17</u> (Licenses and Permits), <u>5.22</u> (Conduct of the Business) and <u>5.27</u> (Information Supplied)), and (b) for purposes of determining the amount of Losses resulting therefrom.

12.10 Effect of Investigation. A claim for indemnity may be made by the Indemnitees under <u>Section 12.2</u> or <u>Section 12.3</u> despite the fact that such Indemnitee had actual knowledge prior to the execution and delivery of this Agreement, and the consummation of the transactions contemplated hereby, of the breach of, or of any facts or circumstances constituting or resulting in the breach of, such representation or warranty. In furtherance of the foregoing, each party hereto shall be entitled to rely upon, and shall be deemed to have relied upon, all representations, warranties and covenants of each other party set forth in this Agreement which have been or are made in favor of such party, and the rights of Indemnitees under this <u>Article 12</u> shall not be affected, notwithstanding (a) any investigation or examination conducted with respect to, or any knowledge acquired (or capable of being acquired) about the accuracy or inaccuracy of or compliance with, any representation, warranty, covenant, agreement, undertaking or obligation made by or on behalf of the parties hereto, (b) the waiver of any condition based on the accuracy of any representation or warranty, or on the performance of or compliance with any covenant, agreement, undertaking or obligation, or (c) the consummation of the Closing hereunder.

Capital Pumping

12.11 Resolution of Disputes Under Article 12. Any controversy or claim arising out of or relating to this <u>Article 12</u> or the breach, termination, enforcement, interpretation or validity thereof, shall be submitted to and determined by arbitration before one arbitrator mutually agreed upon by the parties. If the parties cannot reach an agreement on the arbitrator within 30 days after each party has submitted names of potential arbitrators, then the arbitrator shall be chosen by the Judicial Arbitration and Mediation Service (JAMS). If selected by JAMS, the arbitrator shall either be a retired Texas state court judge or an attorney with at least 15-years' experience in transactions of the type provided for in this Agreement. The arbitration shall be administered by JAMS pursuant to its Comprehensive Arbitration Rules and Procedures and shall take place in Dallas, Texas. Judgment on any award may be entered in any court having jurisdiction. This clause shall not preclude the parties from seeking provisional remedies in aid of arbitration from a court of appropriate jurisdiction. This arbitration provision shall not preclude any Indemnitee from joining an Indemnitor or pursuing a third party complaint against an Indemnitor in connection with a Third Party Claim in the event the Indemnitor does not assume the defense of such Third Party Claim in accordance with this <u>Article 12</u>, but the failure to so join or pursue a third party complaint shall not waive, release, or otherwise adversely affect any right to commence arbitration under this section.

ARTICLE 13 MISCELLANEOUS

13.1 Waiver. Any party hereto may waive any of the terms or conditions of this Agreement by an agreement in writing executed by the party providing such waiver. No waiver by any of the parties hereto of any default, misrepresentation or breach of representation, warranty, covenant or other agreement hereunder, whether intentional or not, shall be deemed to extend to any prior or subsequent default, misrepresentation or breach or affect in any way any rights arising by virtue of any prior or subsequent such occurrence. No waiver by any of the parties of any of the provisions hereof shall be effective unless explicitly set forth in writing and executed by the party sought to be charged with such waiver. Notwithstanding anything to the contrary contained herein, this <u>Section 13.1</u>, <u>Section 13.5</u>, <u>Section 13.12</u> and <u>Section 13.17</u> (and any definition set forth in, or other provision of, this Agreement to the extent that a waiver of such definition or other provision would amend or modify the substance of this <u>Section 13.1</u>, <u>Section 13.5</u>, <u>Section 13.5</u>, <u>Section 13.17</u>) may not be waived in a manner adverse to any Debt Lender without the prior written consent of such Debt Lender (and any such waiver without such prior written consent shall be null and void).

13.2 Notices. All notices, demands, requests, consents, approvals, declarations and other communications required or permitted to be given under this Agreement or any other Transaction Document shall be in writing and shall be sent by hand delivery, overnight courier, U.S. mail, email or electronic delivery as follows:

c/o A. Keith Crawford 3200 Steck Ave., Suite 220

If to any Seller, or any Company

Party on or prior to the Closing Date:

With a copy to:

Austin, TX 78757 Email: [to be provided] McGinnis Lochridge LLP 600 Congress Avenue, Suite 2100 Austin, TX 78701

Email: emchorse@mcginnislaw.com

61

Attn: Ed McHorse

 If to Buyers, CPHI or any
 c/o Brundage-Bone Concrete Pumping, Inc.

 Company after the Closing Date:
 c/o Brundage-Bone Concrete Pumping, Inc.

 6461 Downing St.
 Denver, CO 80229

 Attn: Bruce Young, CEO
 Email: Bruce Young@BrundageBone.com

 With a copy to:
 Ballard Spahr LLP

 1225 17th Street, Suite 2300
 Denver, CO 80202

 Attn: Michele Rowland
 Email: rowlandm@ballardspahr.com

or to such other address or email address as a party may designate from time to time in writing to the other parties. Unless otherwise provided herein, notices shall be deemed to be validly served, given or delivered (a) on the date of delivery if hand delivered, (b) one Business Day after deposit with a reputable overnight courier with all charges prepaid, (c) three Business Days after deposit in the mail when sent by U.S. mail and (d) when sent by email or electronic delivery, immediately upon confirmation that such email or electronic delivery has been transmitted.

13.3 Assignment. This Agreement shall not be assignable by any party hereto without the prior written consent of the other parties hereto; <u>provided</u>, <u>however</u>, that Buyers and CPHI may at any time, and the Companies may after Closing, without the prior written consent of any other party, assign their rights and interest in this Agreement to any Affiliate of Buyers or CPHI, to the provider of the R&W Insurance Policy and/or as security to their secured lenders.

13.4 Successors and Assigns. This Agreement and the other Transaction Documents shall inure to the benefit of and be binding upon each party's successors and assigns. Each party who is an individual is entering into this Agreement on behalf of himself or herself and his or her heirs, estate, executors, administrators, personal or legal representatives, trustees, successors in interest, assigns, distributes, devises and legatees, bankruptcy trustees, liquidators and Affiliates, and this Agreement shall be binding on them. All references to such individual party herein shall mean such individual and his or her heirs, estate, executors, administrators, devises and legatees, bankruptcy trustees, liquidators and affiliates.

13.5 Rights of Third Parties. Nothing expressed or implied in this Agreement is intended or shall be construed to confer upon or give any Person, other than the parties hereto, any right or remedies under or by reason of this Agreement; <u>provided</u>, <u>however</u>, that, notwithstanding the foregoing in the event the Closing occurs, the Indemnitees (and their successors, heirs and representatives) are intended third-party beneficiaries of, and may enforce, <u>Article 12</u>. The Debt Lenders shall be intended third-party beneficiaries of <u>Section 13.1</u>, this <u>Section 13.5</u>, <u>Section 13.12</u> and <u>Section 13.17</u>.

13.6 Joint and Several Obligations. All obligations, liabilities, duties, covenants, agreements and responsibilities of a Company Party or of the Company Parties set forth in this Agreement (including those of as an indemnifying party pursuant to <u>Article</u> <u>12</u>) or any other Transaction Document shall be joint and several (and not several) obligations, liabilities, duties, covenants, agreements and responsibilities of all the Company Parties.

62
13.7 Expenses. Except as otherwise specifically set forth herein, each party shall bear its own expenses incurred in connection with this Agreement and the transactions contemplated hereby whether or not such transactions shall be consummated, including all fees of its legal counsel, financial advisers and accountants. Except as otherwise provided herein, the prevailing party in any legal proceeding brought pursuant to, or to enforce, a claim for indemnification or otherwise pursuant to this Agreement shall recover its costs and reasonable attorneys' fees from the non-prevailing party.

13.8 Governing Law; Jurisdiction; Service of Process.

13.8.1 <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with, the internal laws of the State of Texas, without regard to conflict of laws principles.

13.8.2 <u>Jurisdiction</u>. Each party hereby irrevocably submits to the exclusive jurisdiction of the United States District Court or the Texas State Court located in Austin, Texas, in respect of any claim relating to the interpretation and enforcement of the provisions of this Agreement and each other Transaction Document, or otherwise in respect of the transactions contemplated hereby and thereby, and hereby waives, and agrees not to assert, as a defense in any action, suit or proceeding in which any such claim is made that it is not subject thereto or that such action, suit or proceeding may not be brought or is not maintainable in such courts or that the venue thereof may not be appropriate or that this Agreement or any other Transaction Document may not be enforced in or by such courts.

13.8.3 <u>Service of Process</u>. Each of the parties irrevocably consents to the service of process outside the territorial jurisdiction of the courts referred to in <u>Section 13.8.3</u> in any such action or proceeding by giving copies thereof by hand delivery of air courier to his, her or its address as specified in or pursuant to <u>Section 13.8</u>. However, the foregoing shall not limit the right of a party to effect service of process on the other party by any other legally available method.

1 3 . 8 . 4 <u>WAIVER OF JURY TRIAL</u>. EACH PARTY HEREBY WAIVES (TO THE FULLEST EXTENT PERMITTED BY LAW) ITS RESPECTIVE RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF THIS AGREEMENT. THE PARTIES ACKNOWLEDGE THAT THIS WAIVER IS A MATERIAL INDUCEMENT TO ENTER INTO THIS AGREEMENT. THIS WAIVER IS IRREVOCABLE, MEANING THAT IT MAY NOT BE MODIFIED EITHER ORALLY OR IN WRITING, AND THIS WAIVER SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS TO THIS AGREEMENT OR TO ANY OTHER DOCUMENTS OR AGREEMENTS RELATING TO THE TRANSACTIONS CONTEMPLATED HEREBY. IN THE EVENT OF LITIGATION, THIS AGREEMENT MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.

13.9 **Captions; Counterparts**. The captions in this Agreement are for convenience only and shall not be considered a part of or affect the construction or interpretation of any provision of this Agreement. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

13.10 Schedules, Annexes and Exhibits. The Schedule, Annexes and Exhibits referenced herein are a part of this Agreement as if fully set forth herein. All references herein to Schedules, Annexes and Exhibits shall be deemed references to such parts of this Agreement, unless the context shall otherwise require.

63

Capital Pumping

13.11 Entire Agreement . This Agreement (together with the Schedules, Annexes and Exhibits) and the other Transaction Documents constitute the entire agreement among the parties relating to the transactions contemplated hereby and supersede any other agreements, whether written or oral, that may have been made or entered into by or among any of the parties relating to the transactions contemplated hereby and thereby. No representations, warranties, covenants, understandings or agreements, oral or otherwise, relating to the transactions contemplated by this Agreement exist between the parties, except as expressly set forth in this Agreement and the other Transaction Documents.

13.12 Amendments. This Agreement may be amended or modified in whole or in part, only by a duly authorized agreement in writing executed in the same manner as this Agreement and which makes reference to this Agreement. The approval of this Agreement or the Equity Financing or the Debt Financing by the stockholders of CPHI shall not restrict the ability of the Board of Directors of CPHI to cause CPHI and/or Buyers to terminate this Agreement in accordance with <u>Section 11.1.2</u> or to cause CPHI or Buyers to enter into an amendment to this Agreement. Notwithstanding anything to the contrary contained herein, <u>Section 13.1</u>, <u>Section 13.5</u>, this <u>Section 13.12</u> and <u>Section 13.17</u> (and any definition set forth in, or other provision of, this Agreement to the extent that an amendment or modification of such definition or other provision would amend or modify the substance of <u>Section 13.1</u>, <u>Section 13.5</u>, this <u>Section 13.12</u> and <u>Section 13.17</u>) may not be amended or modified in a manner that is adverse to any Debt Lender without the prior written consent of such Debt Lender (and any such amendment or modification without such prior written consent shall be null and void).

13.13 Publicity. The parties agree that, from the date hereof through the Closing Date, no public release or announcement concerning the transactions contemplated hereby shall be issued or made by or on behalf of any party without the prior written consent of the other parties and otherwise as a party may reasonably determine is necessary to comply with applicable Law (including under the Securities Act of 1933, as amended and the Securities Exchange Act of 1934, as amended or as required by any national securities exchange) or the requirements of any agreement to which CPHI or either any Buyer or any of their Affiliates is a party. In furtherance of the foregoing, the Company Parties shall cooperate to prepare a joint press release to be issued on or promptly (and in any event within two Business Days) after the date of this Agreement and a joint press release to be issued on the Closing Date.

13.14 Severability. If any provision of this Agreement is held invalid or unenforceable by any court of competent jurisdiction, the other provisions of this Agreement shall remain in full force and effect. The parties further agree that if any provision contained herein is, to any extent, held invalid or unenforceable in any respect under the Laws governing this Agreement, they shall take any actions necessary to render the remaining provisions of this Agreement valid and enforceable to the fullest extent permitted by Law and, to the extent necessary, shall amend or otherwise modify this Agreement to replace any provision contained herein that is held invalid or unenforceable provision giving effect to the intent of the parties.

13.15 Representation by Counsel. In connection with this Agreement, the parties have relied solely upon their own judgment, belief and knowledge and have sought the advice and recommendations of their own independent selected counsel concerning the nature, extent and duration of their rights and claims relating to this Agreement, the rights affected by this Agreement, the form and content of this Agreement, and the advisability of entering into and executing the Agreement. No party has been unduly influenced to any extent whatsoever by any other party. The parties have read this Agreement carefully. The contents hereof are known and understood by the parties, and this Agreement is freely and voluntarily signed.

64

Capital Pumping

13.16 Conflict of Terms. In the event this Agreement conflicts with any binding provision of the letter of intent dated January 15, 2019, between CP, CPHI and A. Keith Crawford, as amended, this Agreement will govern.

13.17 Certain Matters Regarding Debt Lenders.

Notwithstanding anything herein to the contrary, (i) each Seller (on behalf of itself and its respective 13.17.1 officers, directors, employees, members, managers, partners and controlling Persons), each Company (on behalf of itself and its officers, directors, employees, members, managers, partners and controlling Persons) and each of the other parties hereto agrees that any claim, controversy or dispute of any kind or nature (whether in contract or in tort, in Law, in equity or otherwise) involving or against a Debt Lender that is in any way related to this Agreement, the Incremental Amendment, the Debt Financing contemplated thereby or any of the transactions contemplated hereby or thereby will be governed by and construed in accordance with the internal Laws of the State of New York without regard to the conflicts of Law principles that would require the application of any other Law, and (ii) each Seller (on behalf of itself and its respective officers, directors, employees, members, managers, partners and controlling Persons), each Company (on behalf of itself and its officers, directors, employees, members, managers, partners and controlling Persons) and each of the other parties hereto (A) agrees that it will not bring or support any Person, or permit any of its Affiliates to bring or support any Person, in any action, suit, proceeding, cause of action, claim, cross-claim or third-party claim of any kind or description, whether in Law or in equity. whether in contract or in tort or otherwise, against the Debt Lenders in any way relating to this Agreement, the Incremental Amendment, the Debt Financing contemplated thereby or any of the transactions contemplated hereby or thereby in any forum other than exclusively in the Supreme Court of the State of New York, County of New York, or, if under applicable Law exclusive jurisdiction is vested in the federal courts, the United States District Court for the Southern District of New York in the County of New York (and appellate courts thereof), (B) submits for itself and its property with respect to any such action to the exclusive jurisdiction of such courts, (C) waives and hereby irrevocably waives, to the fullest extent permitted by applicable Law, any objection which it may now or hereafter have to the laying of venue of, and the defense of an inconvenient forum to the maintenance of, any such action in any such court, (D) agrees that service of process, summons, complaint, notice or document or any other process that might be served in any action or proceeding may be made on any party by sending or delivering a copy of the process to such party to be served at the address of such party and in the manner provided for the giving of notices in Section 13.2 and shall be effective service of process against it for any such action brought in any such court, and (E) HEREBY WAIVES ANY RIGHT TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE INCREMENTAL AMENDMENT, THE DEBT FINANCING CONTEMPLATED THEREBY OR ANY OF THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY, INCLUDING ANY DISPUTE ARISING OUT OF OR RELATING IN ANY WAY TO THE INCREMENTAL AMENDMENT OR THE PERFORMANCE THEREOF OR THE TRANSACTIONS CONTEMPLATED THEREBY, WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT, TORT OR OTHERWISE, INCLUDING ANY ACTION OR PROCEEDING AGAINST ANY DEBT LENDER.

13.17.2 Notwithstanding anything herein to the contrary, each Seller (on behalf of itself and its respective officers, directors, employees, members, managers, partners and controlling Persons), each Company (on behalf of itself and its officers, directors, employees, members, managers, partners and controlling Persons) and each of the other parties hereto agrees that (i) none of it, its Affiliates and its and their respective officers, directors, employees, members, managers, partners and controlling Persons) and each of the other parties hereto agrees that (i) none of it, its Affiliates and its and their respective officers, directors, employees, members, managers, partners and controlling Persons shall have any rights or claims against any Debt Lender in connection with this Agreement, the Incremental Amendment, the Debt Financing contemplated thereby or thereby, whether at Law or equity, in contract, in tort or otherwise; provided, however, that the foregoing will not limit the rights of the parties to the Incremental Amendment or the definitive documentation for the Debt Financing contemplated thereby, and (ii) in no event will any Debt Lender be liable for punitive, special, exemplary, indirect, consequential or incidental damages or liabilities, or damages or liabilities argued to be associated with lost profits or diminution in value or damages or liabilities based on any type of multiple or damages of a tortious nature in connection with this Agreement, the Incremental Amendment, the Debt Financing contemplated thereby or any of the transactions contemplated thereby or any of the transactions contemplated thereby or any of the transactions contemplated thereby or damages or liabilities argued to be associated with lost profits or diminution in value or damages or liabilities based on any type of multiple or damages of a tortious nature in connection with this Agreement, the Incremental Amendment, the Debt Financing contemplated thereby or any of the transactions contemplated hereby or thereby, whether

Capital Pumping

Conflicts and Privilege. From and after the Closing Date, it is possible that McGinnis Lochridge LLP. ("Company 13.18 Counsel") will represent Sellers, Real Property Owners or other related Persons (individually and collectively, the "Seller Group"), in connection with the Escrow Amount, the transactions contemplated herein or in the Escrow Agreement, and any claims made pursuant to this Agreement, the Escrow Agreement or other Transaction Documents. Buyers and the Companies hereby agree that Company Counsel (or any successor thereof) may represent the Seller Group in the future to the extent related directly to issues that may arise under this Agreement, the Escrow Agreement, the Transition Services Agreement, the Leases, the other Transaction Documents, the administration of the Escrow Amount, and any claims that may be made thereunder pursuant to this Agreement, the Escrow Agreement, or the other Transaction Documents ("Indemnification Matters"). As to all privileged communications prior to the Closing Date between or among Company Counsel, Sellers or the Companies to the extent directly related to the Indemnification Matters, the attorney-client privilege and the expectation of client confidence with the Companies shall not be claimed to prevent or limit Company Counsel's representation hereunder; provided, however, that neither the Companies nor Buyers waive such attorney-client privilege. Company Counsel (or any successor thereof) may serve as counsel to all or a portion of the Seller Group or any member, manager, partner, officer, employee, representative, or Affiliate of the Seller Group in connection with of the Indemnification Matters; provided that in connection with the representation of Seller Group in any Indemnification Matter, no Company Counsel attorney will take any action that would reasonably be expected to jeopardize or waive attorney-client privilege or any other privilege of Buyers, the Companies, or any of their respective Affiliates. Buyers and the Companies each consent to such representation, and acknowledges that such consent voluntary.

[The remainder of this page is left blank intentionally.]

Capital Pumping

IN WITNESS WHEREOF, the parties have hereunto caused this Agreement to be duly executed as of the date first above written.

COMPANY PARTIES:

CAPITAL PUMPING, LP

By: Capital Rentals, LLC Its General Partner

> By: /s/ A. Keith Crawford A. Keith Crawford, Manager

ASC EQUIPMENT, LP

By: Central Texas Concrete Services, LLC Its General Partner

> By: /s/ A. Keith Crawford A. Keith Crawford, Manager

MC SERVICES, LLC

By: /s/ Melinda Crawford Melinda Crawford, Manager

Signature Page to Interest Purchase Agreement

CAPITAL RENTALS, LLC

By: <u>/s/ A. Keith Crawford</u> A. Keith Crawford, Manager

CENTRAL TEXAS CONCRETE SERVICES, LLC

By: <u>/s/ A. Keith Crawford</u> A. Keith Crawford, Manager

A. KEITH CRAWFORD

/s/ A. Keith Crawford (signature)

MELINDA CRAWFORD

/s/ Melinda Crawford (signature)

Signature Page to Interest Purchase Agreement

BUYERS:

BRUNDAGE-BONE CONCRETE PUMPING, INC.,

a Colorado corporation

/s/ Bruce Young

Name: Bruce Young Title: President and Chief Executive Officer

CPH ACQUISITION, LLC,

a Delaware limited liability company

BY: BRUNDAGE-BONE CONCRETE PUMPING, INC., its Managing Member

> /s/ Bruce Young Name: Bruce Young Title: President and Chief Executive Officer

Solely for the purposes of <u>Section 8.3</u>:

CONCRETE PUMPING HOLDINGS, INC., a Delaware corporation

/s/ Bruce Young

Name: Bruce Young Title: Chief Executive Officer

Signature Page to Interest Purchase Agreement



Concrete Pumping Holdings to Acquire Capital Pumping

- Transformative Acquisition Increases Efficiencies of Scale and Strategically Expands Presence in the Fragmented Texas Region –

DENVER, CO – March 18, 2019 – Concrete Pumping Holdings, Inc. (Nasdaq: BBCP) ("CPH" or the "Company"), a leading provider of concrete pumping services and concrete waste management services in the U.S. and U.K., has entered into a definitive agreement to acquire Capital Pumping, LP and its affiliates ("Capital Pumping") in an all-cash transaction. The purchase agreement was unanimously approved by CPH's board of directors and is expected to close in the second calendar quarter of 2019, subject to regulatory approvals and other customary closing conditions.

The transformative acquisition greatly enhances the Company's positioning and scale in the fragmented concrete pumping industry, specifically within the Texas region. The increased scale will provide CPH and its customers with advantages including increased breadth of services, expanded fleet availability and a platform to establish itself as the employer of choice in the industry.

"The purchase of Capital Pumping is a unique opportunity that is consistent with our acquisition strategy and adds a well-run business that shares our core values of safety, people and reliability," said Concrete Pumping Holdings CEO Bruce Young. "Capital Pumping's leading position in the highly fragmented Texas region, along with its efficient asset base that has an average fleet age of only three years, instantly attracted us to the business. Additionally, we see a compelling opportunity for Capital Pumping's customer base to access our Eco-Pan concrete waste removal services."

Matthew Homme, board member of CPH and Managing Director of Peninsula Pacific, a private equity firm based in Los Angeles, CA, said: "Under owner Keith Crawford's leadership, Capital Pumping has become a dynamic and well-regarded name in concrete pumping, and we look forward to continuing the proud stewardship of the business as part of expanding the CPH platform."

Capital Pumping generated Adjusted EBITDA¹ in 2018 of approximately \$23 million, excluding synergies that are expected to be realized through operational efficiencies. The acquisition provides compelling opportunities for CPH to grow its scale and diversity as well as enhance its financial profile. Such opportunities include:

- Increasing efficiencies of scale in Texas, which is highly fragmented;
- Strategically expanding its presence in the high-growth Texas region;
- Providing a larger customer base to expand the Eco-Pan business;
- Adding a highly complementary business and the youngest fleet of scale in the industry; and
- Providing compelling adjusted EBITDA, tax, capital expenditure and real estate synergies.

Young continued: "As we look to drive profitable growth over the long-term, Capital Pumping will play a key role in that strategy. We expect the deal will be accretive to CPH's profitability levels, supporting our ability to increase free cash flow and de-lever our balance sheet."

¹ Adjusted EBITDA is a financial measure that is not calculated in accordance with Generally Accepted Accounting Principles in the United States ("GAAP"). See "Non-GAAP Financial Measure" below for a discussion of Adjusted EBITDA and a reconciliation to net income.



Tariq Osman, Vice Chairman of CPH, and Partner and Managing Director of Argand Partners, a private equity firm based in New York, NY, commented: "We are excited to bring such a strong team and brand into the CPH organization and are confident that the combined company can drive long-term shareholder value."

Transaction Financing and Capital Structure Initiatives

CPH intends to finance the acquisition with a combination of proceeds from the issuance and sale of equity or equity-linked securities and additional term loans under its \$357 million Term Loan B facility. The proposed financing is subject to market conditions and other factors which could cause the terms, structure or nature of such financing to change materially.

On a pro forma basis giving effect to the acquisition of Capital Pumping as of the beginning of CPH's 2019 fiscal year, CPH expects its net leverage to decline by the end of its fiscal year 2019 to 3.5x or below from approximately 4.4x as of December 6, 2018, which is the date CPH consummated its business combination with Industrea Acquisition Corp. CPH believes that it will generate cash flows following the acquisition which will allow it to further reduce its net leverage ratio.

Additionally, in order to simplify its capital structure and reduce potential dilution from its outstanding warrants, CPH intends to commence an offer to exchange its common stock for its outstanding warrants, which is intended to be completed prior to or concurrently with the closing of the acquisition. The timing and terms of any offer to exchange shares of common stock for warrants is subject to market conditions and other factors, and no assurance can be given as to the commencement or completion of such offer to exchange.

Advisors

UBS Investment Bank, Robert W. Baird & Co. ("Baird") and Stifel Nicolaus & Company, Incorporated ("Stifel") acted as financial advisors to CPH. Ballard Spahr LLP and Winston & Strawn LLP acted as legal counsel to CPH. WoodRock / TCB Capital Markets acted as exclusive financial advisor and McGinnis Lochridge LLP acted as legal counsel to Capital Pumping.

Conference Call

CPH will hold a conference call Monday, March 18, 2019, at 10:00 a.m. Eastern time to discuss the transaction. The Company's CEO Bruce Young and CFO Iain Humphries will host the call, followed by a question and answer period.

Date: Monday, March 18, 2019 Time: 10:00 a.m. Eastern time (8:00 a.m. Mountain time) Toll-free dial-in number: 1-877-407-9039 International dial-in number: 1-201-689-8470 Conference ID: 13688042

Please call the conference telephone number 5-10 minutes prior to the start time. An operator will register your name and organization. If you have any difficulty connecting with the conference call, please contact Liolios at 1-949-574-3860.



The conference call will be broadcast live and available for replay <u>here</u> and via the investor relations section of the Company's website at <u>www.concretepumpingholdings.com</u>. An investor presentation relating to the transaction will also be posted on the Company's website.

A replay of the conference call will be available after 1:00 p.m. Eastern time on the same day through April 15, 2019.

Toll-free replay number: 1-844-512-2921 International replay number: 1-412-317-6671 Replay ID: 13688042

About Concrete Pumping Holdings, Inc.

CPH is the leading provider of concrete pumping services and concrete waste management services in the U.S. and the U.K., operating under the only established, national brands in both regions (Brundage-Bone and Camfaud, respectively). The Company's large fleet of specialized pumping equipment and trained operators position it to deliver concrete placement solutions that facilitate substantial labor cost savings to customers, shorten concrete placement times, enhance worksite safety and improve construction quality. CPH is also the leading provider of concrete waste management services in the U.S., operating under the only established, national brand – Eco-Pan. Highly complementary to its core concrete pumping service, Eco-Pan provides a full-service, cost-effective, regulatory-compliant solution to manage environmental issues caused by concrete washout. As of October 31, 2018, CPH provides concrete pumping services in the U.S. from a footprint of 80 locations across 22 states, concrete pumping services in the U.K. from 28 locations, and route-based concrete waste management services from 13 locations in the U.S. For more information, please visit www.concretepumpingholdings.com or CPH's brand websites at www.brundagebone.com, www.camfaud.co.uk, or www.eco-pan.com.

About Capital Pumping, LP

Capital Pumping, LP ("Capital Pumping") is a leading provider of concrete pumping services in Texas. Founded in and family-owned since 1971, Capital Pumping provides highly trained pump operators, experienced mechanics and specialized equipment to provide high quality service to its customers.

Forward-Looking Statements

This press release includes "forward-looking statements" within the meaning of the "safe harbor" provisions of the Private Securities Litigation Reform Act of 1995. The Company's actual results may differ from their expectations, estimates and projections and consequently, you should not rely on these forward-looking statements as predictions of future events. Words such as "expect," "estimate," "project," "budget," "forecast," "anticipate," "intend," "plan," "may," "will," "could," "should," "believes," "predicts," "potential," "continue," and similar expressions are intended to identify such forward-looking statements. These forward-looking statements include, without limitation, the Company's expectations with respect to future performance, including the performance of Capital and the expected benefits from the Capital acquisition, the expected financing for the acquisition, and the potential offer to exchange CPH shares of common stock for warrants. These forward-looking statements involve significant risks and uncertainties that could cause the actual results to differ materially from the expected results. Most of these factors are outside the Company's control and are difficult to predict. Factors that may cause such differences include, but are not limited to: the outcome of any legal proceedings that may be instituted against the Company or its subsidiaries; the ability to recognize the anticipated benefits of the Business Combination, which may be affected by, among other things, competition, the ability of the Company to grow and manage growth profitably and retain its key employees; changes in applicable laws or regulations; the possibility that the Company may be adversely affected by other economic, business, and/or competitive factors; and other risks and uncertainties indicated from time to time in the Company's filings with the Securities and Exchange Commission. The Company cautions that the foregoing list of factors is not exclusive. The Company cautions readers not to place undue reliance upon any forward-looking statements, which speak only as of the date made. The Company does not undertake or accept any obligation or undertaking to release publicly any updates or revisions to any forward-looking statements to reflect any change in its expectations or any change in events, conditions or circumstances on which any such statement is based.



Non-GAAP Financial Measures

Adjusted EBITDA is a financial measure that is not calculated in accordance with Generally Accepted Accounting Principles in the United States ("GAAP"). The Company believes that this non-GAAP financial measure provides useful information to management and investors regarding certain financial and business trends relating to the Company's and Capital's financial condition and results of operations. Management also uses this non-GAAP financial measure to compare performance to that of prior periods for trend analyses, determining incentive compensation and for budgeting and planning purposes. Adjusted EBITDA is also used in quarterly financial reports prepared for the Company's board of directors. The Company believes that this non-GAAP measure provides an additional tool for investors to use in evaluating ongoing operating results and in comparing financial results with competitors who also present similar non-GAAP financial measures.

Adjusted EBITDA is defined as net income calculated in accordance with GAAP plus interest expense, income taxes, depreciation, amortization, transaction expenses, gain (loss) on sale of assets, non-recurring adjustments, management fees and other one-time and non-operational expenses.

Current and prospective investors should review the Company's audited financial statements, which are filed with the U.S. Securities and Exchange Commission, and not rely on any single financial measure to evaluate the Company's business. Other companies may calculate Adjusted EBITDA differently and therefore this measure may not be directly comparable to similarly titled measures of other companies.



The following is a reconciliation of Capital's Adjusted EBITDA to net income for its fiscal year ended December 31, 2018.

	Fiscal Year Ending December 31,			
<u>(</u> \$000s)	<u>2018</u>			
Net Income	\$	14,247		
Interest Expense, net		589		
Misc. Income		(27)		
D&A		10,943		
Reported EBITDA	\$	25,752		
Adjustment for Normalized Level of Gain on Sale	\$	(2,794)		
Normalized Repairs and Maintenance Expense		(265)		
Bad Debt Expense Reversal		(77)		
Owner Compensation ¹		68		
Accounting Fees		(63)		
Other Adjustments ²		162		
Capital Adjusted EBITDA	\$	22,783		

1) Current Owner of Capital will not remain with the Company post-transaction. Existing management team will continue to run Capital post-transaction.

2) Other Adjustments include rent expense related to the Austin office (which will not be included in the transaction), removal of taxes from G&A, and other non-recurring or one-off items.

Contact:

Liolios, Investor Relations Cody Slach 1-949-574-3860 BBCP@Liolios.com









CPH to Acquire Capital Pumping, a Leading U.S. Concrete Pumping Provider

March 18, 2019

Disclaimer

Important Information

entation") is not intended to form the basis of any investment decision. It is for informational purposes only and does not constitute an offer to sell, a solicitation of an offer to buy, or a recommendation to purchase any equity, debt This presentation ("Investor Pres This presentation ("Investor Presentation") is not interded to form the basis of any investment decision. It is for informational purposes only and does not constule an offer to sell, a solicitation of an offer to buy, or a recommendation to purch the print consent of the company or other financial instruments of concrete pumping holdings, inc, the "Company" or any of the Company" settiliates. Any reportation or a vibration of the purch settilines and exertal sources. No representation in state as to the reasonableness of the assumptions made within or the accurscy or completeness of any projections or modeling or any other information to purport to be all-inclusive. The data contained herein is derived from various internal and external sources. No representation in which exists one of its contents, without on the accurscy or completeness of any projections or modeling or any other information to the information in this investor Presentation. Crash information contained herein has been derived who sources prepared by thing partice. The basis of the purposes used herein the Company with respective administration. Certain the Company index on the print constant of the accurscy or completeness to the accurscy or sources prepared by the print parties. While sources prepared by the print parties. While sources prepared by the print parties. While sources prepared by the print parties. The data contained herein has been derived who sources prepared by the print parties. The print process used herein the Company in waters prepared by the information in this investor Presentation or warranty, express or implied, is on will be given by the Company or any of their respective administ, directors, efficient, employees or any other personals to the accurscy or completeness of the information or any other withen, carly express or implied, is on will be given by the cortex or discussor or instantement, hereitigned or otherwise metals and expression and participant and excurscy or completeness of the information or any other

Forward-Looking Statements

Forward-Looking Statementa This Investor Presentation includes "forward-looking statements" within the meaning of the "sale harbor" provisions of the Private Securities Litigation Reform Actof 1995. The Company's actual results may differ from its expectations, estimates and projections and consequently, you should not rely on these forward-looking statements as predictions of iture events. Works such as "expect," "leadinate," "houde," "budget," "forecast," "anticipate," "Intend," "plan," "thay," "will,""could," "binuity," "bound," "binuity are appresentation includes "forecast," "anticipate," "Intend," "plan," "thay," "will,""could," "binuity," budget," forecast," "anticipate," intend," "plan," "thay," "will,""could," "binuity," budget," forecast," "anticipate," intend," "plan," "thay," "will,""could," believes," "predicts," "potential," "continue," and similar expressions are intended to identify such toward-looking statements. These forward-looking statements involve significant risks and uncertainties that could cause the actual results to differ materiality from the company is control and are difficult to predict. Factors that may cause such differences include, but are not limited to the inability to obtain or maintain the listing of the Company's states of common stock on The Naskaq. Stock Natket, the actility to econice the ancipitate lower oregulates, and conservations, and the company is applicable to be diversely affected by, among other things, competition, the activity of the Company to grow and manage growth profilably and retain its key employees, costs related by the usiness combination, change commission. The Company cautions that the forgong list offsctors is not exclusive. The Company cautions that the torography results enders on unice and uncertainte indicated the new enders on unice and uncertainte to the interval to the date made. The Company cautions that the torography results and torographic enders on an other enders on the indicated predicts and cause statements, which are company obligation or undertaking to release publicity any updates or revisions to any forward-looking statements to reflect any change in its expectations or any change in events, conditions or circumstances on which any such statement is based

Industry and Market Data

In this investor Presentation, we rely on and refer to information and statistics regarding market participants in the sectors in which the Company competes and other industry data. We obtained this information and statistics from third-party sources, including regords by market research firms, and company filings. npany filings

Historical and Projected Financial Information

Historical and Projected Financial Information Annual financial intomation of the Company is based on its fiscal year end of October 31. Annual financial information of Capital is based on Capital's fiscal year end of December 31. This Investor Presentation contains financial forecasts, which were prepared in good failth by the Company on a basis believed to be reasonable. Such financial intercasts have not audited, reviewed, compiled or performed any procedures with respect to the projections for the purpose of their inclusion in this investor Presentation, and accordingly, they have not expressed an opinion nor provided any other twom of assurance with respect thereto for the purpose of their Investor Presentation. These projections are for illustrative purposes only and should not be relied upon as being necessarily indicative of future results. Certain of the above-mentioned projected information has been provided for purposes of providing comparisons with historical data. The assumptions and estimates understring themation are interently uncertain and are subject to a vide variety of significant business, economic and competitive risks and uncertainties that could cause actual results to differ materially thom those contained in the prospective financial information. Projections are interently uncertain due to a number of factors outside of the Company's control. Accordingly, there can be no assume that the prospective financial information. Projections are interently uncertain due to a number of factors outside of the Company's control. Accordingly, there can be no assume that the prospective financial information are interent the factors outside of the Company's forthol. Accordingly, there can be no assume that the prospective financial information the respective financial information. indicative of Adure performance of the Company or that actual results will not differ materially from those presented in the prospective financial information. Inclusion of the prospective financial information in this investor Presentation should not be regarded as a representation by any person that the results contained in the prospective financial information will be achieved.

Non-GAAP, Financial Measures

Non-AAP interclaimseures This investor Presentation includes non-GAAP financial measures, including Adjusted EBITDA of the Company and Capital. The Company defines Adjusted EBITDA as net income (loss) plus interest expense, income taxes, depreciation and amortization, as further adjusted to eliminate the impact of other non-cash or non-core operating expenses. These measures should not be used as subsitutes for their most comparable measures calculated in accordance with GAAP. See Reconciliation of Non-GAAP (Neasu on Sildes 23 and 24. The Company's and Capital's management and investors regarding certain financial and business tends relating to the Company's and Capital's financial condition and results of operations. The Company's and Capital's management uses Adjusted EBITDA to compare performance to that of prior periods for them danges and for budgeting and planning purposes. Measures

A reconciliation of non-GAAP forward looking information to their corresponding. GAAP measures for the Company has not been provided due to the lack of predictability regarding the various reconciling items such as provision for income taxes and depreciation and amortization, which are expected to have a material impact on these measures and cannot be reasonably predicted without unreasonable efforts. You should not rely on any single financial measure to evaluate Capitar's business. Other companies may calculate Adjusted EBITDA differently, and therefore it may not be directly comparable to similarly titled measures of other companies.

	OR OFTER
CONCERNING	GRATIAL
SPECIAL PLANE	2.1
200670036042	- 21

Introduction to CPH Senior Management



CPH's Proven M&A Platform

M&A Playbook

- Acquirer of Choice: Completed 45+ acquisitions since 1983 (avg. pre-synergy Adjusted EBITDA multiples <4.5x)
- Benefits of Scale: Track record of increasing Adjusted EBITDA margins of target within first few years through utilization increases, price optimization and cost synergies
- Clear Acquisition Criteria: Strong management, good employee and customer relationships, well maintained fleet and meaningful potential for synergies
- Attractive Tax Benefits Available: Transactions typically structured for 100% cost expensing for tax purposes
- Strong Acquisition Pipeline: ~\$100 million of additional Adjusted EBITDA identified

Acquisitions since 2015

Company Name	Locations	Purchase Price (millions)	
Solid Rock	TX	\$1.1	2.6x
Dyna Pump	ТХ	\$0.3	1.6x
Action	SC, TN, AL	\$5.6	7.3x
AJ / Kenyon	SC	\$1.7	2.1x
Oxford	U.K.	£45.5	4.4x
Reilly	U.K.	£10.2	4.0x
O'Brien	со	\$21.0	4.0x
Capital (pending)	ΤХ	\$129.2	5.7x

Note: Figures above are indicative of historical acquisition results. There can be no assurances that future acquisitions will occur or perform in line with historical achievements. (i) Estimated acquisition Adjusted EBITDA multiples are before synergies.



Transaction Highlights

CPH acquiring Capital Pumping ("Capital"), a leading U.S. concrete pumping provider, with FY 2018 revenue of \$49 million and Capital Adjusted EBITDA of \$23 million⁽ⁱ⁾

Strategic Overview	 Transformational transaction consistent with CPH's acquisition strategy Increases efficiencies of scale in Texas, which is highly fragmented Strategically expands CPH's presence in high-growth Texas region Adds highly complementary and youngest fleet of scale in the industry Compelling EBITDA, tax, capex and real estate synergies Leverages CPH's core competency in M&A integration Capital shares CPH's core values of safety, people and reliability
Financial Overview	 \$129.2 million cash purchase price, funded with proceeds from the expected issuance and sale of equity or equity-linked offering, and the issuance of additional term loan debt by CPH to occur prior to or concurrently with the closing of the acquisition Purchase multiple of 5.7x FY 2018 Capital Adjusted EBITDA⁽ⁱ⁾, 3.6x including the benefit of EBITDA, tax, capex and real estate synergies ~\$4.9 million of identified annual EBITDA synergies⁽ⁱⁱ⁾ ~\$18.6 million NPV of identified tax benefits ~\$9.0 million NPV of identified capex synergies⁽ⁱⁱ⁾ ~\$1.2 million NPV of identified real estate synergies⁽ⁱⁱ⁾ Expected net leverage of ≤3.5x by year end FY 2019 with a strong path to further deleveraging Acquisition expected to close by the end of calendar Q2 2019 CPH intends to commence an offer to exchange its common stock for its outstanding warrants, which is intended to be completed prior to or concurrently with the closing of the acquisition

(i) Capital Adjusted EBITDA is a non-GAAP financial measure. Refer to Slides 23 and 24 for a reconciliation of Adjusted EBITDA to net income. 5⁽ⁱⁱ⁾ Majority of total cash flow synergies expected to be realized within 24 months.





Capital Pumping Overview

- Capital Pumping is a leading provider of concrete pumping services to the Texas commercial and residential construction sectors
- Product offering includes boom pumps, trailer pumps and placing booms
- Capital operates the youngest fleet of scale in the industry with a weighted average asset life of 2.9 years
- Founded in 1971 and currently family-owned



Attractive Transaction Economics



Illustrative Transaction Sources and Uses

(\$ in millions)

Sources		Uses			
New Equity Securities	\$95	Transaction Purchase Price	\$129		
Incremental Term Loan B ⁽ⁱ⁾	40	Transaction Fees & Expenses	6		
Total Sources	\$135	Total Uses	\$135		

Note: The financing structure is presented for illustrative purposes only. The actual terms of the transaction, including but not limited to the aggregate amount of equity or equity-linked securities to be issued, the terms of any equity or equity-linked securities that may be issued and the amounts (if any) to be drawn under the existing Term Loan, will be determined by CPH based on market conditions.

(i) CPH plans to raise debt through its existing Term Loan's incremental availability basket. For further details on CPH's existing Term Loan, please refer to Slide 22.



Transaction Structure and Cash Flow Generation Supports Deleveraging



Increased Scale in Highly Fragmented Industry



Note: "U.S. Concrete Pumping - Brundage Bone" position based on FY 2018 Revenue Pro Forma for Acquisitions. Capital position based on FY 2018 Revenue.



Scale Provides Advantages



Compelling Combined Fleet Profile



Overview of Transaction Synergies

(\$ in millions)



Attractive Combined Financial Profile

(\$ in millions)	Concrette PUMPING Holddings FY 2018	FY 2018	
Adjusted Revenue	\$251 ⁽ⁱ⁾	\$49	\$307 ⁽ⁱⁱ⁾
Adjusted EBITDA	83 ⁽ⁱⁱⁱ⁾	23 ⁽ⁱⁱⁱ⁾	111 ^(iv)
Adjusted EBITDA Margin (%)	33%	46%	36%
Сарех	28	9	28 ^(v)
Adjusted EBITDA Less Capex	55	14	83
Adjusted EBITDA Less Capex (as a % of Revenue)	22%	28%	27%

Note: Combined financials do not represent a CPH forecast but rather represent a pro forma illustration of the combined company's potential financial performance. CPH and Capital have fiscal year ends of October 31 and December 31, respectively.

(i) Po Forms trace constant currency adjustments. Adjusted Revenue for CPH is a non-GAAP financial measure that is calculated by adding the following items to CPH's FY 2018 GAAP revenue of \$243.2 million: (i) \$7.0 million of pre-acquisition revenue from the acquisition of OPH's in April 2018; and (i) \$1.1 million of constant currency adjustment based on a GBP to USD exchange ratio of 1.370. (ii) Adjusted EBITDA is a non-GAAP financial measure. Refer to Slides 23 and 24 for a reconciliation of Adjusted EBITDA is not income. (ii) Adjusted EBITDA is a non-GAAP financial measure. Refer to Slides 23 and 24 for a reconciliation of Adjusted EBITDA is not income. (ii) Represents CPH's FY 2018 Adjusted EBITDA pus identified number transaction EBITDA synergies of \$4.9 million, expected to be realized within 24 months. Adjusted EBITDA is a non-GAAP financial measure. (ii) Represents CPH's FY 2018 Capex plus Capitar's FY 2018 Capex less NPV or identified transaction Capex synergies of \$3.9 million. 14



Combined Platform with Significant Scale and Diversity



Complementary Texas Footprint



Increased Exposure to Highly Attractive Texas Region





2019 Non-Residential Construction





Population Growth in Southern Metropolitan Areas(iii)

	2017	2018
Atlanta, GA	4%	6%
Austin, TX	2%	7%
Birmingham, AL	(10%)	13%
Charleston, SC	(12%)	4%
Charlotte, NC	5%	4%
Dallas, TX	2%	10%
Houston, TX	1%	14%
Jacksonville, FL	3%	4%
Miami, FL	18%	0%
Nashville, TN	(496)	(5%)
New Orleans, LA	38%	(14%)
Oklahoma City, OK	(3%)	17%
Orlando, FL	7%	(5%)
Raleigh, NC	(10%)	5%
San Antonio, TX	10%	13%
Tampa, FL	(4%)	9%
Tulsa, OK	10%	12%

Texas Employment Gains Through 2020 (%)^(v)

2020 U.S. Employment Growth Rate of 1.2%^(v)



20

0

- Sources: (i) HS State Construction Forecast. (ii) Global Insight. (iii) Oldoastie Business Intelligence 2018 North American Construction Forecast Report. (iv) Texas A&M Real Estate Center November 2018 Texas Housing Insight Report. (iv) The Perman Group. (iv) Represents the annual growth in employment from 2017-2020 sourced from OECD. 17

2013 2014 2015 2016 2017 2018P 2019P 2020P 2021P





18

-PUNDANO

Attractive Financial Profile & Valuation Versus Peers



Appendix

Large, Growing Market Supported by Compelling Tailwinds



Illustrative Shares and Other Equivalents Outstanding

	Common Stock	Other Shares and Equ	Total Potential Outstanding Stock (Fully Diluted)	
	Outstanding Shares Pro Forma for Capital Transaction	Shares Underlying Convertible Securities ^(iii, iv)	Shares Underlying Outstanding Warrants ^(v)	
Shares By Type				
Peninsula Pacific®	11,005,275	-1	-	11,005,275
Nuveen ⁽⁰⁾	-	2,450,980	-	2,450,980
Other Public Shares	12,497,246%	10	-	12,497,246
Shares Underlying Public Warrants	2	-	23,000,000	23,000,000
Total Unrestricted Shares (Public Shares)	23,502,521	2,450,980	23,000,000	48,953,501
CPH Management & Employees (Current and Former)	2,942,048	2,354,507	-	5,296,555
Argand Partners	12,187,638	10	10,822,500	23,010,138
Non-Executive Directors ⁽ⁱⁱ⁾	143,750	13	277,500	421,250
Total Restricted Shares	15,273,436	2,354,507	11,100,000	28,727,943
Fully Diluted Total Outstanding Shares	38,775,957	4,805,487	34,100,000	77,681,444
Cumulative Fully Diluted Total Outstanding Shares ⁽ⁱⁱ⁾	38,775,957	43,581,444	77,681,444	77,681,444

or concurrently with the closing of the acquisition

(i) Assumes transaction closes 180 days after December 6th, 2018 and the lock-up for Peninsula Pacific and Nuveen has expired.

(ii) Pro forma for the Capital Transaction assuming a \$95 million equity offering at an illustrative \$9.59/share, which is the 30-day VWAP as of March 15, 2019.

(iii) Nuveen may elect to convert its Preferred Stock into shares of Common Stock at a 1:1 ratio at any time 180 days after December 6th, 2018. The total number of shares of Common Stock into which the Preferred Stock will be converted will be 2,450,980 shares of Common Stock (subject to anti-dilution protection rights afforded to the holder of the Preferred Stock).

(iv) CPH Management & Employees (Current and Former) hold i) 2,459,406 "in the money" options with a strike price of \$0.87 (which results in a further 2,236,259 shares of Restricted Common Stock assuming a conversion stock price of \$9.59/share based on the Treasury Stock Method), and ii) 224,673 "in the money" options with a strike price of \$6.09 (which results in a further 118,248 shares of Restricted Common Stock assuming a conversion stock price of \$9.59/share based on the Treasury Stock Method). \$9.59 is the 30-day WWAP as of March 15, 2019.

(v) Each warrant is currently exercisable for one share of Common Stock at an exercise price of \$11.50/share. The warrants held by Argand Partners and Non-Executive Directors can be exercised on a cashless basis, but are assumed to be exercised for cash in this analysis. The Company may redeem the outstanding warrants at a price of \$0.01 per warrant if the last sale price of the Common Stock equals or exceeds \$18.00/share for 20 out of 30 assumed to b trading days.

(vi) Non-Executive Directors includes 57,500 shares of Common Stock and 111,000 warrants held by Tom Armstrong and Genry Rooney who were directors of Industrea Acquisition Corp., but are not on the board of the Company. (vii) Cumulative Fully Diluted Total Outstanding Shares in the "Other Shares and Equivalents Outstanding" columns represent the cumulative amount of outstanding shares of Common Stock if each of the potential events in items in a source were to occur in the order presented. 21



CPH Credit Facilities Summary

Credit Facilities	- \$357 million Term Loan Facility ⁽ⁱ⁾ - \$60 million ABL Revolver ⁽ⁱⁱ⁾
Interest Rate	- Term Loan Facility: Libor + 600bps - ABL Revolver: Libor + 175-225bps based on leverage levels
Tenor	- Term Loan Facility: 7 Years (December 6 th , 2025) - ABL Revolver: 5 Years (December 6 th , 2023)
Term Loan Amortization	- 1.25% per quarter, bullet at maturity
Term Loan Call Protection	- 101 Soft Call for 12 Months (December 6⁵, 2019)
Incremental	- Term Loan Facility: Greater of \$40 million and 0.5x EBITDA free and clear, plus unlimited at 3.5x net first lien leverage - ABL Revolver: Up to \$30 million
Financial Covenants	- Term Loan Facility: None - ABL Revolver: Springing 1:1 Fixed Charge Coverage Ratio if at any time total Excess Availability is less than the greater of (i) 10% of the Line Cap, (ii) \$5 million, and (iii) 12.5% of the U.K. Borrowing Base

(i) Pro toma for the Capital transaction, CPH's Term Loan facility will increase by \$40 million from its initial level of \$357 million to \$397 million. (ii) The ABL and cash balances were approximately \$17.3 million and \$4.8 million as of January 31, 2019, respectively.



Reconciliation of Capital Net Income to Capital Adjusted EBITDA

	Fiscal Year Ending December 31,			
(\$000s)	2016	2017	2018	
Net Income	\$13,551	\$11,677	\$14,247	
Interest Expense	769	721	628	
Interest Income	(34)	(20)	(39)	
Misc. Income	-	-	(27)	
D&A	8,434	9,548	10,943	
Reported EBITDA	\$22,720	\$21,927	\$25,752	
Adjustment for Normalized Level of Gain on Sale	(1,161)	(1,744)	(2,794)	
Normalized Repairs and Maintenance Expense	(52)	299	(265)	
Bad Debt Expense Reversal	-	-	(77)	
Owner Compensation ⁽ⁱ⁾	90	67	68	
Accounting Fees	(34)	(25)	(63)	
Other Adjustments ⁽ⁱⁱ⁾	272	193	162	
Capital Adjusted EBITDA	\$21,835	\$20,718	\$22,783	

(i) Current Owner of Capital will not remain with the Company post-transaction. Existing management team will continue to run Capital post-transaction. (ii) Other Adjustments include rent expense related to the Austin office (which will not be included in the transaction), removal of taxes from G&A, and other non-recurring or one-off items.



Reconciliation of CPH Net Income to Adjusted EBITDA

	Years Ended October 31,						
(\$000s)		FY2015	FY2016		FY2017	F	Y2018
Revenue Pro Forma for Acquisitions							
Revenue, reported	5	147,361	\$ 172,426	\$	211,211	\$	243,22
U.K. Concrete Pumping - Camlaud revenue (pre-acquisition)		45,685	50,530		8,357		
O'Brien revenue (pre-acquisition)		11,182	13,553		13,796		6,99
Revenue Pro Forma for Acquisitions		204,228	236,519		233,364		250,21
Constant currency adjustment ®		(5,000)	(814		3,277		1,11
Adjusted Revenue Pro Forma for Acquisitions	\$	199,228	\$ 235,705	\$	235,641	\$	251,32
Net income, reported	\$	3,509		\$	913	5	28,38
U.K. Concrete Pumping - Camfaud net income (pre-acquisition)		10,057	11,341		404		
O'Brien net income (pre-acquisition)		3,702	4,799	2	4,909		2
Net Income Pro Forma for Acquisitions		17,268	22,374		6,226		28,64
interest expense, reported	\$	20,492	\$ 19,516	\$	22,748	\$	21,43
U.K. Concrete Pumping - Camfaud interest expense (pre-acquisition)		575	565		588		
O'Brien interest expense (pre-acquisition)		38	-	2			
Interest Expense Pro Forma for Acquisitions		21,105	20,081		23,336		21,4
income tax expense / (benefit), reported	\$	2,020	\$ 4,454	\$	3,757	\$	(9,7
U.K. Concrete Pumping - Camfaud income tax expense (pre-acquisition)			141		87		
O'Brien income tax expense (pre-acquisition)		-	-	6			
Income Tax Expense Pro Forma for Acquisitions		2,020	4,595		3,844		(9,7)
Depreciation and amortization, reported	\$	20,603	\$ 22,310	\$	27,154	\$	25,6
U.K. Concrete Pumping - Camfaud depreciation and amortization (pre-acquisition)		3,607	3,984		1,025		
O'Brien depreciation and amortization (pre-acquisition)		-			93		
Depreciation and Amortization Pro Forma for Acquisitions		24,210	26,294	2	28,272		25,66
EBITDA Pro Forma for Acquisitions		64,604	73,344	2	61,678		65,95
EBITDA adjustments:							
Debt refinancing costs	\$	964	\$ 691	\$	5,401	\$	
Acquisition costs		290	3,644		4,343		7,5
One-time employee costs ^(I)		-	29		997		
Other adjustments III		2,461	4,761		4,964		9,1
Constant currency adjustment *		(1,626)	(247		1,031		6
Adjusted EBITDA Pro Forma for Acquisitions	\$	66,692	\$ 82,222	\$	78,414	\$	83,38
Capex Pro Forma for Acquisitions							
Maintenance Capex		12,438	19,311		12,747		20,64
Growth Capex		17,283	11,323		10,484		7,3
Pro Forma Total Capex		29,721	30.634	2	23.231		27.96

Note: CPH's U.K. segment (Cambud) was acquired in November 2016 and is consolidated in the facal year 2018 and 2017 financial statements. Financial results of Cambud are captured segarately prior to this data and are labeled as " "pre-acquisition" and are consolidated within CPH's "reported" financials for periods after November 2016. O'Enen was acquired in April 2018 and 12118 financial results are included as "pre-acquisition" financials for 2016, 2017, 2016 and 2015. (i) Constant currency based on a GBP to USD exchange and the CH 370. (ii) One-time employee costs include severance, relocation, fining and recruiting expenses. (iii) O'Ene adjustments include management & board tess, transaction-related and other non-ordinary course legal fees, stock option expense, start-up costs, and other transaction-oriented, project-oriented, normalizing and non-operating incometexperse items.

24